DUAL LABOR MARKETS AND THE EQUILIBRIUM DISTRIBUTION OF FIRMS

2024

BANCO DE ESPAÑA Eurosistema

Documentos de Trabajo N.º 2442

Josep Pijoan-Mas and Pau Roldan-Blanco DUAL LABOR MARKETS AND THE EQUILIBRIUM DISTRIBUTION OF FIRMS

DUAL LABOR MARKETS AND THE EQUILIBRIUM DISTRIBUTION OF FIRMS (*)

Josep Pijoan-Mas (**)

CEMFI AND CEPR

Pau Roldan-Blanco^(***)

BANCO DE ESPAÑA, CEMFI AND UNIVERSIDAD AUTÓNOMA DE BARCELONA

(*) The authors are grateful to Andreas Gulyas, Cristina Lafuente, Ludovic Panon, Edouard Schaal, and Ronald Wolthoff for their insightful discussions. We also thank Manuel Arellano, Samuel Bentolila, Adrien Bilal, Olympia Bover, Alex Clymo, Juan J. Dolado, Mike Elsby, Niklas Engbom, Javier Fernández-Blanco, Axel Gottfries, Juan F. Jimeno, Leo Kaas, Philipp Kircher, Matthias Kredler, Kieran Larkin, Salvatore Lo Bello, Sevi Rodríguez Mora, Tom Schmitz, Alireza Sepahsalari, Federico Tagliati, Gianluca Violante, Ludo Visschers and Felix Wellschmied for helpful suggestions, Ivan Auciello-Estévez and Beatriz González for their assistance and insights with the data analysis, and seminar and conference participants at Banco de España, CEMFI, the University of Edinburgh, University of Mannheim, Universitat d'Alacant, Stockholm University, IER Hitotsubashi University, the 2022 ECB-CEPR Labour Market Workshop, the IZA Workshop on the Macroeconomics of Labor Productivity, the 2022 European Winter Meeting of the Econometric Society, the VI Workshop of the Spanish Macro Network, the 2022 SED Meetings, the 2023 BSE Summer Forum, CEPR's ESSIM 2023, the 2023 Kent Macro Workshop, the 2023 SaM Workshop in Bristol, and the Third Annual DC SaM Workshop for helpful comments.

Pijoan-Mas acknowledges financial support from the Spanish Government under Grant PID2022-142790NB-I00 funded by MICIU/AEI/10.13039/501100011033 and by "ERDF/EU".

(**) CEMFI: Calle Casado del Alisal, 5, 28014 Madrid, Spain. Contact: pijoan@cemfi.es.
(***) Banco de España: Calle Alcalá, 48, 28014 Madrid, Spain. Contact: pau.roldan@bde.es.

Documentos de Trabajo. N.º 2442 October 2024

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the Internet at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2024

ISSN: 1579-8666 (on line)

Abstract

We study the effects of dual labor markets (i.e. the co-existence of fixed-term and openended contracts) on the allocation of workers within and across firms, the equilibrium distribution of firms, aggregate productivity, and welfare. Using rich Spanish administrative data, we document that the use of fixed-term contracts is very heterogeneous across firms within narrowly defined sectors. In particular, there is a strong relationship between the share of temporary workers and firm size, which is positive when looking at withinfirm variation but negative when looking at the variation between firms. To explain these facts, we use a directed search model of multi-worker firms, with ex-ante firm heterogeneity in technology types, and ex-post firm heterogeneity in transitory productivity, the composition of employment by contract type (fixed-term or open-ended) and human capital accumulated on the job. In counterfactual exercises, we find that limiting the use of fixed-term contracts decreases the share of temporary employment and increases aggregate productivity, but it also reduces total employment and leads to an overall decline in total output and welfare. The increase in productivity comes from an improved selection of firms, which more than offsets an increased misallocation of workers across firms.

Keywords: dual labor markets, temporary contracts, firm dynamics, unemployment.

JEL classification: D83, E24, J41, L11.

Resumen

En este documento estudiamos los efectos de la dualidad en el mercado de trabajo - es decir, la coexistencia de contratos temporales y contratos indefinidos- en la asignación de trabajadores dentro de la empresa y entre empresas distintas, la distribución de empresas en equilibrio, la productividad agregada y el bienestar social. Usando datos administrativos para España, documentamos que el uso de la temporalidad es muy heterogéneo entre empresas dentro de un mismo sector. En concreto, existe una fuerte relación entre la proporción de trabajadores temporales y el tamaño de la empresa, que es positiva cuando se observa la variación intraempresa, pero negativa cuando se observa la variación entre empresas. Con el objetivo de explicar estos resultados, construimos un modelo de búsqueda dirigida con empresas de múltiples trabajadores, heterogeneidad ex ante en tipos de tecnología y heterogeneidad ex post en productividad transitoria y en la composición del empleo por tipo de contrato y capital humano acumulado en la empresa. A través de ejercicios contrafactuales, encontramos que limitar el uso de contratos temporales disminuye la temporalidad e incrementa la productividad agregada, pero también reduce el empleo total y conduce a una disminución general en la producción total y el bienestar social. El incremento de la productividad proviene de una mejor selección de empresas, lo que compensa con creces un aumento en la mala asignación de trabajadores entre empresas.

Palabras clave: mercado de trabajo dual, contratos temporales, dinámicas empresariales, desempleo.

Códigos JEL: D83, E24, J41, L11.

1 Introduction

Many labor markets are characterized by a two-tier system, with the co-existence of open-ended contracts (OECs), protected by large termination costs, and fixed-term contracts (FTCs) of short duration. While extensive research has examined the impact of labor market duality on workers, its effects on firm choices and outcomes remain largely unexplored. In this paper, we analyze how different firms use each type of contract, and how their hiring and firing choices affect the allocation of workers across firms, firm entry and exit, and the equilibrium distribution of firms. This allows us to assess the effects of dual labor markets on the aggregate productivity of countries and to study the macroeconomic effects of policies aimed at reducing the share of temporary workers in the economy.

We start by documenting new facts about the heterogeneous use of FTCs across firms. To do so, we exploit Spanish administrative firm-level data for the period 2004-2019. The case of Spain is of particular interest because its labor market duality is very stark, with a high incidence of FTCs and a strong employment protection for OECs.¹ We uncover three important facts. First, there is a large degree of heterogeneity in the usage of FTCs across firms. For instance, among firms with 11 to 50 employees, the average temporary share is 25%, but the firms at the 10th and 90th percentiles of the distribution employ nearly 0% and 68% of workers under FTCs, respectively. Second, although the use of FTCs varies greatly across narrowly-defined industries, provinces, and time, most of the variation occurs due to firm-specific factors within industry, province, and time period, of which time-invariant unobserved firm characteristics play an overwhelmingly relevant role. And third, exploiting the panel dimension of our data, we find that the *within-firm* variation uncovers a positive correlation between firm size and the share of temporary workers, while the *between-firm* variation shows a negative relationship between firm size and the share of temporary workers.

Next, we write a model of firm dynamics with search-and-matching frictions and a two-tier labor market structure to understand these facts and the macroeconomic consequences of dual labor markets. As in Kaas and Kircher (2015) and Schaal (2017), a set of multi-worker firms operating a decreasing returns-to-scale technology can direct unemployed workers by posting (and committing to) dynamic long-term contracts. As in Gavazza, Mongey and Violante (2018) and Carrillo-Tudela, Gartner and Kaas (2023), firms may improve their hiring prospects by exerting higher recruiting effort. In our model, firms may simultaneously post two types of contract, open-ended and fixedterm, which differ in terms of recruiting costs, the rate at which workers separate from the firm, and the ability of the worker to accumulate human capital on the job. We place a large emphasis on firm heterogeneity. Firms differ ex-ante in a permanent technology type and ex-post in persistent and transitory productivity, in the number of workers, and in the skill composition of their workforce. A firm's technology type determines the permanent component of firm productivity, as well as the relative productivity of workers of different human capital levels. Human capital is accumulated within the firm and is firm-specific. These two properties serve to give a value to worker stability (better achieved through OECs) while keeping a tractable framework where unemployed workers remain identical. In addition, there is firm exit and entry. Firms exit the economy if they are hit by a destruction shock, or if they lose or fire their last remaining worker, which generates endogenous firm selection. A free-entry condition pins down the aggregate measure of operating firms. In

¹In 2019, the share of workers under an FTC was 26.3% in Spain, the second highest among OECD countries, just behind the 27.0% in Chile and ahead of the 24.4% in South Korea, 21.8% in Poland, 20.3% in Portugal, or 20.3% in Netherlands (see https://data.oecd.org/emp/temporary-employment.htm).

equilibrium, unemployed workers remain ex-ante indifferent between applying for a job under either contract, as well as between the firms that offer them, because less ex-post attractive offers are posted in tighter markets. Firms wishing to expand quickly can either offer higher value to workers (attracting longer queues), exert higher recruiting effort, or both. These choices, when aggregated across firms, lead to endogenously different matching efficiencies across OE and FT labor markets, which is key to fit the data on aggregate labor market flows by contract type.

We calibrate the model to match cross-sectional features of our firm-level data as well as to the aggregate flows into and out of unemployment by contract type. The calibration delivers two key results: (i) high-skilled workers are more productive in high-type firms, implying positive assortative matching between firms and workers; and (ii) recruiting costs are higher for OECs, implying that, for a given market tightness, there are on average more matches per unit of time in the FT market than in the OE market. These two calibration results explain the main empirical facts that we documented in the data. First, because of (i), high-type firms use OECs more extensively to retain a larger share of their workers and enhance their chances of accumulating human capital on the job. In the cross-section of firms, this generates the inverse between-firm relationship between firm size and the temporary share that we see in the data. Second, because of (ii), as firms of given productivity grow toward their optimal size, they accumulate relatively more FT workers, which generates the positive within-firm relationship between firm size and the temporary share that we see in the data. This is because firms face a trade-off between the lower costs of attracting workers to FTCs (due to the higher job-filling rates for given labor market tightness) and the higher worker turnover of FT positions. This trade-off is solved differently by different firms: because of decreasing returns to scale in production, the larger the firm is for a given productivity level, the lower its opportunity cost of leaving a vacancy unfilled. This implies that firms choose a higher temporary share as they grow toward their optimal size.

The calibrated economy exhibits two additional features that will be consequential for our counterfactuals. First, labor market frictions imply a significant degree of misallocation of workers between firms, with output losses equal to 3.9% relative to an economy where the marginal product of each type of worker is equalized across all existing firms. This happens because there is dispersion of employment across identical firms and because the most productive firms do not get enough employment and operate with a suboptimal level of worker skills. Second, there is substantial firm selection: the share of highly productive firms is 26% among entrants and 40% among incumbents. This selection happens through the higher exit rate of the less productive firm types, which on average operate with lower levels of employment and higher shares of temporary workers.²

In the last part of the paper, we use the model for counterfactual analysis with the goal to assess the effects of dual labor markets on the aggregate economy. We do so by comparing the effects of different policy tools aimed at reducing the share of workers employed under FTCs. In our economy, changes in the menu of contracts change the hiring and firing behavior of firms, shaping firm growth, firm entry and exit, and the equilibrium distribution of firms and workers.³

²In an extension, we study a version of our economy that generates selection upon entry.

³This is consistent with recent empirical evidence showing that changes in various labor market policies may affect the equilibrium distribution of workers and firms. For instance, Dustmann, Lindner, Schönberg, Umkehrer and vom Berge (2021), Luca and Luca (2019), and Chava, Oettl and Singh (2019) show that changes in minimum wages in Germany or the U.S. reallocate workers to larger and more productive firms, or increase the exit rates of smaller and less productive firms. Likewise, Carry (2022) shows that changes in the minimum workweek regulation in France led to distributional effects among both firms and workers.

In the last part of the paper, we use the model for counterfactual analysis with the goal to assess the effects of dual labor markets on the aggregate economy. We do so by comparing the effects of different policy tools aimed at reducing the share of workers employed under FTCs. In our economy, changes in the menu of contracts change the hiring and firing behavior of firms, shaping firm growth, firm entry and exit, and the equilibrium distribution of firms and workers.³

We start by changing the maximum duration of FTCs. This is a key characteristic of dual labor markets, as in all countries there is a legal (and relatively short) maximum duration that a worker can stay at the firm under an FTC. Reducing the maximum duration is also a common policy tool for countries that want to limit the use of FTCs (i.e., Spain in its 2022 labor market reforms). The effects of a reduction on the maximum duration of FTCs are *ex ante* ambiguous. On the one hand, reducing the maximum duration of FTCs brings flexibility gains, as firms are less likely to pay firing costs when they get hit by negative productivity shocks. On the other hand, it brings turnover losses, as firms are more likely to pay hiring costs to keep their employment level constant. Given the short duration of FTCs in Spain, in our calibrated economy the second effect dominates, making FTCs less attractive to firms when their maximum duration is reduced. Firms react by hiring proportionately more from the OE market (through higher recruiting effort and the offer of better contracts) and increasing the promotion rate of their incumbent FT workers. This leads to a sharp decline in the aggregate temporary share. However, the policy comes with the cost of an increase in unemployment because there is a decline in the unemployment-to-employment flow that more than offsets the mild decrease in the employment-to-unemployment flow.

Moreover, the policy leads to a small increase in aggregate productivity. We show, through a decomposition, that this is because the productivity losses that stem from increased misallocation of workers across firms (i.e., from firms having less flexibility to adjust employment to shocks as they use fewer FTCs) are slightly lower than the gains due to (i) better firm selection (i.e. worse firm types being relatively more damaged and exiting more), and (ii) increased human capital accumulation from longer worker tenure in the firm. All in all, aggregate output and welfare decline.

As an alternative policy, we introduce a tax on the use of FTCs, mirroring several policies in place in France, Portugal, and Spain. When calibrated to produce the same reduction in the aggregate share of FTCs as our baseline policy of shortening FTC duration, the effect of the tax on aggregate productivity is quantitatively similar. However, the negative effects on unemployment, output, and welfare are less severe, mostly due to the effects on job destruction being less severe. As a result, the tax policy leads to lower worker turnover, lower unemployment, and higher output and welfare than placing limits on the duration of FTCs. Finally, we show that an outright ban on FTCs would be suboptimal: even though FTCs are an impediment to human capital accumulation within the firm, they provide flexibility to firms who need to readjust employment in response to adverse shocks. In net, welfare would decrease slightly (by 0.31%) if FTCs were banned completely.

In sum, these various policy exercises show that, through the lens of our calibrated model, restricting the use of FTCs does not typically lead to productivity losses as commonly thought: although the allocation of workers across firms worsens, this is always offset by a better selection of firms in equilibrium. However, restricting the use of FTCs is still a bad idea, because it leads to an increase in unemployment and a fall in total output and welfare.

Related literature There is a large literature studying the effects of the duality of employment contracts on the labor market outcomes of workers, the average unemployment rate, the volatility

of employment, and the dynamics of labor market flows. Blanchard and Landier (2002), Cahuc and Postel-Vinay (2002), Bentolila, Cahuc, Dolado and Le Barbanchon (2012), and Sala, Silva and Toledo (2012) study the effect of dual labor markets in models with search and matching frictions à *la* Mortensen and Pissarides (1994). In these models, search is random and firms do not choose which type of contracts to offer. If they did, they would hire all new workers in FTCs as in Costain, Jimeno and Thomas (2010) because, from the firm's side, the flexibility of FTCs dominates OECs.⁴ Hence, these models are not designed to understand the differential choices of FTCs vs OECs across firms. Furthermore, because firms can only hire one worker, they cannot be used to link contract choices to firm dynamics. Our paper contributes to this literature on both of these fronts.

Several papers provide arguments for the coexistence of FTCs and OECs. In a similar framework as the papers above, Cahuc, Charlot and Malherbet (2016) allow for firms to be heterogeneous in their expected job duration and to choose the type of contract associated to their vacancies. Firms prefer to use FTCs for jobs of short expected duration (in order to save on firing costs) and OECs for jobs of long expected duration (to save on vacancy posting costs). In the context of directed search models, Berton and Garibaldi (2012) argue that an advantage of OECs over FTCs for firms is that the vacancy-filling rate will be higher in equilibrium when posting OECs as more job-seekers will self-select into the OE market, which offers them higher-value jobs ex-post. Our model features a similar equilibrium logic. However, by properly parameterizing the hiring cost function of the OE and FT markets, we can obtain vacancy-filling rates that are larger in the FT than in the OE markets, a feature that is needed to match the large gap in labor market flows between contract types in the data. Other explanations for the coexistence of OECs and FTCs are that duality diminishes on-the-job search and hence allows firms to overcome financial constrains (as in Caggese and Cuñat (2008)).

We also relate to a macro literature studying the equilibrium dynamics of multi-worker firms in the context of frictional labor markets. We use a directed search framework with dynamic long-term contracts in the spirit of Kaas and Kircher (2015) and Schaal (2017). We adapt this framework to a continuous-time setting with slow-moving state transitions similar to Roldan-Blanco and Gilbukh (2021), and extend it to incorporate segmented labor markets, different ex-ante firm types, and double-sided ex-post heterogeneity. An alternative approach in the literature has been to assume random search and study firm dynamics in the context of decreasing returns and Nash bargaining (as in Elsby and Michaels (2013) or Acemoglu and Hawkins (2014)), or in settings with on-the-job search and a variety of wage-setting protocols (e.g. Moscarini and Postel-Vinay (2013), Coles and Mortensen (2016), Bilal, Engbom, Mongey and Violante (2022), Gouin-Bonenfant (2022), Elsby and Gottfries (2022), Audoly (2023), and Gulyas (2023)). These models have proved very successful for non-dual markets. Our contribution to this literature is to provide a quantitative model for the firm and aggregate labor market dynamics of tiered markets.

On the empirical side, several papers have studied the effects of employment protection legislation in the context of dual labor markets. For example, Daruich, Di Addario and Saggio (2023) show that relaxing constraints on FTCs relative to OECs in Italy failed to increase overall employment, while Cahuc, Carry, Malherbet and Martins (2022) show that a policy intended to restrict the use of FTCs by new establishments of large firms in Portugal did not increase the number of permanent

⁴Still, the endogenous firing decision in these models allows to make separations contingent on contract type and hence the aggregate share of FTCs is not undetermined in equilibrium.

contracts and ended up decreasing employment in large firms. We complement these studies by arguing that changing the duration of FTCs yields significant effects on aggregate productivity and employment that mask various selection and reallocation forces.

Finally, even though dual labor markets are typically associated to European economies due to specific legislation, recent papers have documented a *de facto* duality in the U.S. labor market as well (e.g. Gregory, Menzio and Wiczer (2022) and Ahn, Hobijn and Şahin (2023)). Furthermore, Gulyas (2023) documents a similar between-firm and within-firm variation of firm size and the share of low-wage workers in Germany, a country with an extremely low share of temporary employment. Given this, we view our framework as potentially useful for a wider set of economies.

Outline The rest of the paper is organized as follows. Section 2 describes the data and our main empirical findings. Section 3 present the model and Section 4 discusses the estimation of its parameters. Section 5 describes our various policy experiments. Section 6 offers concluding remarks. Proofs and additional results, tables, and figures can be found in the Online Appendix.

2 Empirical Findings

Data We use annual data for Spain from the *Central de Balances Integrada* (CBI) dataset, a comprehensive and unbalanced panel of confidential firm-level balance-sheet data compiled and processed by the *Central de Balances*, a department within Banco de España.⁵ This dataset covers the quasi-universe of Spanish firms, including large and small firms as well as privately held and publicly traded firms. Among many other items from the balance sheet of firms, the data provide information on total employment and the type of employment contract. We use data for the period 2004-2019. We restrict our sample to firms observed for at least 5 years and whose average employment over the

Share of	Distribution of firm-level share of temporary workers						
firms (%)	Mean	p10	p25	p50	p75	p90	p95
100	0.181	0.000	0.000	0.027	0.294	0.591	0.800
77.65	0.164	0.000	0.000	0.000	0.250	0.541	0.776
19.04	0.250	0.000	0.031	0.163	0.391	0.677	0.825
1.78	0.255	0.000	0.034	0.160	0.393	0.701	0.861
0.99	0.237	0.000	0.029	0.147	0.361	0.645	0.833
0.30	0.222	0.000	0.026	0.137	0.329	0.589	0.796
	Share of firms (%) 100 77.65 19.04 1.78 0.99 0.30	Share of firms (%) Mean 100 0.181 77.65 0.164 19.04 0.250 1.78 0.255 0.99 0.237 0.30 0.222	Share of firms (%) Mean Distribut p10 100 0.181 0.000 77.65 0.164 0.000 19.04 0.250 0.000 1.78 0.255 0.000 0.99 0.237 0.000 0.30 0.222 0.000	Share of firms (%) Mean Distribution of firm-1 p10 p25 100 0.181 0.000 0.000 77.65 0.164 0.000 0.000 19.04 0.250 0.000 0.031 1.78 0.255 0.000 0.034 0.99 0.237 0.000 0.029 0.30 0.222 0.000 0.026	Share of firms (%) Mean Distribution of firm-level share of p10 p25 p50 100 0.181 0.000 0.000 0.027 77.65 0.164 0.000 0.000 0.000 19.04 0.250 0.000 0.031 0.163 1.78 0.255 0.000 0.034 0.160 0.99 0.237 0.000 0.029 0.147 0.30 0.222 0.000 0.026 0.137	Share of firms (%) Mean Distribution of firm-level share of temporary p10 p25 p50 p75 100 0.181 0.000 0.000 0.027 0.294 77.65 0.164 0.000 0.000 0.000 0.250 19.04 0.250 0.000 0.031 0.163 0.391 1.78 0.255 0.000 0.034 0.160 0.393 0.99 0.237 0.000 0.029 0.147 0.361 0.30 0.222 0.000 0.026 0.137 0.329	Share of firms (%) Mean Distribution of firm-level share of temporary workers p10 p25 p50 p75 p90 100 0.181 0.000 0.000 0.027 0.294 0.591 77.65 0.164 0.000 0.000 0.000 0.250 0.541 19.04 0.250 0.000 0.031 0.163 0.391 0.677 1.78 0.255 0.000 0.029 0.147 0.361 0.645 0.30 0.222 0.000 0.026 0.137 0.329 0.589

Table 1: Temporary share, descriptive statistics

Note: Selected moments of the distribution of the share of workers under temporary contracts. Each row corresponds to one firm size category, each column to a different moment of the distribution. p10 to p95 refer to percentiles 5 to 95 of the distribution. CBI data pooled over all years.

period is at least one worker. After some cleaning, we keep data for 7,153,669 firm-year observations, corresponding to 705,879 different firms. Remarkably, the average share of FTCs in our sample aggregates very well to the aggregate temporary share from the labor force survey data (see Figure E.1 in the Online Appendix E). Our companion paper, Auciello-Estévez, Pijoan-Mas, Roldan-Blanco and Tagliati (2023), describes the data in more detail and expands on some of the empirical results presented below.

⁵See Almunia, López-Rodríguez and Moral-Benito (2018) for details.

Cross-sectional distribution In our sample, the average share of temporary workers across firms is 18.1%, while the median share is 2.7% (see first row in Table 1). This gap reflects a highly right-skewed distribution. Part of this skewness is due to a large incidence of very small firms (1 or 2 workers) with no temporary workers. However, when we look at the temporary share within firm size bins, the highly right-skewed distribution is still apparent (see rows 2 to 6 in Table 1). For instance, within the subset of firms between 11 and 50 workers, the average share of temporary workers is 25.0%, and the share of temporary workers at the 25th, 50th and 90th percentiles is 3.1%, 16.3%, and 67.7%, respectively. Finally, in Table 1 we also observe that the use of temporary workers tends to increase with firm size —a feature that we will discuss in more detail later— and that most of Spanish firms (more than 95%) have 50 workers or less.

Aggregate determinants Next, in order to understand the heterogeneity in the use of temporary contracts across firms, we propose a regression of the type:

$$TempSh_{ft} = (\alpha_i + \alpha_p + \alpha_t) + \alpha_f + \mathbf{X}_{ft}\boldsymbol{\beta} + \boldsymbol{\epsilon}_{ft}$$
(1)

where $TempSh_{ft}$ is the share of temporary workers of firm f at time t; α_i , α_p , and α_t , are 4-digit industry, province, and year fixed effects; α_f are firm fixed effects; and \mathbf{X}_{ft} is a vector of covariates.⁶

We find that the aggregate fixed effects (α_i , α_p , α_t) are important. The temporary share differs widely across sectors (ranging from 7.7% in "*Real estate activities*" to 43.1% in "*Employment ac*-



Figure 1: Temporary share, by firm size: within- and between-firm variation.

Notes: The green line reports the coefficients of the size dummies of a regression of temporary share that controls for aggregate and firm-level fixed effects. The red line reports the firm fixed effects of the same regression against dummies of average firm size. The blue line reports the size dummies of a regression of temporary share that controls for aggregate but not firm-level fixed effects.

tivities"), across provinces (ranging between 11.5% in Barcelona to 38.8% in Huelva), and over time (the temporary share is strongly procyclical, ranging from 23.9% in 2006 to 15.8% in 2012). However, the R^2 of regression (1) with only the aggregate fixed effects (α_i , α_p , α_t) is 16%. That is, 84% of the variation in the usage of FTCs remains within industry, province, and time period. In

⁶We index each data point by ft instead of ipft since there is no variation in industry i and (almost) no variation in province p at the firm level.

particular, firm fixed effects explain nearly half of the overall variation: the R^2 of regression (1) with aggregate fixed effects increases from 16% to 62% with the inclusion of α_f into the specification.

Temporary contracts and firm size To uncover the firm-level determinants of temporary employment, we run regression (1) with size-bin dummies (2-5 employees, 6-10, 11-20, 21-30, ...) in the vector \mathbf{X}_{ft} together with the aggregate and firm-level fixed effects. As shown in Figure 1, there is a complex relationship between the use of temporary contracts and firm size. The green line represents the estimated β coefficients in the regression with both aggregate and firm fixed effects included, and it captures an increasing within-firm variation between the temporary share and firm size. The red line plots the estimated α_f against the (time-series) average of employment of each firm, and it captures a declining between-firm variation between the temporary share and firm size. Both lines are scaled to deliver the same value for the size bin of 2-5 workers. For comparison, the blue line represents the estimated β coefficients in a regression that does not include firm fixed effect, and hence it does not distinguish between within- and between-firm variation. The within-firm variation suggests that firms make use of FTCs to grow or decline in size: at the firm level, FTCs are useful to help employment track productivity or demand changes. The between-firm variation suggests that there are important technology differences across firms, whereby firms that are (permanently) larger prefer a lower share of temporary contracts, while firms that are (permanently) smaller prefer a higher share of temporary contracts.

Taking stock Our empirical analysis shows that most variation in the temporary share across firms is explained by firm-specific factors, and not by industry, province, or year effects. Firms use temporary contracts when they grow or decline (within-firm, the temporary share *increases* with employment), but larger firms make less use of temporary contracts (between-firm, the temporary share *declines* with employment). With these results in mind, in the next section we write a firm-dynamics model with a dual labor market structure which we calibrate to replicate these and other features of the data to study the implications of labor market duality for the aggregate economy.

3 Model

3.1 Environment

Time is continuous and infinite. We consider a stationary economy that is populated by a mass of workers with fixed unit measure and an endogenous measure of firms. Both firms and workers are risk-neutral and infinitely-lived, and share a common time discount rate, $\rho > 0.^7$

Technology A firm has a type $\varphi \in \Phi \equiv \{\varphi_1, \dots, \varphi_{k_{\varphi}}\}$ that is kept fixed throughout its life. There are two possible worker skill levels, high (*H*) and low (*L*). Let $n_j = 0, 1, 2, \dots$ denote the number of workers with skill $j \in \mathcal{J} \equiv \{H, L\}$ in a firm. All firms operate the following decreasing returns to scale production function:

⁷Throughout the paper, our convention on notation is the following: for some generic object, we reserve a roman font A, or Greek letters α , for generic variables and deep parameters; the calligraphic font A is for (countable) sets; the arrow notation \vec{a} is for vectors; the bold font **A** is for value functions; and the typeset font **A** is for measures of agents.

⁸The elasticity of substitution is $\frac{1}{1-\alpha} > 0$. Skill types are complements in production if $\alpha < \nu$. In this case, the production function is supermodular, i.e. $\frac{\partial^2 Y}{\partial n_H \partial n_L} > 0$, $\forall z, \varphi$.

⁹The transition rates satisfy, $\forall z \in \mathcal{Z}$: $\lambda(z|z) \leq 0$; $\lambda(z'|z) \geq 0$, $\forall z' \neq z$; $\sum_{z' \in \mathcal{Z}} \lambda(z'|z) = 0$; and $\sum_{z' \in \mathcal{Z}} \lambda(z'|z) < +\infty$.

$$y(n_H, n_L, z, \varphi) = e^{z + \zeta(\varphi)} \left(\omega(\varphi) n_H^{\alpha} + (1 - \omega(\varphi)) n_L^{\alpha} \right)^{\frac{1}{\alpha}},$$
(2)

where $\nu \in (0, 1)$ governs the degree of decreasing returns to scale, $\alpha < 1$ governs the elasticity of substitution of workers of different skill levels,⁸ $\omega(\varphi) \in (0, 1)$ measures the relative productivity of high-skilled workers for a firm of permanent type $\varphi \in \Phi$, and $\zeta(\varphi) > 0$ denotes the permanent component of firm productivity for a firm of type $\varphi \in \Phi$. The random variable $z \in \mathbb{Z} \equiv \{z_1 < \cdots < z_{k_z}\}$ stands for the idiosyncratic and transitory productivity of the firm, which follows a continuous-time Markov chain with intensity rates $\lambda(z'|z)$.⁹

Firms hire workers by posting two types of contract, *fixed-term* (FT) and *open-ended* (OE), indexed by $i \in \mathcal{I} \equiv \{FT, OE\}$. A worker is therefore uniquely identified by its type, $(i, j) \in \mathcal{I} \times \mathcal{J}$.

Skill accumulation All starting jobs are in low-skilled positions and, as in Ljungqvist and Sargent (2007), workers can access high-skilled positions through skill upgrades that arrive on the job. The skills are lost upon displacement. These assumptions make human capital firm-specific.¹⁰ We assume that only OE workers can upgrade to high-skilled jobs. In particular, a low-skill OE worker transitions to a high-skill job with intensity $\tau > 0$ (a parameter), with the opposite transition (skill obsolescence) being impossible. This assumption captures the idea that workers under FTCs do not have time to accumulate skills due to their short tenure at the firm.¹¹ An alternative interpretation is that worker training by the firm is much more likely to be offered to workers under OECs than under FTCs due to the higher turnover of the latter type of worker.¹² This gives an important motive to offer OECs to workers: it is the only way to ensure that a fraction of workers know the functioning of the firm and can be entrusted with tasks requiring a higher skill.

Worker transitions Firms may choose to promote their FT workers into an OEC, with the opposite transition being illegal. Firms choose a promotion rate p for each one of their FT workers (if any), which carries a promotion cost of $\xi n_{FT} p^{\vartheta}$ units of the firm's output, with $\xi > 0$ and $\vartheta > 1$. When an FT worker is promoted into an OEC, her job description does not change (i.e. FT workers remain low-skill upon promotion). One can think of the promotion costs as administrative or legal costs for contract conversion, as extra screening cost before conversion, or even as training costs (given that all converted FT workers are exposed to a future skill upgrade).

Firms may lose workers for three different reasons: (i) because of an exogenous firm exit shock, with intensity $s^F \ge 0$, dissolving the firm entirely and sending all of its workers into unemployment; (ii) because the contract expires, at rate $s_i^W \ge 0$ for each contract type $i \in \mathcal{I}$; or (iii) because the firm endogenously decides to fire workers. For the latter case, the firm must choose a per-worker

⁸The elasticity of substitution is $\frac{1}{1-\alpha} > 0$. Skill types are complements in production if $\alpha < \nu$. In this case, the production function is supermodular, i.e. $\partial^2 Y / \partial n_H \partial n_L > 0$, $\forall z, \varphi$. ⁹The transition rates satisfy, $\forall z \in \mathcal{Z}: \lambda(z|z) \leq 0$; $\lambda(z'|z) \geq 0$, $\forall z' \neq z$; $\sum_{z' \in \mathcal{Z}} \lambda(z'|z) = 0$; and $\sum_{z' \in \mathcal{Z}} \lambda(z'|z) < +\infty$.

⁹The transition rates satisfy, $\forall z \in \mathbb{Z}$: $\lambda(z|z) \leq 0$; $\lambda(z'|z) \geq 0$, $\forall z' \neq z$; $\sum_{z' \in \mathbb{Z}} \lambda(z'|z) = 0$; and $\sum_{z' \in \mathbb{Z}} \lambda(z'|z) < +\infty$. ¹⁰Firm-specific human capital allows to simplify the model because it makes all unemployed individuals identical. This assumption is supported by an empirical literature highlighting the importance of firm-specific human capital for worker wage growth (see e.g. Topel (1991), Dustmann and Meghir (2005), or Buchinsky, Fougère, Kramarz and Tchernis (2010)).

¹¹In our calibrated model, a newly hired FT worker only has an average of half a year before its contract is terminated, which is little time to acquire firm-specific skills. If we allowed workers under FTCs to accumulate human capital, in equilibrium only a tiny fraction of them would do so, and this would come at a high cost in terms of extra state space.

¹²Indeed, several papers show that having a temporary contract diminishes the probability of receiving on-the-job training (e.g. Alba-Ramirez (1994), Dolado, Felgueroso and Jimeno (2000), Bratti, Conti and Sulis (2021), Cabrales, Dolado and Mora (2017)). Additional evidence of this mechanism is that wage returns to experience are much larger for experience years accumulated under OECs than FTCs (see e.g. Garcia-Louzao, Hospido and Ruggieri (2023)).

firing rate $\delta_{ij} \geq 0$ for each worker type $(i, j) \in \mathcal{I} \times \mathcal{J}$, which carries a layoff cost equal to $\chi n_{ij} \delta_{ij}^{\psi}$ units of the firm's output, with $\chi > 0$ and $\psi > 1$.¹³ This cost is meant to capture expenses associated to laying off workers, such as administrative expenses and legal costs.¹⁴

Potential entrants In the event that a firm loses all of its workers, it exits the market and becomes a so-called *potential entrant*. In order to post a contract, potential entrants must incur a flow cost $\kappa > 0$ and, upon entry, draw a permanent type $\varphi \in \Phi$ and an initial idiosyncratic productivity $z \in \mathcal{Z}$ from some π_{φ} and π_z distributions, respectively.¹⁵ Firms enter with one worker, whether it is under an OEC or an FTC, which they attract with the same search-and-matching technology as that of operating firms. We describe this technology next.

Search and matching Search is directed. Every instant of time, a firm (i) opens one vacancy of each type $i \in \{FT, OE\}$, (ii) chooses the terms of the offered contracts, and (iii) chooses the recruiting intensity rates (v_{OE}, v_{FT}) for each vacancy.¹⁶ We assume that a recruiting intensity $v_i > 0$ entails a recruiting cost of $A_i v_i^{\zeta}$ units of the firm's output, where $A_i > 0$ and $\zeta > 1$ are parameters. The recruiting cost shifter $A_i > 0$ is potentially contract-specific.¹⁷

Denote the employment vector of a firm by $\vec{n} \equiv (n_{OEH}, n_{OEL}, n_{FT}) \in \mathcal{N}$, where we write \mathcal{N} for the set of all triplets of integers excluding the zero vector, i.e. $\{(0,0,0)\} \notin \mathcal{N}$. Let $(\vec{n}_t^{t+s}, z_t^{t+s})$ be the full history of possible firm states between dates t and t + s. At any t and for all s > 0, a contract of type $i \in \mathcal{I}$ for skill type $j \in \mathcal{J}$ offered by a firm of permanent type $\varphi \in \Phi$ is a set of complete state-dependent sequences of wages $w_{ij}(\vec{n}_t^{t+s}, z_t^{t+s}; \varphi)$, recruiting intensities $v_i(\vec{n}_t^{t+s}, z_t^{t+s}; \varphi)$, firing rates $\delta_{ij}(\vec{n}_t^{t+s}, z_t^{t+s}; \varphi)$ and, only for workers employed under FTCs, intensities $p(\vec{n}_t^{t+s}, z_t^{t+s}; \varphi)$ of promotion into an OEC, conditional on no worker separation and firm survival.

We assume the following commitment structure. On the worker's side, workers may forfeit their contract and quit the firm, but in that case they must go back to unemployment (and consume flow utility b > 0), from where they can regain employment. On the firm's side, by contrast, there is full commitment to both the contract type as well as to the contractual terms, which cannot be revised or renegotiated for the duration of the match. Therefore, contracts must always comply with the firm's initial promises. Moreover, we assume that the firm cannot discriminate between workers with the same contract type and job description, i.e. all n_{ij} workers of type (i, j) obtain the same contract (though, of course, their individual employment histories may differ).

Given these assumptions, a distinct labor market segment (or "submarket") is indexed by (i) the contract's type $i \in \{FT, OE\}$, and (ii) the long-term value that the worker can expect to obtain

¹³Notice that, mirroring Spanish law, we do not allow firing cost parameters to depend explicitly on contract type, implying that differences in firing costs across firms will be an endogenous outcome of the model.

¹⁴The cost, however, does not include pure transfers between the employer and the worker (severance payments). A severance payment in this model would be a pure lump-sum transfer which would have no affect on the equilibrium allocation, as the optimal contract maximizes the joint surplus of the firm and all of its workers.

¹⁵Note that this implies that entrants cannot choose their type. See Section 5.1.4 for a model extension where they do. ¹⁶This implies that firms wishing to grow fast must increase the filling rate of vacancies as they cannot do it by raising the number of vacancies. We make this assumption for tractability, but Gavazza *et al.* (2018) and Carrillo-Tudela *et al.* (2023) show that, in practice, the vacancy yield is much more important for firm growth than the vacancy rate.

¹⁷This could be due to several reasons. For instance, there might be a tougher screening process for incoming OE workers, as these workers will eventually be assigned to high-skilled tasks, which may require more talent or specific knowledge, making hiring under this contract more costly for the same recruiting intensity. Another interpretation is that the value of properly screening candidates is larger for OE workers, who expected to stay longer with the firm. Finally, in Spain there are intermediary companies (*empresas de trabajo temporal*) connecting workers seeking and firms offering FTCs, easing the matching process and lowering the costs of recruiting FT workers.

from it, denoted W. Each worker can simultaneously search in at most one submarket (i, W), and each firm can simultaneously post only one offer W of each contract type i. We denote by $V_i(W)$ the intensity-weighted measure of vacancies in submarket (i, W).¹⁸ Likewise, we denote by $U_i(W)$ the measure of unemployed workers applying to the same submarket (i, W). Then, the measure of matches in market segment (i, W) is equal to $M(V_i(W), U_i(W))$, where $M : \mathbb{R}^2_+ \to \mathbb{R}_+$ is a constant-returns-to-scale matching function whose parameters are common across all submarkets.

As there is a continuum of firms within each submarket and each firm faces the same hiring frictions, by the law of large numbers an individual firm exerting recruiting effort v in submarket (i, W) obtains $v\eta(\theta_i(W))$ new hires, where $\theta_i(W) \equiv V_i(W)/U_i(W)$ is the effective market tightness and $\eta: \theta \mapsto M(1, \theta^{-1})$ is the aggregate job-filling rate per effective vacancy. Notice $\eta(\theta)$ is constant across all firms facing the same market tightness θ . Therefore, an individual firm's job-filling rate is a function of this aggregate meeting rate η , which is determined in equilibrium but which the firm takes as given, as well as the recruiting effort v, which is a choice variable for the firm. Similarly, we denote by $\mu(\theta) = \theta \eta(\theta)$ the aggregate job-finding rate in a submarket with tightness θ .¹⁹

Recursive contracts Because contracts are large and complex objects, we focus on their recursive formulation. We focus on a Markov Perfect Equilibrium, in which contracts are only functions of the firm's state. A firm's state is (i) its permanent type $\varphi \in \Phi$, (ii) its employment vector and productivity $(\vec{n}, z) \in \mathcal{N} \times \mathcal{Z}$, and (iii) the set $\vec{W} = (W_{OEH}, W_{OEL}, W_{FT}) \in \mathbb{R}^3_+$ of outstanding values that the firm promised to its incumbent workers. Then, a recursive contract is defined by:²⁰

$$ec{C}_{ij} \equiv \left\{ w_{ij}, v_i, \delta_{ij}, p, W'_{ij}(ec{n}', z')
ight\}.$$

For each worker $(i, j) \in \mathcal{I} \times \mathcal{J}$, a contract includes a wage w_{ij} , a recruiting effort v_i , a per-worker layoff rate δ_{ii} , a promotion rate *p* for FT workers, and a continuation promise $W'_{ii}(\vec{n}', z')$ for each new possible set of states (\vec{n}', z') of the firm. The new state vector, conditional on firm survival, is

$$(\vec{n}',z') \in \left\{ \begin{array}{l} (n_{OEH}, n_{OEL}+1, n_{FT}, z), (n_{OEH}, n_{OEL}, n_{FT}+1, z), \\ (n_{OEH}+1, n_{OEL}-1, n_{FT}, z) \\ (n_{OEH}-1, n_{OEL}, n_{FT}, z), (n_{OEH}, n_{OEL}-1, n_{FT}, z), (n_{OEH}, n_{OEL}, n_{FT}-1, z), \\ (n_{OEH}, n_{OEL}+1, n_{FT}-1, z), \\ \{(n_{OEH}, n_{OEL}, n_{FT}, z'), : z' \in \mathcal{Z} \} \end{array} \right\},$$

depending on which type of transition the firm has in the next stage (including hiring, promotion, separation, skill upgrades and productivity shocks, respectively). Henceforth, we will make use of the short-hand notation $\vec{n}_{ij}^+ \equiv (n_{ij} + 1, \vec{n}_{-(ij)}), \vec{n}_{ij}^- \equiv (n_{ij} - 1, \vec{n}_{-(ij)}), \vec{n}^p \equiv (n_{OEH}, n_{OEL} + 1, n_{FT} - 1),$ and $\vec{n}^{\tau} \equiv (n_{OEH} + 1, n_{OEL} - 1, n_{FT})$, to denote the various size transitions, where $\vec{n}_{-(ij)} \equiv \vec{n} \setminus \{n_{ij}\}$, for any $i \in \{FT, OE\}$ and $j \in \{H, L\}$.²¹

¹⁸Precisely, the effective measure of vacancies in submarket (i, W) is defined as $V_i(W) \equiv \int_{\Omega_i(W)} v_{if} df$, where v_{if} is the recruiting intensity of firm f for contract $i \in \{FT, OE\}$, and $\Omega_i(W)$ is the set of firms that offer value W for contract i. See further details in Online Appendix A.5.

¹⁹These rates satisfy standard properties: $\eta(\theta)$ is decreasing and convex, $\mu(\theta)$ is increasing and concave, and the usual Inada conditions apply, namely $\lim_{\theta \to 0} \mu(\theta) = \lim_{\theta \to +\infty} \eta(\theta) = 0$, and $\lim_{\theta \to +\infty} \mu(\theta) = \lim_{\theta \to 0} \eta(\theta) = +\infty$. ²⁰To alleviate notation, we do not index recursive contracts explicitly by (n, z, φ, \vec{W}) .

²¹As \vec{n} may contain more than one instance of the same element (e.g. the firm has five FE workers and five OEH workers), the symbols \cup and \setminus represent *multiset* union and difference operators, respectively, meaning $\{a, b\} \cup \{b\} = \{a, b, b\}$ instead of $\{a, b\} \cup \{b\} = \{a, b\}$, and $\{a, b, b\} \setminus \{b\} = \{a, b\}$ instead of $\{a, b, b\} \setminus \{b\} = \{a\}$.

3.2 Equilibrium

3.2.1 Unemployed Worker's Problem

Unemployed workers consume a flow utility b > 0 while searching in the labor market. Search is directed toward the submarket (i, W) that offers the most profitable expected return for workers. Thus, the value of unemployment is $\mathbf{U} = \max_{(i,W)} \{U_i(W)\}$, where $U_i(W)$ solves the following Hamilton-Jacobi-Bellman (HJB) equation:

$$\rho U_i(W) = b + \mu \big(\theta_i(W)\big) \max \Big\{ W - U_i(W), 0 \Big\}.$$
(3)

As workers prefer the most profitable offers, when unemployed they must remain indifferent ex-ante between all of those offers which they decide to apply to. Therefore, the following complementary slackness condition must hold:

 $\forall (i, W) \in \mathcal{I} \times \mathbb{R}_+$: $U_i(W) \leq \mathbf{U}$, with equality if, and only if, $\mu(\theta_i(W)) > 0$.

This condition states that a submarket must maximize the value of remaining unemployed, or else it is never visited by workers. Imposing this condition into equation (3) we find:

$$\theta(W) = \mu^{-1} \left(\frac{\rho \mathbf{U} - b}{W - \mathbf{U}} \right).$$
(4)

Equation (4) defines the equilibrium function that maps promised values to market tightness, for any given value of unemployment **U**. Market tightness is decreasing in *W* (more attractive contracts for workers ex-post attract more workers per job posting ex-ante), and increasing in **U** (a better outside option for workers makes jobs relatively less attractive ex-ante).²² Market tightness, however, does not depend explicitly on the contract type *i* because of the indifference condition and the fact that the value of a job is summarized only by *W*. However, in equilibrium, differences in primitives across contracts will lead different firms to offer different values *W* for different contracts i = OE, FT, giving rise to heterogeneity in aggregate job-filling and job-finding rates by contract type.

3.2.2 Joint Surplus Problem

Next, we characterize the optimal contract menu chosen by firms. For each worker of type (i, j) employed by the firm, this requires finding the vector $\vec{C}_{ij} = \{w_{ij}, v_i, \delta_{ij}, p, W'_{ij}(\vec{n}', z')\}$ that maximizes firm value. As the firm has commitments with its pre-existing workers, this problem is subject to a promise-keeping constraint. As workers do not commit, there is also a worker-participation constraint. We write the firm's and employed worker's value functions in Online Appendix A.1.

As it turns out, however, we can find the optimal contract from a simpler and equivalent problem, which involves maximizing the *joint surplus* of the match, that is, the sum of the firm's value and the values of all of its workers. For a match between a firm of type φ in state (\vec{n}, z, \vec{W}) and its workers, we define the joint surplus as:

$$\Sigma(\vec{n}, z, \varphi) \equiv \mathbf{J}(\vec{n}, z, \varphi, \vec{W}) + \sum_{i \in \mathcal{I}} \sum_{j \in \mathcal{J}} n_{ij} W_{ij},$$
(5)

²²To save on notation, we write $\theta(W)$ instead of $\theta(W, \mathbf{U})$, but the reader should bear in mind that **U** is an endogenous object that we solve for in equilibrium, but one which both firms and workers take as given when making decisions.

where $\mathbf{J}(\cdot)$ denotes the value of the firm, and recall that $\vec{W} \equiv \{W_{ij}\}$ are the outstanding promised values. As discussed in detail below, the joint surplus is independent of promised values and, anticipating this result, on the left-hand side of equation (5) we have written $\Sigma(\vec{n}, z, \varphi)$ instead of $\Sigma(\vec{n}, z, \varphi, \vec{W})$. In Online Appendix A.1, we then show that $\Sigma(\vec{n}, z, \varphi)$ solves the HJB equation:²³

$$(\rho + s^{F})\Sigma(\vec{n}, z, \varphi) = \max_{p, \{v_{i}, \delta_{ij}, W_{iL}'(\vec{n}_{iL}^{+}, z)\}} \left\{ \underbrace{\underbrace{S(\vec{n}, z, \varphi)}_{\text{Flow surplus}} + \underbrace{\sum_{i \in \mathcal{I}} \sum_{j \in \mathcal{J}} n_{ij}(\delta_{ij} + s_{i}^{W}) \left(\Sigma(\vec{n}_{ij}^{-}, z, \varphi) - \Sigma(\vec{n}, z, \varphi)\right)}_{\text{Worker separation and firing}} + \underbrace{\sum_{i \in \mathcal{I}} v_{i} \eta \left(W_{iL}'(\vec{n}_{iL}^{+}, z)\right) \left(\Sigma(\vec{n}_{iL}^{+}, z, \varphi) - \Sigma(\vec{n}, z, \varphi)\right)}_{\text{Hiring an FT or OEL worker}} + \underbrace{n_{OEL} \tau \left(\Sigma(\vec{n}^{\tau}, z, \varphi) - \Sigma(\vec{n}, z, \varphi)\right)}_{\text{Skill upgrade of OEL workers}} + \underbrace{\sum_{z' \in \mathcal{Z}} \lambda(z'|z) \left(\Sigma(\vec{n}, z', \varphi) - \Sigma(\vec{n}, z, \varphi)\right)}_{\text{Productivity shock}}\right\}, (6)$$

subject to $W'_{iL}(\vec{n}^+_{iL'}z) \ge \mathbf{U}, \forall i \in \mathcal{I}$, a worker participation constraint that entices workers to remain matched by promising them more utility that their outside option.²⁴ Equation (6) states that the joint surplus is composed of the flow surplus, plus the changes in joint surplus value due to worker separation, firing, hiring, promoting, skill upgrading, and productivity shocks, where:²⁵

$$\mathbf{S}(\vec{n}, z, \varphi) \equiv \underbrace{y(\vec{n}, z, \varphi)}_{\text{Firm's output}} + \underbrace{\sum_{i \in \mathcal{I}} \sum_{j \in \mathcal{J}} n_{ij}(\delta_{ij} + s_i^W + s^F) \mathbf{U}}_{\text{Workers' outside options}} - \underbrace{\sum_{i \in \mathcal{I}} A_i v_i^{\varsigma}}_{\text{Recruiting costs}} - \underbrace{\xi n_{FT} p^{\vartheta}}_{\text{Promotion}} - \underbrace{\sum_{i \in \mathcal{I}} \sum_{j \in \mathcal{J}} \chi n_{ij} \delta_{ij}^{\psi}}_{\text{Firing costs}} - \underbrace{\sum_{i \in \mathcal{I}} v_i \eta \left(W_{iL}'(\vec{n}_{iL}^+, z) \right) W_{iL}'(\vec{n}_{iL}^+, z)}_{\text{Expected value delivered to new hires}}$$
(7)

Equipped with this formulation, we arrive at our main equivalence result (for the proof, see Online Appendix A.1):

Proposition 1 The firm's and joint surplus problems are equivalent.

This result is reminiscent of other directed search models with multi-worker firms and long-term contracts (e.g. Schaal (2017)). As in those models, the firm's choices guarantee that the joint surplus is maximized because the contract space is complete, all agents have linear utilities, and these utilities are transferable. Thus, the set of optimal contracts can be found in two stages. In the first stage, we take first-order conditions of problem (6) to find the optimal recruiting intensity, job-filling rates, firing rates, and promotion rates for each job and each contract. Then, in the second stage, we find the set of promised utilities \vec{W} that implement this allocation. We detail these two stages below.

²⁴Notice that the promise-keeping constraint does not appear as a constraint in the joint surplus problem because definition (5) imposes that it holds with equality in equilibrium. See Online Appendix A.1 for the formal argument.

²³Throughout we will write the job-filling rate as $\eta(W(\cdot)) \equiv \eta(\theta(W(\cdot)))$, with $\theta(W)$ defined in equation (4).

²⁵In words, the flow surplus is the sum of the firm's flow output and worker's outside options in case of separation (first line of (7)), net of four types of costs: recruiting costs, promotion costs, firing costs, and the costs of having to deliver the promised value in case of successful hiring (in expected terms).

Stage 1 Taking first-order conditions of problem (6) for $W'_{iL}(\vec{n}^+_{iL}, z, \varphi)$ yields:

$$\frac{\partial \eta(W)}{\partial W}\Big|_{W_{iL}^{\prime}(\vec{n}_{iL}^{+},z,\varphi)}W_{iL}^{\prime}(\vec{n}_{iL}^{+},z,\varphi) + \eta\left(W_{iL}^{\prime}(\vec{n}_{iL}^{+},z,\varphi)\right) = \left(\Sigma(\vec{n}_{iL}^{+},z,\varphi) - \Sigma(\vec{n},z,\varphi)\right)\frac{\partial \eta(W)}{\partial W}\Big|_{W_{iL}^{\prime}(\vec{n}_{iL}^{+},z,\varphi)},$$
(8)

showing that a firm in state (\vec{n}, z, φ) equates the expected marginal cost (left-hand side) to the expected marginal benefit (right-hand side) of promising value $W'_{iL}(\vec{n}^+_{iL'}z, \varphi)$ to new hires. In particular, the marginal cost is given by the increase in the joint surplus that goes to the new hires and the marginal benefit is given by the joint surplus that the new hires will produce.

The optimal recruiting, firing and promotion rates for a firm in state (\vec{n}, z, φ) satisfy, respectively:

$$v_{i}(\vec{n}, z, \varphi) = \left[\eta \left(W_{iL}'(\vec{n}_{iL}^{+}, z, \varphi) \right) \left(\frac{\Sigma(\vec{n}_{ij}^{-}, z, \varphi) - \Sigma(\vec{n}, z, \varphi) - W_{iL}'(\vec{n}_{iL}^{+}, z, \varphi)}{\varsigma A_{i}} \right) \right]^{\frac{1}{\varsigma - 1}}, \quad (9)$$

$$\delta_{ij}(\vec{n}, z, \varphi) = \left[\frac{\boldsymbol{\Sigma}(\vec{n}_{ij}, z, \varphi) - \boldsymbol{\Sigma}(\vec{n}, z, \varphi) + \mathbf{U}}{\psi \chi}\right]^{\frac{1}{\psi - 1}},$$
(10)

$$p(\vec{n}, z, \varphi) = \left[\frac{\Sigma(\vec{n}^p, z, \varphi) - \Sigma(\vec{n}, z, \varphi)}{\vartheta \xi}\right]^{\frac{1}{\vartheta - 1}}.$$
(11)

In all three cases, firms equate marginal costs to marginal gains, given by the corresponding changes in the joint surplus value. We note that, despite not having a fixed cost in production, there is still endogenous firm exit. This will happen whenever the joint surplus $\Sigma(\vec{n}, z, \varphi)$ of one-worker firms is below the value of unemployment **U** of its worker, in which case δ_{ij} will be positive.²⁶ This situation is more likely to happen for low values of *z* and for less productive firm types φ .

Stage 2 The equilibrium promised values that implement these policies can be found by ensuring that the resulting surplus is distributed across agents in a way that maximizes firm profits while keeping ex-ante promises at every point of the state space, as explained in Online Appendix A.1. In practice, this allows us to construct the whole sequence of promised values through an iterative procedure: the promised value of a firm with size vector $(n_{ij}, \vec{n}_{-(ij)})$ must coincide with the optimal upsize policy (for a type-*ij* worker) of a firm with size vector $(n_{ij} - 1, \vec{n}_{-(ij)})$, as well as with the optimal downsize policy of a firm with size vector $(n_{ij} + 1, \vec{n}_{-(ij)})$.

For the special case of $(n_{ij}, \vec{n}_{-(ij)}) = (1, \vec{0})$, promised values must, in turn, be consistent with the optimal choices of potential entrants. These firms have no workers and perceive a value J^e , with:

$$\rho \mathbf{J}^{e} = -\kappa + \sum_{\varphi \in \Phi} \pi_{\varphi}(\varphi) \, \widetilde{\mathbf{J}}^{e}(\varphi), \tag{12}$$

where $\tilde{\mathbf{J}}^{e}(\varphi)$ is the expected value of entry for firms of type φ defined by:

$$\widetilde{\mathbf{J}}^{e}(\varphi) \equiv \sum_{z \in \mathcal{Z}} \pi_{z}(z) \sum_{i \in \mathcal{I}} \left[\max_{W_{i}} \left\{ \eta(W_{i}) \left(\mathbf{J} \left(\vec{n}_{iL}^{e}, z, \varphi, \{W_{i}\} \right) - \mathbf{J}^{e} \right) \text{ s.t. } W_{i} \ge \mathbf{U} \right\} \right],$$
(13)

where we have used the notation $\vec{n}_{ij}^e \equiv (n_{ij}^e, \vec{n}_{-(ij)}^e) = (1, \vec{0})$. Taking first-order conditions pins down the initial promise by entering firms and, by the iterative procedure outlined above, the

²⁶Note that the joint surplus $\Sigma(\vec{n}_{ij}, z, \varphi)$ of losing the last worker is zero due to the free entry condition, see below.

entire sequence of promised values. We assume free entry into the labor market, i.e. we allow the aggregate measure of firms to freely adjust in equilibrium. Thus, in an equilibrium with positive firm entry, we must have $J^e = 0$, which pins down the average labor market tightness of the economy.

3.2.3 Closing the Model

To close the characterization of the equilibrium, we must find the steady state distribution of firms and workers. The law of motion for the measure of firms in each state, $f_t(\vec{n}, z, \varphi)$, is characterized by a set of Kolmogorov forward equations, which we provide in full in Online Appendix A.2. This appendix also shows how to obtain the measure of operating firms, F, as well as the unemployment rate, U, in a stationary equilibrium in which $\frac{\partial}{\partial t}f_t(\vec{n}, z, \varphi) = 0$.

4 Estimation

This section describes how we bring the model to the data. We start by describing our data sources and several details about the model parameterization. Next, we discuss our calibration strategy (Sections 4.1 and 4.2), and validate it with a global identification exercise and a set of non-targeted moments (Section 4.3). Finally, we explore some features of the calibrated economy, with a special emphasis on the sources of static employment misallocation (Section 4.4).

Data sources We compute moments from the *Central de Balances* (CBI) firm-level data introduced in Section 2. We focus on a sub-sample of firms with at most 60 workers, representing 97.3% of firms from the full sample.²⁷ These data come at the yearly frequency. For our variables of interest (firm size, share of temporary workers, and value added per worker), we regress out the aggregate fixed effects (industry, time, province) to ensure that differences across firms do not reflect these other factors, which are not in the model. We also use aggregate data on worker flows into and out of employment by contract type from *Encuesta de Población Activa* (EPA), the Spanish labor force survey. These data come at a quarterly frequency.

Parameterization First, we set the model period to one quarter to match the EPA time frequency, as the CBI data (at a yearly frequency) is used for stock variables. Second, the dynamics of the productivity shock *z* is represented by the vector of values $\{z_1, \ldots, z_{k_z}\}$ and the $k_z(k_z - 1)$ matrix of intensity rates $\{\lambda(z'|z)\}$. As this is a potentially large number of parameters, we recover them from the discretization of an Ornstein-Uhlenbeck diffusion process for idiosyncratic productivity (in logs):

$$d\ln(z_t) = -\rho_z \ln(z_t) dt + \sigma_z dB_t, \tag{14}$$

where B_t is a Wiener process, and (ρ_z, σ_z) are positive persistence and dispersion parameters.²⁸ Third, we choose a standard Cobb-Douglas specification for the matching function: $M(V, U) = V^{\gamma}U^{1-\gamma}$, where $\gamma \in (0, 1)$ is the matching elasticity, implying meeting rates $\mu(\theta) = \theta^{\gamma}$ for the worker, and

²⁷All of the empirical facts that we established in Section 2 for the full sample also hold for this sub-sample. For instance, we still find that the share of temporary workers increases (respectively, decreases) in firm size when looking at within-firm (respectively, between-firm) variation. See Figure E.2 in the Online Appendix.

²⁸Particularly, we recover the $\{\lambda(z'|z)\}$ intensity rates and $\{z_i\}_{i=1}^{k_z}$ productivity levels from discretizing this process using the Euler-Maruyama and Tauchen (1986) methods (details in Online Appendix D.1). For the entrant firms' productivity distribution π_z we take the ergodic distribution associated with the (calibrated) Markov chain implied by equation (14).

 $\eta(\theta) = \theta^{\gamma-1}$ for the firm. This functional form leads to convenient analytical representations for the promised value and the job-filling rate (see Online Appendix A.3). Finally, to avoid a problem of state space dimensionality, we set $(N_{OEH}, N_{OEL}, N_{FT}) = (30, 15, 15)$, which imposes an upper bound of 60 workers per firm, and we assume $k_{\varphi} = 2$ firm types and $k_z = 5$ productivity levels.²⁹

4.1 Externally Set Parameters

Given the parameterization described above, we have 24 parameters to calibrate. Of these, 9 parameters, namely $\mathbf{p}_{\text{ext}} \equiv (\rho, \gamma, \rho_z, \sigma_z, \zeta, \psi, \vartheta, \tau, \zeta(\varphi_1))$, are set externally (see Table 2). We fix the discount rate to $\rho = 0.0123$, corresponding to an annualized discount rate of $(1 + \rho)^4 - 1 \approx 5\%$. For the matching elasticity, we choose $\gamma = 0.5$, a standard value in the literature (e.g. Petrongolo and Pissarides (2001)).³⁰ The productivity parameters (ρ_z, σ_z) introduced in equation (14) are calibrated to match a yearly autocorrelation of firm-level TFP of 0.81 and a yearly volatility of 0.34. We take these values from Ruiz-García (2021), which estimates an AR(1) process for firm-level TFP using Spanish firm-level balance sheet data from CBI, the same data source that we use in our empirical analysis. As we explain in Online Appendix D.1, these targets imply that $\rho_z = 0.0513$

Table 2: Externally Set Parameters					
Parameter		Value	Target/Source		
ρ	Discount rate	0.0123	5% annual discount rate		
γ	Matching elasticity	0.5000	Petrongolo and Pissarides (2001)		
$ ho_z$	Mean-reversion in productivity	0.0513	Ruiz-García (2021)		
σ_z	Productivity dispersion	0.1833	Ruiz-García (2021)		
ς	Recruiting cost curvature	2.0000	Linear marginal cost of recruiting		
ψ	Firing cost curvature	2.0000	Linear marginal gain of promoting		
θ	Promotion cost curvature	2.0000	Linear marginal gain of firing		
τ	Rate of skill upgrade	0.1250	Baley <i>et al.</i> (2023)		
$\zeta(\varphi_1)$	Permanent productivity φ_1 firms	1.0000	Normalization (without loss)		

Note: Set of externally calibrated parameters. See Section 4.1 for details.

and $\sigma_z = 0.1833$. We set the cost curvature parameters of the recruiting, firing, and promotion technologies to $\zeta = \psi = \vartheta = 2$, so that the recruiting intensity, layoff, and promotion rates are linear in the corresponding net surplus changes.³¹ We set the skill conversion rate to $\tau = 1/8$, such that the average duration before a skill upgrade is 8 quarters. This follows from Baley, Figueiredo, Mantovani and Sepahsalari (2023), who show that the returns to (occupational) experience are concave and almost exhausted after two years. Finally, we normalize the permanent productivity component of type- φ_1 firms to $\zeta(\varphi_1) = 1$, which comes without loss as our economy exhibits size-neutrality —only the *relative* size $\zeta(\varphi_2)/\zeta(\varphi_1)$ matters, and this ratio will be calibrated internally.

²⁹Given these choices, the support of the state space has about 80,000 points and there are about 6 billion potential transitions between states, making the calculation of the invariant distribution of firms computationally challenging. Online Appendix A.2 discusses this issue and how we get around it.

³⁰This value is routinely used in models estimated to U.S. data, but it has also been used for European labor markets, specifically in models of dual labor markets (e.g. Thomas (2006), Costain *et al.* (2010) and Bentolila *et al.* (2012)).

4.2 Internally Set Parameters

The parameters $\mathbf{p}_{\text{int}} \equiv (\kappa, \zeta(\varphi_2), \nu, \omega(\varphi_1), \alpha, \omega(\varphi_2), \chi, A_{OE}, A_{FT}, s_{OE}^W, s_{FT}^W, \xi, \pi_{\varphi}, s^F, b)$, are calibrated internally. As the joint estimation of these 15 parameters and the assignment of firms to types φ is numerically unfeasible, we use the two-step procedure in Bonhomme, Lamadon and Manresa (2022). First, we assign individual firms to types by use of some statistics from the data without explicitly solving the model. Then, we estimate the model parameters by the Simulated Method of Moments conditional on this assignment, using the algorithm described in Online Appendix D.2.³²

4.2.1 Assigning Firms to Permanent Types

The model predicts that firms of different technology types φ differ ex-post in their size, due to $\zeta(\varphi)$, and in their temporary share conditional on firm size, due to $\omega(\varphi)$. In the data, we

$$\left(\vec{M}^{\text{data}} - \vec{M}^{\text{mod}}(\mathbf{p}_{\text{ext}}, \mathbf{p}_{\text{int}})\right)^{\top} \mathcal{W}^{-1}\left(\vec{M}^{\text{data}} - \vec{M}^{\text{mod}}(\mathbf{p}_{\text{ext}}, \mathbf{p}_{\text{int}})\right),\tag{15}$$

where \vec{M}^{data} is a vector of moments from the data (conditional on the assignment of firms to types), $\vec{M}^{\text{mod}}(\mathbf{p}_{\text{ext}}, \mathbf{p}_{\text{int}})$ is the model counterpart, and \mathcal{W} is a diagonal matrix of weights containing the squares of the data moments in the diagonal.

	Tuble of Tublestiment of minub in the data to y types			
	Firm size (average, in <i>#</i> workers)	Temporary share (average, in %)	Share of firms (total, in %)	
Firms of type φ_1	9.8	4.3	19.7	
Firms of type φ_2	6.8	23.2	80.3	
All firms	7.4	19.5	100.0	

Table 3: Assignment of firms in the data to φ types

Note: Results from classifying firms in the data into the two permanent types of the model. See Section 4.2.1 for details.

have documented a between-firm negative correlation between employment and temporary share (Figures 1 and E.2). Therefore, we classify firms into two different technology types with the aim of reproducing this relationship. First, we regress the temporary share against dummies of firm size and unobserved firm fixed effects and we keep the estimated firm fixed effects. These can be thought of as capturing the "permanent temporary share" of each firm. Second, we take the time series average of firm size for each firm. This can be thought of as the "permanent size" of each firm. Third, we group all firms into 50 groups (2% of the population each) based on the "permanent temporary share" and the average "permanent size" in each group. Finally, we run a *k*-means algorithm with these two variables across all 50 groups to create only two groups, which are our two types (φ_1, φ_2).

The results of this procedure are in Table 3. The first type (labeled φ_1), represent a smaller share of firms (19.7%), make a moderate use of temporary contracts (4.3% of the firm's employment, on average), and are larger in size (9.8 employees). The remainder share of firms (80.3%), those of type φ_2 , make ample use of temporary contracts (23.2%), and are smaller (6.8 employees).

³¹We do this for symmetry with the linearity in the hiring policy when $\gamma = 0.5$ (equation (A.27) in the Online Appendix). ³²The parameters included in \mathbf{p}_{int} are chosen to minimize:

4.2.2 Targeted Moments and Model Fit

Once we have classified firms in the data into the model's types, we select various moments to identify our parameters. In this section, we discuss intuitively how each parameter is identified by each moment. Section 4.2.3 will verify these intuitions with a formal identification exercise.

Average and relative firm size We target the average firm size and the relative size of firms of different types, which identifies κ and $\zeta(\varphi_2)/\zeta(\varphi_1)$ respectively. First, average firm size equals the ratio of the employment rate, E, to the measure of operating firms, F. This is mainly affected by the entry cost parameter, κ . To see this, note that changes in κ monotonically shifts the firm's expected value of entry through the free entry condition. In equilibrium, the expected value of entry decreases monotonically with the average labor market tightness F/(1 - E) because it determines the average probability of hiring. The steady state employment rate E is pinned down by the aggregate labor market flows, which are either direct calibration targets or implied by our calibration targets (see details below). Hence, κ determines the mass of operating firms F and through that average firm size, E/F. Our target for average firm size is 7.35 and our economy delivers 7.01 (see Table 4).

Second, the *relative* firm size between φ_2 and φ_1 firm types is driven by the ratio of their permanent productivities, $\zeta(\varphi_2)/\zeta(\varphi_1)$. Because $\zeta(\varphi_1)$ is normalized to 1, this pins down $\zeta(\varphi_2)$. In the data, average firm size of φ_2 firms is a fraction 0.693 of the average size of φ_1 firms. The calibration predicts a relative size of 0.650, and pins down $\zeta(\varphi_2) = 0.8760$ (see Table 4).

Productivity and temporary share by firm characteristics Next, we want the model to be consistent with the observed relationship between (i) firm productivity and firm characteristics, and (ii) temporary share and firm characteristics. These relationships should help identify the parameters in the production function, $(\nu, \alpha, \{\omega(\varphi)\})$, and the costs of firing workers, χ .

To see this, let $n = n_H + n_L$ denote the total number of workers in a firm, our observable measure of size in the data. Using equation (2), we can write log output per worker as:

$$\ln\left(\frac{y(n_H, n_L, z, \varphi)}{n}\right) = z + \zeta(\varphi) - (1 - \nu)\ln(n) + \frac{\nu}{\alpha}\ln\left(\omega(\varphi)\left(\frac{n_H}{n}\right)^{\alpha} + (1 - \omega(\varphi))\left(\frac{n_L}{n}\right)^{\alpha}\right).$$
(16)

If we could observe all the variables of this equation, a non-linear least squares regression could recover the degree of decreasing returns to scale ν from the partial effect of firm size on firm productivity. Moreover, we could recover the relative productivity of high and low skilled workers by firm type, $\omega(\varphi)$, and the elasticity of substitution between the two, $\frac{1}{1-\alpha}$, from the partial effect of changes in the skill composition of the firm on firm productivity. However, in our CBI data we observe neither firm productivity *z* nor the share of high-skilled workers, n_H/n .

To circumvent this issue, we follow an indirect inference strategy and consider a simplified version of the above equation that can be estimated both in the data and in the model. In particular, we compute a second-order expansion of equation (16), we measure output per worker as the ratio of value added (VA_{it}) to employment (Emp_{it}), we proxy the skill rate with the temporary share (as they are strongly negatively correlated in the model but the former is unobserved), and we send the firm temporary TFP component z to the error term. The resulting regression is:

$$\ln\left(\frac{VA_{it}}{Emp_{it}}\right) = \text{constant} + \beta_0^A \mathbf{1} \left[\varphi_i = \varphi_2\right] + \beta_1^A \ln(Emp_{it}) + \beta_2^A TempSh_{it} + \beta_3^A TempSh_{it}^2 + \epsilon_{it}^A,$$
(17)

for firm *i* at time *t*, where $\mathbf{1}[\varphi_i = \varphi_2]$ is an indicator variable for permanent firm type. We run this regression by OLS both in the model and the data, and we target the regression coefficients β_1^A (which helps identify ν), and β_2^A and β_3^A (which help to jointly identify $\omega(\varphi_1)$, $\omega(\varphi_2)$ and α). Note that we pool firms of different types together but control for firm type φ by including the fixed effect $\mathbf{1}[\varphi_i = \varphi_2]$, which eliminates endogeneity concerns due to $\zeta(\varphi)$.

Next, we aim to match the effect of firm size and firm type on the temporary share, which should help identify $\omega(\varphi_2)/\omega(\varphi_1)$ and χ . To do so, we follow again an indirect inference approach and run the following OLS regression both in the model and the data:

$$TempSh_{it} = \beta_0^B \mathbf{1} [\varphi_i = \varphi_2] + \sum_{\ell=1}^{N_{\text{bins}}} \beta_\ell^B \mathbf{1} [Emp_{it} \in \text{SizeBin}_\ell] + \epsilon_{it}^B.$$
(18)

The β_0^B coefficient indicates the differential choice of temporary share by firms of different type (and same size), and should roughly pin down the ratio $\omega(\varphi_2)/\omega(\varphi_1)$, as $\omega(\varphi)$ determines the relative demand of skill and hence of temporary workers in firms of type φ . The $\{\beta_\ell^B\}$ coefficients, in turn, indicate the partial effect of firm size on the temporary share within a firm type, capturing the within-firm variation in Figure E.2.³³ This moment should inform about the firing cost shifter, χ , as this parameter drives the share of each worker type that a firm intends to keep in equilibrium.

Panel A of Table E.1 in Online Appendix E shows the empirical coefficients for regression (17). We find that conditional on type, larger firms are associated with slightly higher productivity ($\beta_1^A = 0.081$). This is the result of the positive correlation between firms size (*n*) and transitory firm productivity (*z*) in the cross-section, which dominates the direct negative effect of decreasing returns to scale that we obtain in the calibration. However, as it can be seen in Figure E.3 of Online Appendix E, lower values of ν lead to lower (even negative) values of β_1^A , as implied by the more severe decreasing returns to scale. We also find that, conditional on type, a larger temporary share is associated with a lower firm productivity ($\beta_2^A < 0$ and $\beta_3^A < 0$), consistent with the idea that low-skilled workers, which tend to have FTCs, are less productive. Finally, we also find that low types are, on average, less productive ($\beta_0^A < 0$), which is not surprising given the results of our classification approach described in Table 3.

Panel B of Table E.1, in turn, shows the empirical coefficients of regression (18). The β_0^B coefficient equals 0.201 in the data, which says that, conditional on firm size, φ_2 firms (the less productive ones) use a higher fraction of temporary contracts, 20.1 percentage points more. On the other hand, the $\{\beta_\ell^B\}$ coefficients are monotonically increasing in size bins ℓ , confirming that the within-firm positive relationship between firm size and temporary employment that we saw in Figures 1 and E.2 also holds when we condition on firm type instead of individual firm fixed effects.

In Panel B of Table 4, we report the calibrated model's fit on this regression evidence. In our indirect inference approach, we obtain $\nu = 0.881$, showing moderate decreasing returns to scale. Because temporary contracts are more prevalent among less skilled workers, this recovers $\omega(\varphi_1) = 0.645 > 0.316 = \omega(\varphi_2)$. Moreover, we find $\alpha = 0.542$, which delivers a considerably high elasticity of substitution between worker skill types, $\frac{1}{1-\alpha} = 2.183$. Then, we pin down a value for χ using $\beta_2^B - \beta_1^B$ (equal to 0.032 in the data), which captures the intensity by which the share of temporary workers changes within firm type, on average, when firms transition from the first to the second size bin. We choose to match the gap between these coefficients because the first two size

 $[\]overline{}^{33}$ In practice, we build size bins using 5-worker increments. As our sample contains firms with up to 60 employees, this implies $N_{\text{bins}} = 12$ size bins: 1-5, 6-10, 11-15, and so on up to 56-60.

bins comprise the vast majority of firms in our data. Given these parameter values, the calibrated model matches all of these coefficients well.

Worker stocks and flows Next, with the idea to identify $(A_{FT}, A_{OE}, s_{FT}^W, s_{OE}^W, \xi)$, we target the employment-to-unemployment (EU) and unemployment-to-employment (UE) quarterly flow rates by contract type from the Spanish labor force survey (EPA), as well as the average rate of temporary employment. In the data, we have a 3-state Markov chain (unemployed, employed with FTC, and employed with OEC), with 6 independent transitions. This system pins down two steady-state ratios: the temporary share and the unemployment rate. Consistent with the model, we set the OE-to-FT rate to zero, and target four of the remaining five transitions (leaving the FT-to-OE rate free) plus the average temporary share (leaving the unemployment rate free). In the model we compute the same workers flows, see Online Appendix A.4 for details.

Which parameters are informed by these five targets? First, the two UE rates are primarily affected by the recruiting cost parameters (A_{OE} , A_{FT}), as these act as shifters for the number of matches that take place in each market per unit time, given market tightness. In the data, UE transitions are far more frequent among FT workers (18.77% quarterly) than among OE workers (2.79%). The calibrated model matches these rates closely (18.41% and 2.52%, respectively) with $A_{FT} = 0.0099$ and $A_{OE} = 1.8510$. This means that the model rationalizes the difference in UE rates by making OE hires more costly: for given market tightness, there are more matches in the FT market per unit of time for every match that takes place in the OE market. These differences in recruiting costs endogenously imply that matching efficiency is much higher in the FT labor market.³⁴

Secondly, the EU rates are most directly affected by the worker exogenous separation rates, (s_{FT}^W, s_{OE}^W) . In the data, transitions into unemployment are far more common in the FT market (12.85% quarterly) than in the OE market (1.38%). These EU rates are closely matched by the calibrated model (11.92% and 1.46%, respectively). The calibration delivers that the separation rate of FT workers is high (equal to 0.46, i.e. an average duration of about 6 and a half months on the job, conditional on no endogenous separations) and that OE jobs are virtually never destroyed exogenously (s_{OE}^W is estimated to be nearly equal to zero). The exogenous separations in FTCs may be driven by voluntary quits or by regulations on the maximum duration of these contracts. The latter will be our preferred interpretation in our policy analysis of Section 5.

Finally, the promotion cost parameter ξ determines the flow of workers being promoted from FTCs to OECs. If we were to target this flow, then together with the other four flows described above we would uniquely pin down the unemployment rate and the temporary share. To identify ξ , we therefore target the average temporary share in the EPA data (21.9%).³⁵ Our calibrated model matches the average temporary share (21.8%), and implies an unemployment rate of 14.9%.³⁶

Other moments First, the probability of entering as φ_1 -type, $\pi_{\varphi}(\varphi_1)$, is pinned down by the share of firms of this type among all operating firms in the stationary solution of the model.³⁷ In the calibration, we obtain $\pi_{\varphi}(\varphi_1) = 26\%$, which is substantially lower than the 40.2% of φ_1 -type firms

³⁴Online Appendix A.5 describes how to compute aggregate matching efficiency in market i = OE, *FT*, which we denote Γ_i . In the calibrated economy, we find a large gap: $\Gamma_{FT}/\Gamma_{OE} \approx 46$.

³⁵Recall that the EPA numbers are always slightly higher than the firm-level ones (see Figure E.1). We choose the EPA number to be consistent with the fact that our targets for the worker flow rates have been computed using these data.

³⁶This is in the middle of the unemployment figures observed in Spain during our period of analysis (2004-2019). In the data, the unemployment rate obtains its lowest value in 2007 (at 8.2%) and its highest value in 2013 (at 26.1%).

³⁷In the model, we compute F_{φ_1}/F , where $F_{\varphi_1} \equiv \sum_i \sum_j \sum_z f(\{n_{ij}\}, z, \varphi_1)$, and $f(\{n_{ij}\}, z, \varphi_1)$ is the stationary measure of type- φ_1 firms of productivity z with n_{ij} workers of skill j = L, H in contracts of type i = FT, OE.

Parame	ter	Value	Moment	Model	Data	Source
Panel A	. Absolute and relative firm size					
κ	Fixed firm entry cost	823.38	Average firm size	7.01	7.35	CBI
$\zeta(\varphi_2)$	Permanent productivity of φ_2 firms	0.876	Relative size of φ_2 firms	0.650	0.693	CBI
Panel B	. Productivity and temporary share by	[,] firm char	acteristics			
ν	Returns-to-scale parameter	0.881	β_1^A coefficient, eq. (17)	0.078	0.081	CBI
$\omega(arphi_1)$	Productivity H workers in φ_1 firms	0.645	β_2^A coefficient, eq. (17)	-0.047	-0.104	CBI
α	Substitutability between worker types	0.542	β_3^A coefficient, eq. (17)	-0.086	-0.092	CBI
$\omega(\varphi_2)$	Productivity H workers in φ_2 firms	0.316	β_0^B coefficient, eq. (18)	0.260	0.201	CBI
χ	Firing cost shifter	0.110	$(\beta_2^B - \beta_1^B)$ gap, eq. (18)	0.041	0.032	CBI
Panel C. Worker stocks and flows						
A_{OE}	Recruiting cost shifter (OECs)	1.851	UE rate, OE (quarterly)	0.025	0.028	EPA
A_{FT}	Recruiting cost shifter (FTCs)	0.0099	UE rate, FT (quarterly)	0.184	0.188	EPA
s_{OE}^W	OEC destruction rate	2×10^{-5}	EU rate, OE (quarterly)	0.015	0.014	EPA
s_{FT}^W	FTC destruction rate	0.460	EU rate, FT (quarterly)	0.119	0.129	EPA
ξ	Promotion cost shifter	0.735	Temporary share	0.218	0.219	EPA
Panel D. Other moments						
$\pi_{\varphi}(\varphi_1)$	Probability of entering as a φ_1 type	0.260	Share of φ_1 -type firms	0.402	0.197	CBI
s^F	Firm destruction rate	0.0075	Firm entry rate (annual)	0.075	0.075	INE
b	Employment opportunity cost	1.480	Leisure to output p.w.	0.819	0.800	•

Note: The model period is one quarter. The table reports the values of the parameters estimated internally by minimizing the criterion function in equation (15). UE and EU rates in the data are averages over HP-filtered quarterly series from the EPA over the period 2005Q1-2018Q4 (data before 2005 is unavailable), see Appendix A.4 for details. *Data sources:* "CBI" means our sub-sample from the Central de Balances Integrada data; "INE" means data from the Instituto Nacional de Estadística; "EPA" means data from the Encuesta de Población Activa.

in the stationary distribution, indicating higher survival probabilities for the high-type firms.

Second, to pin down s^F we target the entry rate of firms. In the data, firm entry rates fluctuated around 7.5% annually after the Great Recession (García-Perea, Lacuesta and Roldan-Blanco (2021)). In the model, the inflow of firms equals its outflow, so we compute the firm entry rate as the ratio of entrants to the total measure of operating firms.³⁸ This gives $s^F = 0.0075$ (see Table 4, Panel D).

Finally, we calibrate the opportunity cost of employment, b, to match that the income flow from unemployment represents 80% of the average worker productivity (following Costain *et al.* (2010)). The calibrated model delivers b = 1.48 and predicts a value of 81.9% for our target.

4.2.3 Global Identification Results

To validate the identification of the 15 internally estimated parameters, we run the following exercise. For each parameter-moment pair established in the text (see Table 4 for the summary), we allow for quasi-random variation in all remaining parameters and solve the model for each such parameter configuration.³⁹ As a result, for each level of the identified parameter we obtain a

³⁸Formally, $\frac{\mathbf{F}^e}{\mathbf{F}} \sum_{\varphi} \pi_{\varphi}(\varphi) \left\{ \sum_{z^e} \pi_z(z^e) \sum_i \eta_i \left(W'_{iL}(\vec{n}^e_{iL}, z^e, \varphi) \right) \right\}$, where \mathbf{F}^e is the measure of potential entrants.

³⁹More precisely, we use the following procedure, inspired by Daruich (2023). First, we set wide enough bounds for each parameter from the \mathbf{p}_{int} parameter vector. Then, we pick quasi-random realizations from the resulting hypercube using a Sobol sequence, which successively forms finer uniform partitions of the parameter space. Finally, for each parameter combination, we solve the model and store the relevant moments. To implement this last step, we use a high-performance computer, which allows us to parallelize the numerical solution and saves us a large amount of computational time. In total, we solve the model 1.3 million parameter combinations.

whole distribution for the targeted moment. In Figure E.3 of Online Appendix E we plot, for each parameter-moment pair, the median of this distribution (black dots) and the inter-quartile range (shaded area), together with the empirical target (dashed line).

We consider that a parameter is well-identified by the moment when (i) the distribution changes across different values of the parameter; (ii) the rate of this change is high; (iii) the inter-quartile range of the moment's distribution is narrow throughout the support for the parameter; (iv) the empirical target falls within the inter-quartile range.⁴⁰ Because all the remaining parameters are not fixed but instead are varying in a quasi-random fashion within a wide support, this method gives us a global view of identification. Figure E.3 shows that the parameters are well-identified by their corresponding moments along most of the aforementioned criteria.

4.3 Non-Targeted Moments

Next, we validate the calibration results by confronting the model against non-targeted moments. We focus on moments related to features of the cross-sectional distribution of firms against firm size, age, temporary employment and employment growth.⁴¹ Figure E.4 in Online Appendix E shows the distribution of firms and employment by firm size and firm age, in the data and in the calibrated model. Along the size dimension, the model aligns well with the data, especially in terms of firm shares, even though nothing from these distributions was used as calibration targets. On the age dimension, there is a slightly larger share of young firms in the model compared to the data, and these firms hold too much employment, but the model correctly predicts the remaining parts of the age distribution, both in terms of firm shares as well as employment shares.

Second, we validate the calibrated model's predictions on firm employment growth.⁴² Figure E.5 compares the distributions of yearly employment growth for total employment, FT employment and OE employment, in the data versus the calibrated model. The calibrated model correctly predicts the dispersion in all three distributions, an indication that the parameters of our idiosyncratic productivity process are well identified. The model is remarkably able to replicate the symmetry in the distribution of growth rates for all types of employment, and the share of firms that obtain their first, or lose their last, FT worker (i.e. the spikes at 2 and –2 in the middle plot of Figure E.5).

4.4 Misallocation

Before proceeding to our policy exercises, we study the extent of worker misallocation generated by the search and matching frictions and the dual labor market structure. Online Appendix B.1 derives the allocation of workers that would result from (i) maximizing output without being subject to the constraints imposed by the search frictions of the market economy, but (ii) taking the allocation of firms across productivity bins (φ , z) as given. In this Benchmark economy the marginal

⁴⁰Criterion (i) implies that the moment is globally sensitive to variation in the parameter; (ii) gives an idea of how strong this sensitivity is; (iii) measures how much other parameters matter to explain variation in the moment; and (iv) implies that the empirical target is not an outlier occurrence.

⁴¹To compute these moments in the model, we must rely on simulated data. We simulate 10,000 firms over 400 quarters (i.e. 100 years). We partition each quarter into 2,000 sub-periods to ensure that Poisson rates approximate probabilities well. To allow for convergence, we report results computed using only the last year of our simulated data.

⁴²For any firm *i*, year-on-year employment growth is always computed as $g_{i,t} = \frac{Emp_{i,t+1} - Emp_{i,t}}{\frac{1}{2}(Emp_{i,t+1} + Emp_{i,t})}$, where $Emp_{i,t}$ denotes total, FT, or OE employment. This measure, which we borrow from Davis, Haltiwanger and Schuh (1998), is appealing because it accounts for entry ($g_{i,t} = 2$) and exit ($g_{i,t} = -2$) of firms, treating them symmetrically.

product of either type of labor is equalized across all firms. This means that (i) the allocation of workers and skills is identical between firms of the same productivity class (φ, z) ; (ii) more productive firms (higher φ or higher z) employ more workers (due to decreasing returns to scale); and (iii) the allocation of worker skills to firms is increasing in φ and independent from z because the skill share parameter $\omega(\varphi)$ is increasing in φ and independent from z.

The competitive equilibrium allocation differs from this benchmark in two ways. First, within firm productivity class (z, φ) , firms in the competitive equilibrium display different amounts of workers and skills. Second, the average amount of workers and skills allocated across firms of the same productivity class (z, φ) in the competitive equilibrium differs from the Benchmark allocation. Both of these translate into output losses. Quantitatively, we find that our calibrated economy produces 3.9% less output per worker than the Benchmark economy with the same level of employment and the same distribution of firms, showing that search and matching frictions in Spain are quite damaging for welfare. Of this loss, 40.3% is due to the between productivity class (z, φ) misallocation, and the remainder is due to the within productivity class (z, φ) component. Figure E.6 in Online Appendix B.2 shows the underlying heterogeneity in misallocation across productivity types (z, φ) .

5 The Macroeconomic Implications of Dual Labor Markets

We are ready to quantify the macroeconomic effects of dual labor markets and the impact of policies that regulate FTCs. We study three distinct policies: limiting the maximum duration of FTCs (Section 5.1), taxing the use of FTCs (Section 5.2), and banning the use of FTCs (Section 5.3).

5.1 Reducing the Maximum Duration of FTCs

A key characteristic of dual labor markets is the (relatively short) time limit that workers can spend in a firm under an FTC. When the limit is reached, firms are forced to convert the contract into an OEC or let the worker go. Reducing the maximum duration of FTCs is also a common policy tool for countries that want to limit the use of FTCs, as in the 2022 labor market reforms implemented in Spain. In this Section, we explore the consequences of changes in the legal maximum duration of FTCs. To do so, we solve for a series of economies in which we vary the exogenous separation rate for FT workers, s_{FT}^W , such that average duration for FTCs moves between one month and 4 years (in the calibrated economy, the average duration is 6.5 months). We leave all other parameters unchanged at their calibrated values, and compare across steady-state solutions. The results for this series of exercises are reported in Figure 2. Table 5 provides exact numbers for the two policies at the extreme, one with one month FTC duration and another one with 4 years duration.

A policy that reduces the legal duration of FTCs from the baseline duration down to one month (Column (A) in Table 5 and the left-most point in each panel of Figure 2) achieves the intended effect of reducing the overall temporary share of the economy (see Panel (f)), from 21.8% to 0.9%. The policy also increases aggregate productivity, although only modestly by 0.26%. However, this policy would come at the expense of a large reduction in output (Panel (r)) via an increase in the unemployment rate (Panel (q)) that is larger than the increase in aggregate productivity. All in all,

limiting FTC duration to one month would reduce welfare by 0.87% (Panel (t)).⁴³ In what follows, we discuss these results in more detail.

5.1.1 Aggregate Employment and its Composition

In the model, firms face a trade-off between the opportunity costs (in lost output) from the high worker turnover of workers under FTCs, and the higher recruiting costs of workers under OECs. For given firm productivity (φ , z), firms prefer a higher share of FTCs as they become larger because their marginal product of labor declines and so does the opportunity cost of unfilled vacancies. When FTCs expire more quickly, the turnover of workers with FTCs increases and this trade-off is tilted in favor of OE hiring. As firms' recruiting efforts for OE workers increase, the matching efficiency in this market raises (Panel (i)), while it decreases in the FT market (Panel (h)).

This intuition helps explain the policy effects on aggregate worker flows. On the one hand, the aggregate EU flow declines (Panel (m)) as FTC duration declines from 6.5 months. This is the result of the strong decline in the share of temporary workers, which more than offsets the mechanical increase in the EU flow of FTCs and the very small increase in the EU flow of OECs (Panels (n) and (o)). On the other hand, the aggregate UE flow also declines (Panel (j)). As now firms increasingly prefer to hire through the OE margin, job-finding rates decline in the FT market (Panel (k)), increase slightly in the OE market (Panel (l)), and promotion rates increase sharply (Panel (p)), with an overall decline in the UE flow due to the lower hiring needs of firms as the job destruction rate declines. The combination of declining job-filling rates and declining job-destruction rates lead to an increase in unemployment (Panel (q)), from 14.9% in the baseline to 21.0% in the one-month FTC duration policy.⁴⁴ Finally, as firms now prefer to employ more workers under an OEC, and because only OE workers can accumulate human capital on the job, the share of high-skill workers in the economy increases substantially, from 46.0% in the baseline to 52.5% under the policy (Panel (g)).

⁴³To compute welfare, we use equation (A.37) in Online Appendix A.6. In words, welfare equals the present discounted value of production net of firing, recruiting, and promotion costs over all operating firms, plus the value of home production across all unemployed workers, net of the total entry costs paid by all potential entrant firms.

⁴⁴The dynamics of unemployment are written in the system of equations (A.29a)-(A.29d) of the Online Appendix.



Notes: This figure shows the effects of reducing the maximum duration of FTCs on a number of macroeconomic aggregates of interest. For all panels, the horizontal axis represents $1/s_{FT}^W$, and is measured in quarters. The plots shows different stationary solutions of the model, keeping all parameters fixed at their baseline calibration values except for s_{FT}^W . The red dashed vertical line shows the expected duration in the baseline calibration. Aggregate output, matching efficiency, and welfare are all normalized to one at the baseline calibration. For the computation of EU and UE rates, see Online Appendix A.4.

	(A)	(B)	(C)
	Short duration (1 month)	Baseline (6.5 months)	Long duration (4 years)
Measure of operating firms	0.112	0.121	0.134
Share of type- φ_1 firms	47.5 %	40.2 %	34.3 %
Average firm size	7.03	7.01	6.95
Relative size φ_2 firms	0.54	0.65	0.80
Firm entry rate (annualized)	7.4 %	7.5 %	7.6 %
Average temporary share	0.9 %	21.8 %	50.4 %
within φ_1 firms	0.3 %	5.2 %	13.2 %
within φ_2 firms	1.8 %	32.9 %	74.8 %
Share of <i>H</i> workers	52.5 %	46.0 %	39.3 %
within φ_1 firms	71.7 %	71.3 %	70.2 %
within φ_2 firms	20.2 %	19.7 %	19.0 %
Matching efficiency (FT)	0.28	1.40	0.48
Matching efficiency (OE)	0.11	0.03	0.01
UE rate (total)	8.3 %	20.9 %	20.8 %
UE rate (FT)	2.7 %	18.4 %	17.7 %
UE rate (OE)	5.6 %	2.5 %	3.2 %
EU rate (total)	2.3 %	3.7 %	1.6 %
EU rate (FT)	75.9 %	11.9 %	2.4 %
EU rate (OE)	1.6 %	1.5 %	0.8 %
Promotion rate	10.9 %	3.0 %	0.3 %
Unemployment rate	21.0 %	14.9 %	6.8 %

Table 5: Effects of changes in average FTC duration on macro aggregates

Notes: This table shows the effects of reducing the maximum duration of FTCs on a number of macroeconomic aggregates of interest. Column (B) corresponds to the baseline calibration; in column (C), we set $s_{FT}^W = 1/16$ so that FTCs expire on average after 16 quarters, or 4 years; in column (A) we set $s_{FT}^W = 3$, so that FTCs expire on average after 1 month. UE, EU and promotion rates are quarterly figures, whereas the firm entry rate is an annual figure. For the computation of EU and UE rates, see Online Appendix A.4.

5.1.2 The Measure of Firms and its Composition

As a byproduct of the changing firm incentives described above, there are also important consequences for the total measure of operating firms and the distribution of firms across productivity levels. First, with FTCs being less useful for firms there is a reduction in the value of firm incumbency, which reduces firm entry and the total measure of operating firms F (Panel (a)). As there is a decrease in F but the employment rate E = 1 - U also declines, the net effect on firm size is ambiguous. Quantitatively, Panel (c) shows that firms barely change size across different policies. Second, under the policy reform, there are strong selection effects, with a decrease in the turnover of firms (Panel (e)), a higher share in equilibrium of firms of the more productive permanent type (Panel (b)), and an increase in the relative size of those firms (Panel (d)). Because firms of the less productive type φ_2 rely more on FT workers, the worsening of these contracts damages these firms relatively more, inducing a higher exit risk among them.

5.1.3 Aggregate Productivity

Next, we turn to aggregate productivity. The understand the policy effects on this margin, we provide a novel decomposition. In fact, our decomposition formula should be general to models of firm dynamics featuring decreasing returns to scale in production, firm selection, and worker misallocation. Let us start by defining aggregate output as

$$Y \equiv \sum_{n_H=0}^{+\infty} \sum_{n_L=0}^{+\infty} \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} \left[y(n_H, n_L, z, \varphi) \, \mathbf{f}(n_H, n_L, z, \varphi) \right],\tag{19}$$

where recall that $f(n_H, n_L, z, \varphi) \ge 0$ is the measure of operating firms that are of type φ , have productivity z, and employ $n_H \equiv n_{OEH}$ high-skill workers and $n_L \equiv n_{FT} + n_{OEL}$ low-skill workers. The production function $y(\cdot)$ is given in equation (2). Our object of interest is aggregate productivity, defined as output per worker, Y/E.

Now, to decompose aggregate productivity, we need to introduce some new notation. First, let $n \equiv n_H + n_L$ be total firm employment. Second, let $\hat{n} \equiv n/(E/F)$ be the total employment of a firm relative to the average firm size, so that if $\hat{n} > 1$ (respectively, $\hat{n} < 1$) the firm is larger (respectively, smaller) than the average firm in the economy. And third, let $h \equiv n_H/n$ be the skill share of the firm. Let $\hat{\mathcal{N}}$ and \mathcal{H} denote the supports of \hat{n} and h which, because of our assumptions, are countable sets.⁴⁵ Then, in Online Appendix B.3 we show that aggregate output per worker can be written as

$$\frac{Y}{E} = \underbrace{\left(\frac{F}{E}\right)^{1-\nu}}_{\substack{\text{Firm size}\\\text{component}}} \left[\sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} \underbrace{\frac{F_{z,\varphi}}{F}}_{\substack{\text{Firm}\\\text{selection}\\\text{component}}} \underbrace{\left(\sum_{\widehat{n} \in \widehat{\mathcal{N}}} \sum_{h \in \mathcal{H}} y_{z,\varphi}^{h}(\widehat{n},h) g_{z,\varphi}(\widehat{n},h)\right)}_{\text{Worker reallocation component}} \right]$$
(20)

In these equations, $F_{z,\varphi} \ge 0$ is the measure of operating firms of productivity type (z, φ) , so that $\sum_{z} \sum_{\varphi} F_{z,\varphi} = F$; we define $y_{z,\varphi}^{h}(\hat{n},h) \equiv \hat{n}^{\nu} y(h, 1-h, z, \varphi)$, with $y(\cdot)$ given by equation (2); and $g_{z,\varphi}(\hat{n},h) \in (0,1)$, which is defined precisely in equation (B.6) of the Online Appendix, is the fraction of firms of productivity type (z, φ) that have relative employment \hat{n} and skill share h.

Equation (20) shows that we can decompose aggregate productivity changes into changes in three distinct components. The first term, the *firm size component*, reflects the effect of firm size on aggregate productivity. This term captures productivity gains due to increasing the number of firms per worker, which increases aggregate productivity due to the existence of decreasing returns to

	(A)	(B)
	Short duration (1 month)	Long duration (4 years)
Change in output per worker, of which:	0.26 %	-0.23 %
(a) Firm size channel	-0.05 %	0.09 %
(b) Firm selection channel	4.91 %	-4.37 %
(c) Reallocation channel, of which:	-5.77 %	2.90 %
between-firm component	-5.62 %	2.59 %
within-firm component	-0.70 %	0.75 %
Change in output	-6.86 %	9.29 %
Output loss from misallocation (in levels), of which:	4.29 %	3.57 %
share due to between-firm misallocation	39.52 %	60.15 %
Change in welfare	-0.87 %	1.44 %

Table 6: Effects of changes in average FTC duration on productivity, misallocation and welfare

Notes: This table shows the effect of the policy, expressed in percentage changes with respect to the baseline calibration of 6.5 months FTC duration (with the exception of the output loss from misallocation and the share of it that is due to between-firm misallocation, which are expressed in levels). Welfare is computed as in equation (A.37), see Online Appendix A.6.

⁴⁵For a given E/F, \hat{n} takes values in a countable set because n_H and n_L are both integers. Moreover, h takes values in the subset of rational numbers contained in the unit interval, i.e. $\mathcal{H} = \mathbb{Q} \cap [0, 1]$ (for instance, if n = 1, then $h \in \{0, 1\}$; if n = 2, then $h \in \{0, 1/2, 1\}$; if n = 3, then $h \in \{0, 1/3, 2/3, 1\}$; and so on).

scale ($\nu < 1$). Our results (second row in Table 6) show that the *firm size* channel has a negligible impact on productivity when the duration of FTCs decreases. Indeed, panel (c) of Figure 2 shows that the measure of operating firms and the employment rate move almost one for one and yield negligible changes on firm size.

The second term, the *firm selection component*, reflects the effects of firm composition on aggregate productivity. Changes in the shares of firms in each productivity bin lead to productivity gains if those firms that are more productive become more abundant in the economy. This channel works through firm exit: to the extent that the policy change lowers the survival probability of the less productive firms, aggregate productivity will rise.⁴⁶ We find that the firm selection effect contributes a 4.91% increase in aggregate productivity in response to the policy change (third row in Table 6). Indeed, as FTC duration decreases, the new distribution of firms features a higher share of φ_1 firms, as seen in Panel (b) of Figure 2.

The third term, the *worker reallocation component*, captures the effects of employment reallocation *within* and *across* firms of a given productivity class (φ , z). Through this channel, aggregate productivity improves in response to a policy change when the relative allocation of workers across firms improves, either through the reallocation of total employment (relative to average firm size) or through a reallocation of human capital within the firm. The probability mass function $g_{z,\varphi}(\hat{n}, h)$ measures how firms are distributed in the space of relative sizes and human capital. Changes in this distribution command aggregate productivity gains or losses through employment reallocation dynamics. With the reduction in the duration of FTCs, the allocation of employment across firms worsens: the reallocation channel, in isolation, would lead to a 5.77% decrease if FTCs expired after one month (fourth row in Table 6).

To understand what drives the worker reallocation channel margin, we split the joint probability $g_{z,\varphi}(\hat{n},h)$ into the product of conditional and marginal probabilities:

$$\mathbf{g}_{z,\varphi}(\widehat{n},h) = \mathbf{g}_{z,\varphi}^{A}(h|\widehat{n})\mathbf{g}_{z,\varphi}^{B}(\widehat{n}).$$
(21)

The first term, $g_{z,\varphi}^A(h|\hat{n})$, reflects a *within-firm* reallocation component, as it captures how the skill composition of workers changes within firms of the same productivity (z, φ) and same relative size \hat{n} . The second term, $g_{z,\varphi}^B(\hat{n})$, reflects a *between-firm* reallocation component, as it captures how the relative number of workers \hat{n} changes across firms of different productivities (z, φ) . In our exercise, we find that the worsening in productivity due to the worker reallocation channel is predominantly due to the between-firm component (fifth and sixth rows in Table 6). Intuitively, when FTCs become of shorter duration, firms lose flexibility to adjust their total number of employees in response to productivity shocks. This is the classic misallocation effect due to an increase in firing costs as in Hopenhayn and Rogerson (1993). Instead, the within-firm component brings productivity gains due to the increase in human capital for given firm size due to the decline in worker turnover.

Finally, we can explore how the effects of the policy are heterogeneous across firms of different productivity classes, (z, φ) . We relegate the results and the discussion to Online Appendix B.2.2, where we emphasize heterogeneous effects on employment misallocation.

5.1.4 Extension: Selection Upon Entry

As we have seen, a quantitatively important margin to evaluate the effects of policy on aggregate ⁴⁶See Section 5.1.4 for a model extension where there is also selection upon entry. productivity is the firm selection channel. In the baseline model, firm entry is fully undirected, in the sense that potential entrants know neither their type φ nor their initial productivity shock *z* when they pay the entry cost κ . This means that changes in the economic environment leading to relative changes in the value of entry for different types of firms do not change the composition of entrants. Thus, all selection occurs on the exit margin.

In this Section we explore how the effects of shortening the duration of FTCs change when we allow for selection of entrants. To do so, we assume that potential entrants can choose their type φ . We also assume that the technology choice φ entails a flow entry cost $\kappa(\varphi) - \varepsilon_{\varphi}$, where $\kappa(\varphi)$ is common to all entrants and ε_{φ} is idiosyncratic. Finally, we assume that the common component $\kappa(\varphi)$ is known before entry but the actual realization of ε_{φ} is not. This captures uncertainty about the actual costs of entry that leads to smooth changes in the share of entrants of each type. Under these assumptions, the ex-ante value of entry J^e, common to all entrants, is

$$\mathbf{J}^{e} = \mathbb{E}\left[\max_{i \in \{FT, OE\}} \left\{ \mathbf{J}^{e}(\varphi_{i}) - \kappa(\varphi_{i}) + \varepsilon_{\varphi_{i}} \right\} \right]$$
(22)

where the expectation is taken over all possible realizations of ε_{φ_1} and ε_{φ_2} . Free entry requires that $\mathbf{J}^e = 0$. The idea is that a more productive technology yields higher value of entry $\mathbf{J}^e(\varphi)$ but it also entails a higher entry cost $\kappa(\varphi)$, perhaps due to the investments in more productive technologies. We make this problem tractable with the assumption of Extreme Value distributions for ε_{φ_1} and ε_{φ_2} . Full details can be found in Online Appendix C.

In this Appendix, we show that for any chosen elasticity of entry composition $\pi_{\varphi}(\varphi_1)/\pi_{\varphi}(\varphi_2)$ to the differential value of entry $\mathbf{J}^e(\varphi_1) - \mathbf{J}^e(\varphi_2)$, there is a pair of entry costs $(\kappa(\varphi_1), \kappa(\varphi_2))$ that delivers the exact same model outcomes as the calibrated economy with undirected entry. This is useful because it implies that we do not need to recalibrate the economy with the directed entry. Instead, we chose arbitrarily an elasticity of one, obtain the $\kappa(\varphi_1)$ and $\kappa(\varphi_2)$ that deliver the same model outcomes as the baseline, and perform the policy reform of varying the duration of FTCs.

We find that when we shorten the contract duration of FTCs to 1 month, the share of φ_1 firms increases more than in the case of undirected entry: up to 65.5% instead of 47.5% (see Table E.2 in the Online Appendix). This is by construction, and the exact amount depends on the chosen elasticity of entry composition to the differential value of entry.

The consequences of a larger share of φ_1 firms in the economy are important. First, there is an overall smaller mass of operating firms, as φ_1 firms have a higher demand of workers and crowd out other firms by increasing average labor market tightness and lowering the value of entry for all firms. This results in average firm size increasing from 7.01 to 7.78 workers. Second, the share of workers with an FTC declines more (to 0.6% instead of 0.9%) as φ_1 firms demand fewer FT workers. This implies that the EU flow also declines more (to 1.8% instead of 2.3%), leading to both higher human capital accumulation (60.4% of high-skilled workers instead of 52.5%) and lower unemployment rate (17.8% instead of 21.0%). Third, output per worker barely changes but there are substantial changes in its different components (see Table E.3 in the Online Appendix). The *firm size channel* leads to productivity losses of 1.24% due to the reduction in the number of operating firms. The *firm selection channel* leads to productivity gains as large as 12.38% (due to the larger increase in φ_1 firms), which compares to 4.91% productivity gains in the economy without directed entry. However, the *worker reallocation channel* leads to productivity losses of 14.42%, much larger than the 5.77% losses in the economy without directed entry. This worsening of the worker reallocation

is partly due to (i) the fact that φ_1 firms, by using a lower share of FT workers, are less flexible in growing and shrinking with productivity shocks, and in part also due to (ii) the fact that φ_2 firms experience productivity losses as their share of skilled workers grows above their optimal size.

Finally, due to the lower increase in unemployment, the output loss is halved (3.48% instead of 6.86%) and welfare increases slightly (instead of falling) due to the combination of lower output loss and the saving on the recurrent recruiting costs associated to the FTCs.

5.2 Taxing the Use of FTCs

An alternative policy used to limit the temporary share is the direct taxation of the stock of workers employed under FTCs. This type of policy is currently in place in countries such as France, Portugal, and Spain (see Cahuc, Benghalem, Charlot, Limon and Malherbet (2016)). We study the effect of such a policy by introducing a linear tax on the number of FT workers, so that the flow of firm net revenues becomes $y(\vec{n}, z, \varphi) - \tau_{FT} n_{FT}$. Proceeds are rebated lump sum to ensure that the tax is resource-neutral (i.e. change in welfare only comes from changes in firm policies and ensuing changes in the distribution, and not from the fact that the economy has more or fewer resources).

We pick the tax rate τ_{FT} to produce the same reduction in the aggregate share of FTCs as in the economy where the maximum duration of FTCs is limited to 1 month, and we report the results of these two economies in Online Appendix Tables E.4 and E.5. The effect of this tax on aggregate productivity is very similar to the effect of lowering the maximum duration of FTCs. Firm selection is unchanged and so is the misallocation of workers. However, the effects on unemployment, output, and welfare differ. In particular, there is much less job destruction and worker turnover with a tax on the use of FTCs than with a reduction on the maximum duration of FTCs: the EU rate becomes 1.8% in the economy with a tax on FTCs and 2.3% in the economy that limits the maximum duration of FTCs. These changes mainly come from the EU rate of FT workers and are only partly offset by higher hiring in the economy with limits in the duration of FTCs: the UE rate becomes 7.0% in the tax economy and 8.3% in the economy that limits the maximum duration of FTCs. As a consequence, the unemployment rate grows to 19.6% in the tax economy, below the 21.6% rate in the economy that limits the duration of FTCs.

Finally, because aggregate productivity hardly varies across policies, the effects on aggregate output and welfare are dominated by the effects on total employment. Output falls by 5.2% in the tax economy, compared to 6.86% in the economy that limits the maximum duration of FTCs. For welfare, these figures are -0.65% and -0.85% respectively. We hence conclude this Section by noting that taxes on the stock of workers employed under FTCs are preferable to placing stricter limits to the maximum duration of these contracts, as this latter policy generates more job destruction.

5.3 Outright Ban on the Use of FTCs

To conclude, we explore the consequences of banning the use of FTCs altogether. To implement this counterfactual, we make FTCs identical to OECs by (i) setting $s_{FT}^W = s_{OE}^W$ and $A_{FT} = A_{OE}$, (ii) allowing FTCs to acquire skill j = H at the same rate τ as OECs, and (iii) set the promotion cost shifter ξ to infinity. This implies that, as in the benchmark economy, every instant of time every firm posts two vacancies; however, different from the benchmark economy, the two vacancies are identical to each other and equivalent to the OEC vacancy of the dual markets economy.

The aggregate consequences of banning FTCs are reported in Column (B) of Online Appendix Tables E.4 and E.5. We find that banning FTCs is bad for the economy. Output per worker increases slightly but unemployment grows more, leading to overall output and welfare losses equal to 3.18% and 0.31%. FTCs are useful to firms because they give them flexibility to adjust employment to changing productivity. This is because of both their lower cost of recruiting effort and their short fixed duration. At the same time, they are bad because they help less productive firm types to survive. Our results show that when banning FTCs, the productivity losses due to the worker reallocation channel are similar to the productivity gains of the firm selection channel. However, unemployment rises from 14.9% in the benchmark economy up to 17.8% under the ban. This is mainly due to the large drop in the UE rate (20.9% to 7.5%), which is the result of a fall in aggregate matching efficiency resulting from the disappearance of FTCs.

6 Conclusion

Many labor markets are characterized by a dual structure, whereby firms are allowed to offer both open-ended contracts (OECs) with large termination costs, and fixed-term contracts (FTCs) of short duration. Using rich administrative balance-sheet data for Spain (2004-2019), we document that there exists a high degree of heterogeneity in the use of FTCs between firms. An overwhelming majority of the cross-sectional variation in the temporary share is due to firm-level characteristics. In particular, though larger firms exhibit lower shares of temporary employment when comparing between firms within the same sector, region and time period, this relationship is reversed when exploiting within-firm variation.

We build and calibrate a model of firm dynamics with search and matching frictions to explore the macroeconomic implications of dual labor markets. To match the micro-level facts, our calibrated model grants the permanently larger and more productive firms with a stronger preference for high-skill workers. Workers accumulate firm-specific human capital on the job, and this rationalizes that these firms rely less on temporary employment in the cross-section of firms as worker turnover damages human capital accumulation. On the other hand, within-firm variation is driven by the trade-off between higher worker stability and higher hiring costs of workers under OECs. Firms that are far from their optimal size feature a large marginal product of labor and are especially damaged by worker turnover, so they are willing to pay the higher recruiting costs of OECs.

In our main quantitative exercise, we find that reducing the average duration of FTCs is effective in lowering the share of temporary employment in the economy and to increase aggregate productivity through strong firm selection effects. However, this policy is ill-advised as it increases unemployment and decreases overall welfare via the misallocation of employment both within and across firms. Taxing firms for their use of FTCs would lead similar results, but with lower unemployment increases and hence lower welfare losses. All in all, our paper emphasizes that the firm side is an important dimension to consider when quantifying the aggregate implications of labor market duality, a phenomenon which is pervasive in both emerging and developed economies.

Finally, we note that the core of our analysis features selection of firms due to differential exit after policy reforms. In a robustness exercise, we show that adding selection upon entry, whereby potential entrants choose their technology and this choice responds to the labor market regulations, may affect our results. In this case, the effects of a reduction in the duration of FTCs on aggregate
productivity would not change much: the better selection of firms is partly offset by the larger misallocation of workers due to the use of OECs by these firms. However, the losses due to increased unemployment would be much lower and so would the total output losses. This is again due to the lower use of FTCs by the more productive firms. Overall, there could be small welfare gains due to the saving in recruiting costs associated to FTCs. It is therefore a matter of interest to better understand the choice of technology by entrants, and to measure how this choice is affected by changes in the labor market conditions.

References

- Acemoglu, Daron, and William B. Hawkins. (2014). "Search with multi-worker firms". *Theoretical Economics*, 9(3), pp. 583-628. https://doi.org/10.3982/TE1061
- Ahn, Hie Joo, Bart Hobijn and Ayşegül Şahin. (2023). "The Dual U.S. Labor Market Uncovered". Finance and Economics Discussion Series, 2023-031, Board of Governors of the Federal Reserve System. https://doi.org/10.17016/FEDS.2023.031
- Alba-Ramirez, Alfonso. (1994). "Formal training, temporary contracts, productivity and wages in Spain". Oxford Bulletin of Economics and Statistics, 56(2), pp. 151-170. https://doi. org/10.1111/j.1468-0084.1994.mp56002003.x
- Almunia, Miguel, David López-Rodríguez and Enrique Moral-Benito. (2018). "Evaluating the macrorepresentativeness of a firm-level database: an application for the Spanish economy". Documentos Ocasionales, 1802, Banco de España. https://repositorio.bde.es/handle/123456789/6402
- Auciello-Estévez, Iván, Josep Pijoan-Mas, Pau Roldan-Blanco and Federico Tagliati. (2023). "Dual labor markets in Spain: a firm-side perspective". Documentos Ocasionales, 2310, Banco de España. https:// doi.org/10.53479/29869
- Audoly, Richard. (2023). "Firm Dynamics and Random Search over the Business Cycle". Staff Reports, 1069, Federal Reserve Bank of New York. https://doi.org/10.59576/sr.1069
- Baley, Isaac, Ana Figueiredo, Cristiano Mantovani and Alireza Sepahsalari. (2023). Self-Insurance in Turbulent Labor Markets. https://www.isaacbaley.com/uploads/6/7/3/5/6735245/bfms_turbulencewealth_ may2023.pdf
- Bentolila, Samuel, Pierre Cahuc, Juan J. Dolado and Thomas Le Barbanchon. (2012). "Two-Tier Labour Markets in the Great Recession: France Versus Spain". *The Economic Journal*, 122(562), pp. 155-187. https://doi.org/10.1111/j.1468-0297.2012.02534.x
- Berton, Fabio, and Pietro Garibaldi. (2012). "Workers and Firms Sorting into Temporary Jobs". *The Economic Journal*, 122(562), pp. 125-154. https://doi.org/10.1111/j.1468-0297.2012.2533.x
- Bilal, Adrien, Niklas Engbom, Simon Mongey and Giovanni L. Violante. (2022). "Firm and Worker Dynamics in a Frictional Labor Market". *Econometrica*, 90(4), pp. 1425-1462. https://doi.org/10.3982/ ECTA17955
- Blanchard, Olivier, and Augustin Landier. (2002). "The Perverse Effects of Partial Labour Market Reform: Fixed-Term Contracts in France". *The Economic Journal*, 112(480), pp. 214-244. https://doi. org/10.1111/1468-0297.00047
- Bonhomme, Stephane, Thibaut Lamadon and Elena Manresa. (2022). "Discretizing unobserved heterogeneity". *Econometrica*, 90(2), pp. 625-643. https://doi.org/10.3982/ECTA15238
- Bratti, Massimiliano, Maurizio Conti and Giovanni Sulis. (2021). "Employment protection and firmprovided training in dual labour markets". *Labour Economics*, 69, 101972. https://doi.org/10.1016/j. labeco.2021.101972
- Buchinsky, Moshe, Denis Fougère, Francis Kramarz and Rusty Tchernis. (2010). "Interfirm mobility, wages and the returns to seniority and experience in the United States". *Review of Economic Studies*, 77(3), pp. 972-1001. https://doi.org/10.1111/j.1467-937X.2010.598.x
- Cabrales, A., Juan J. Dolado and Ricardo Mora. (2017). "Dual employment protection and (lack of) on-thejob training: PIAAC evidence for Spain and other European countries". SERIEs: Journal of the Spanish Economic Association, 8(4), pp. 345-371. https://doi.org/10.1007/s13209-017-0166-9

- Caggese, Andrea, and Vicente Cuñat. (2008). "Financing Constraints and Fixed-term Employment Contracts". *The Economic Journal*, 118(533), pp. 2013-2046. https://doi.org/10.1111/j.1468-0297.2008.02200.x
- Cahuc, Pierre, Helene Benghalem, Olivier Charlot, Emeline Limon and Franck Malherbet. (2016). "Taxation of temporary jobs: Good intentions with bad outcomes?". IZA Discussion Paper, 10352, IZA Institute of Labor Economics. https://doi.org/10.2139/ssrn.2868318
- Cahuc, Pierre, Pauline Carry, Franck Malherbet and Pedro Martins. (2022). "Employment Effects of Restricting Fixed-Term Contracts: Theory and Evidence". IZA Discussion Paper, 14999, IZA Institute of Labor Economics. https://doi.org/10.2139/ssrn.4114523
- Cahuc, Pierre, Olivier Charlot and Franck Malherbet. (2016). "Explaining the Spread of Temporary Jobs and its Impact on Labor Turnover". *International Economic Review*, 57(2), pp. 522-572. https://doi.org/10.1111/iere.12167
- Cahuc, Pierre, and Fabien Postel-Vinay. (2002). "Temporary jobs, employment protection and labor market performance". *Labour Economics*, 9(1), pp. 63-91. https://doi.org/10.1016/S0927-5371(01)00051-3
- Cao, Shutao, Enchuan Shao and Pedro Silos. (2013). Match Quality, Labor Market Protection, and Temporary Employment. https://pedrosilos.net/images/ftpc_MAR_2013.pdf
- Carrillo-Tudela, Carlos, Hermann Gartner and Leo Kaas. (2023). "Recruitment Policies, Job-Filling Rates and Matching Efficiency". *Journal of the European Economic Association*, 21(6), pp. 2413-2459. https://doi.org/10.1093/jeea/jvad034
- Carry, Pauline. (2022). The Effects of the Legal Minimum Working Time on Workers, Firms and the Labor Market. https://bcf.princeton.edu/wp-content/uploads/2023/01/JMP CARRY-2.pdf
- Chava, Sudheer, Alexander Oettl and Manpreet Singh. (2019). "Does a one-size-fits-all minimum wage cause financial stress for small businesses?". Working Paper Series, 26523, National Bureau of Economic Research. https://doi.org/10.3386/w26523
- Coles, Melvyn, and Dale T. Mortensen. (2016). "Equilibrium Labor Turnover, Firm Growth, and Unemployment". *Econometrica*, 84(1), pp. 347-363. https://doi.org/10.3982/ECTA10700
- Costain, James, Juan F. Jimeno and Carlos Thomas. (2010). "Employment fluctuations in a dual labor market". Documentos de Trabajo, 1013, Banco de España. https://repositorio.bde.es/ handle/123456789/7011
- Daruich, Diego. (2023). "The Macroeconomic Consequences of Early Childhood Development Policies". Working Paper Series, 2018-029B, Federal Reserve Bank of St. Louis. https://doi.org/10.20955/ wp.2018.029
- Daruich, Diego, Sabrina Di Addario and Raffaele Saggio. (2023). "The effects of partial employment protection reforms: Evidence from Italy". *Review of Economic Studies*, 90(6), pp. 2880-2942. https:// doi.org/10.1093/restud/rdad012
- Davis, Steven J., John C. Haltiwanger and Scott Schuh. (1998). Job Creation and Destruction. The MIT Press.
- Dolado, Juan J., Florentino Felgueroso and Juan F. Jimeno. (2000). "Youth Labour Markets in Spain: Education, Training, and Crowding-out". *European Economic Review*, 44(4), pp. 943-956. https://doi.org/10.1016/S0014-2921(99)00050-1
- Dustmann, Christian, Attila Lindner, Uta Schönberg, Matthias Umkehrer and Philipp vom Berge. (2021). "Reallocation Effects of the Minimum Wage". *The Quarterly Journal of Economics*, 137(1), pp. 267-328. https://doi.org/10.1093/qje/qjab028

- Dustmann, Christian, and Costas Meghir. (2005). "Wages, Experience and Seniority". *Review of Economic Studies*, 72(1), pp. 77-108. https://doi.org/10.1111/0034-6527.00325
- Elsby, Michael W. L., and Axel Gottfries. (2022). "Firm Dynamics, On-The-Job Search and Labor Market Fluctuations". *Review of Economic Studies*, 89(3), pp. 1370-1419. https://doi.org/10.1093/restud/rdab054
- Elsby, Michael W. L., and Ryan Michaels. (2013). "Marginal Jobs, Heterogeneous Firms, and Unemployment Flows". *American Economic Journal: Macroeconomics*, 5(1), pp. 1-48. https://doi.org/10.1257/mac.5.1.1
- Garcia-Louzao, Jose, Laura Hospido and Alessandro Ruggieri. (2023). "Dual returns to experience". Labour Economics, 80, 102290. https://doi.org/10.1016/j.labeco.2022.102290
- García-Perea, Pilar, Aitor Lacuesta and Pau Roldan-Blanco. (2021). "Markups and cost structure: Small Spanish firms during the Great Recession". *Journal of Economic Behavior & Organization*, 192, pp. 137-158. https://doi.org/10.1016/j.jebo.2021.10.001
- Gavazza, Alessandro, Simon Mongey and Giovanni L. Violante. (2018). "Aggregate Recruiting Intensity". *American Economic Review*, 108(8), pp. 2088-2127. https://doi.org/10.1257/aer.20161420
- Gouin-Bonenfant, Émilien. (2022). "Productivity dispersion, between-firm competition, and the labor share". *Econometrica*, 90(6), pp. 2755-2793. https://doi.org/10.3982/ECTA18612
- Gregory, Victoria, Guido Menzio and David Wiczer. (2022). "The Alpha Beta Gamma of the Labor Market". Working Paper Series, 28663, National Bureau of Economic Research. https://doi.org/10.20955/ wp.2021.003
- Gulyas, Andreas. (2023). The Puzzling Labor Market Sorting Pattern in Expanding and Contracting Firms. https://drive.google.com/file/d/1-RTpyOu0vLaWW6Go4Z9vEp0pTSGzwXkL/view
- Hopenhayn, H., and R. Rogerson. (1993). "Job turnover and policy evaluation: a general equilibrium analysis". *Journal of Political Economy*, 101(5), pp. 915-938. https://doi.org/10.1086/261909
- Kaas, Leo, and Philipp Kircher. (2015). "Efficient Firm Dynamics in a Frictional Labor Market". *American Economic Review*, 105(10), pp. 3030-3060. https://doi.org/10.1257/aer.20131702
- Ljungqvist, Lars, and Thomas J. Sargent. (2007). "Understanding European Unemployment with Matching and Search-Island Models". *Journal of Monetary Economics*, 54(8), pp. 2139-2179. https://doi. org/10.1016/j.jmoneco.2007.09.001
- Luca, Dara Lee, and Michael Luca. (2019). "Survival of the fittest: The impact of the minimum wage on firm exit". NBER Working Paper Series, 25806, National Bureau of Economic Research. https://doi.org/10.3386/w25806
- Mortensen, D., and C. Pissarides. (1994). "Job Creation and Job Destruction in the Theory of Unemployment". *Review of Economic Studies*, 61, pp. 397-415. https://doi.org/10.2307/2297896
- Moscarini, Giuseppe, and Fabien Postel-Vinay. (2013). "Stochastic Search Equilibrium". *Review of Economic Studies*, 80(4), pp. 1545-1581. https://doi.org/10.1093/restud/rdt012
- Petrongolo, Barbara, and Christopher A. Pissarides. (2001). "Looking into the Black Box: A Survey of the Matching Function". *Journal of Economic Literature*, 39(2), pp. 390-431. https://doi.org/10.1257/jel.39.2.390
- Roldan-Blanco, Pau, and Sonia Gilbukh. (2021). "Firm Dynamics and Pricing under Customer Capital Accumulation". *Journal of Monetary Economics*, 118, pp. 99-119. https://doi.org/10.1016/j. jmoneco.2020.01.011

- Ruiz-García, Juan Carlos. (2021). "Financial Frictions, Firm Dynamics and the Aggregate Economy: Insights from Richer Productivity Processes". Janeway Institute Working Papers, 2103, University of Cambridge. https://www.janeway.econ.cam.ac.uk/publication/jiwp2103
- Sala, Hector, José I. Silva and Manuel Toledo. (2012). "Flexibility at the Margin and Labor Market Volatility in OECD Countries". *The Scandinavian Journal of Economics*, 114(3), pp. 991-1017. https:// doi.org/10.1111/j.1467-9442.2012.01715.x
- Schaal, Edouard. (2017). "Uncertainty and Unemployment". *Econometrica*, 85(6), pp. 1675-1721. https://doi.org/10.3982/ECTA10557
- Tauchen, George. (1986). "Finite State Markov-Chain Approximations to Univariate and Vector Autoregressions". *Economics Letters*, 20(2), pp. 177-181. https://doi.org/10.1016/0165-1765(86)90168-0
- Thomas, Carlos. (2006). Firing costs, labor market search and the business cycle. London School of Economics.
- Topel, Robert. (1991). "Specific capital, mobility, and wages: Wages rise with job seniority". *Journal of Political Economy*, 99(1), pp. 145-176. https://doi.org/10.1086/261744

Dual Labor Markets and the Equilibrium Distribution of Firms

by Josep Pijoan-Mas and Pau Roldan-Blanco

Appendix Materials (for Online Publication Only)

Table of Contents

A	Additional Theoretical Results	OA - p.2
	A.1 Firm's and Joint Surplus Problem Equivalence	OA - p.2
	A.2 Equilibrium Distributions of Firms and Workers	OA - p.7
	A.3 Cobb-Douglas Matching Function	OA - p.11
	A.4 Worker Flow Rates	OA - p.12
	A.5 Aggregate Matching Function and Aggregate Matching Efficiency .	OA - p.15
	A.6 Aggregate Welfare	OA - p.16
В	Productivity and Misallocation	OA - p.16
	B.1 Output-Maximizing Benchmark Allocation of Workers to Firms	OA - p.16
	B.2 Misallocation	OA - p.19
	B.3 Productivity Decomposition	OA - p.20
С	Model Extension with Directed Entry	OA - p.21
D	Numerical Appendix	OA - p.22
	D.1 Idiosyncratic Productivity	OA - p.22
	D.2 Stationary Solution Algorithm	OA - p.23
Ε	Additional Figures and Tables	OA - p.25

A Additional Theoretical Results

A.1 Firm's and Joint Surplus Problem Equivalence

In order to show the equivalence, first we write the employed worker's and the firm's problem.

Employed Worker's Problem Given a menu of contracts $\vec{C} = \{w_{ij}, v_i, \delta_{ij}, p, W'_{ij}(\vec{n}', z') : (i, j) \in \mathcal{I} \times \mathcal{J}\}$ currently paid by the firm, the value of a worker of type (i, j) satisfies the following HJB equation:

$$\rho \mathbf{W}_{ij}(\vec{n}, z, \varphi; \vec{C}) = w_{ij} + (\delta_{ij} + s_i^{W} + s^{F}) \left(\mathbf{U} - \mathbf{W}_{ij}(\vec{n}, z, \varphi; \vec{C}) \right)$$
Same type co-worker separates:

$$+ (n_{ij} - 1) (\delta_{ij} + s_i^{W}) \left(W'_{ij}(\vec{n}_{ij}, z) - \mathbf{W}_{ij}(\vec{n}, z, \varphi; \vec{C}) \right)$$
Different type co-worker separates:

$$+ \sum_{(i',j') \neq (i,j)} n_{i'j'} (\delta_{i'j'} + s_{i'}^{W}) \left(W'_{ij}(\vec{n}_{i'L}, z) - \mathbf{W}_{ij}(\vec{n}, z, \varphi; \vec{C}) \right)$$
Firm hires with either contract:

$$+ \sum_{i' \in \mathcal{I}} v_{i'} \eta \left(W'_{i'L}(\vec{n}_{i'L}, z) \right) \left(W'_{ij}(\vec{n}_{i'L}, z) - \mathbf{W}_{ij}(\vec{n}, z, \varphi; \vec{C}) \right)$$
FT worker is promoted:

$$+ n_{FT} p \left(W_{ij}^{p}(\vec{n}^{p}, z) - \mathbf{W}_{ij}(\vec{n}, z, \varphi; \vec{C}) \right)$$
Skill upgrade for OE low-types:

$$+ n_{OEL} \tau \left(W_{ij}^{\tau}(\vec{n}^{\tau}, z) - \mathbf{W}_{ij}(\vec{n}, z, \varphi; \vec{C}) \right)$$
(A.1)

The right-hand side of this equation incorporates the different sources of value for the employed worker. On the first line, the worker gets a wage w_{ij} , and the option value from separation into unemployment (second additive term), either because she gets laid off (at rate δ_{ij}), or her contract expires (at rate s_i^W), or because the firm exits (at rate s^F). The second and third lines include the change in value from the separation of a co-worker that was employed under the same contract and job type (second line), or under the other contract type, or job type, or both (third line). The fourth line is the change in value due to the firm hiring an additional worker for a low-type job with a contract of type i'. The fifth and sixth lines of equation (A.1) refer to the change in value when either a promotion or skill accumulation take place, where we have defined:

$$\begin{split} W_{ij}^{p}(\vec{n}^{p},z) &\equiv \begin{cases} \frac{1}{n_{FT}} \Big(W_{OEL}'(\vec{n}^{p},z) + (n_{FT}-1) W_{FT}'(\vec{n}^{p},z) \Big) & \text{if } i = FT, \ \forall j \in \{L,H\} \\ W_{OE,j}'(\vec{n}^{p},z) & \text{if } i = OE, \ \forall j \in \{L,H\} \\ \end{cases} \\ W_{ij}^{\tau}(\vec{n}^{\tau},z) & \text{if } i = FT, \ \forall j \in \{L,H\} \\ \frac{1}{n_{OEL}} \Big(W_{OEH}'(\vec{n}^{\tau},z) + (n_{OEL}-1) W_{OEL}'(\vec{n}^{\tau},z) \Big) & \text{if } (i,j) = (OE,L) \\ W_{OEH}'(\vec{n}^{\tau},z) & \text{if } (i,j) = (OE,H) \end{cases} \end{split}$$

Finally, the last line of equation (A.1) includes the change in value due to a productivity shock.

Firm's Problem An operating firm of type $\varphi \in \Phi$ in state (\vec{n}, z, \vec{W}) must choose a menu of contracts $\vec{C}_{ij} = \{w_{ij}, v_i, \delta_{ij}, p, W'_{ij}(\vec{n}', z')\}$ for each $(i, j) \in \mathcal{I} \times \mathcal{J}$, where recall that $\vec{W} = \{W_{ij}\}$ denotes the set outstanding

promises to its current workers. Let $J(\vec{n}, z, \varphi, \vec{W})$ be the value of this firm. The HJB equation is:

$$\rho \mathbf{J}(\vec{n}, z, \varphi, \vec{W}) = \max_{\{w_{ij}, v_i, \delta_{ij}, p, W'_{ij}(\vec{n}', z')\}} \left\{ y(\vec{n}, z, \varphi) - \xi n_{FT} p^{\vartheta} + s^F \left(\mathbf{J}^e - \mathbf{J}(\vec{n}, z, \varphi, \vec{W}) \right) \right.$$

$$Wage bill, firing and recruiting costs: + \sum_{i \in \mathcal{I}} \left[\sum_{j \in \mathcal{J}} \left(-w_{ij} n_{ij} - \chi n_{ij} \delta^{\vartheta}_{ij} - A_i v_i^{\xi} + n_{ij} (\delta_{ij} + s_i^W) \left(\mathbf{J}(\vec{n}_{ij}, z, \varphi, \vec{W}'(\vec{n}_{ij}, z)) - \mathbf{J}(\vec{n}, z, \varphi, \vec{W}) \right) \right) \right]$$

$$Hiring under contract i: + v_i \eta \left(W'_{iL}(\vec{n}_{iL}^+, z) \right) \left(\mathbf{J}(\vec{n}_{iL}^+, z, \varphi, \vec{W}'(\vec{n}_{iL}^+, z)) - \mathbf{J}(\vec{n}, z, \varphi, \vec{W}) \right) \right]$$

$$FT workers promoted: + n_{FT} p \left(\mathbf{J}(\vec{n}^p, z, \varphi, \vec{W}'(\vec{n}^p, z)) - \mathbf{J}(\vec{n}, z, \varphi, \vec{W}) \right) \\$$

$$Froductivity shock: + \sum_{z' \in \mathcal{Z}} \lambda(z'|z) \left(\mathbf{J}(\vec{n}, z', \varphi, \vec{W}'(\vec{n}, z')) - \mathbf{J}(\vec{n}, z, \varphi, \vec{W}) \right) \right\}, \quad (A.2)$$

where J^e is a firm's value of having no workers (satisfying equation (12)). Problem (A.2) is subject to two constraints:

$$\mathbf{W}_{ij}(\vec{n}, z, \varphi; \vec{C}) \geq W_{ij}, \tag{A.3a}$$

$$W'_{ij}(\vec{n}',z') \ge \mathbf{U}, \quad \forall (\vec{n}',z'),$$
 (A.3b)

for all $(i, j) \in \mathcal{I} \times \mathcal{J}$. Constraint (A.3a) is a *promise-keeping* constraint: the firm must deliver an expected value to each worker (left-hand side) that is no lower than the outstanding promise (right-hand side). This constraint is in place because of the firm's initial commitment to the posted contracts. Constraint (A.3b) is a *worker-participation* constraint: for every possible future state (\vec{n}', z') , the value that each worker obtains cannot be lower than the outside option. This constraint is in place because workers do not commit, and must therefore be enticed to remain matched.

With these equations in place, we are now ready to prove the equivalence result (Proposition 1), which we spell more precisely as follows:

Proposition 1 The firm's and joint surplus problems are equivalent, in the following sense:

1. For any set of contracts \vec{C} that solves problem (A.2)-(A.3a)-(A.3b), the subset

$$\vec{C}_{\Sigma} \equiv \left\{ v_i, \delta_{ij}, p, W_{iL}'(\vec{n}_{iL}^+, z) \right\}_{(i,j)\in\mathcal{I}\times\mathcal{J}} \subset \vec{C}$$

solves problem (6).

2. Conversely, if \vec{C}_{Σ} constitutes a solution to problem (6), then there exists a unique set of wages and continuation promises $\vec{C}_{W} \equiv \{w_{ij}, W'_{ij}(\vec{n}_{ij}, z), W'_{ij}(\vec{n}^{p}, z), W'_{ij}(\vec{n}^{\tau}, z), \{W'_{ij}(\vec{n}, z')\}_{z' \in \mathcal{Z}}\}_{(i,j) \in \mathcal{I} \times \mathcal{J}}$ such that $\vec{C}_{W} \cup \vec{C}_{\Sigma}$ solves

problem (A.2)-(A.3a)-(A.3b).

Proof of Proposition 1 Let $\vec{C} = \{\vec{C}_{ij}\}$, with $\vec{C}_{ij} = \{\overline{w}_{ij}, \overline{v}_i, \overline{\delta}_{ij}, \overline{p}, \overline{W}'_{ij}(\vec{n}', z')\}$, denote a given policy. Then, we can re-write the problem of a type- φ firm in (A.2)-(A.3a)-(A.3b) as follows:

$$\mathbf{J}(\vec{n},z,\varphi,\vec{W}) = \max_{\vec{C}} \, \widetilde{\mathbf{J}}(\vec{n},z,\varphi,\vec{W} \mid \vec{C}), \quad \text{subject to} \begin{cases} \mathbf{W}_{ij}\left(\vec{n},z,\varphi;\vec{C}\right) \ge W_{ij}, & \forall(i,j) \\ \overline{W}'_{ij}(\vec{n}',z') \ge \mathbf{U}, & \forall(\vec{n}',z'), \; \forall(i,j) \end{cases}$$

where:

$$\begin{split} \widetilde{\mathbf{J}}\left(\vec{n},z,\varphi,\vec{W} \mid \vec{C}\right) &\equiv \frac{1}{\overline{\rho}\left(\vec{n},z,\varphi\mid\vec{C}\right)} \left\{ y(\vec{n},z,\varphi) - \xi n_{FT} \overline{p}^{\vartheta} + \sum_{i \in \mathcal{I}} \left[-A_i \overline{v}_i^{\varsigma} + \sum_{j \in \mathcal{J}} \left(-\overline{w}_{ij} n_{ij} - \chi n_{ij} \overline{\delta}_{ij}^{\psi} + n_{ij} \left(\overline{\delta}_{ij} + s_i^{W} \right) \mathbf{J}\left(\vec{n}_{ij}^{-}, z, \varphi, \vec{W}'(\vec{n}_{ij}^{-}, z)\right) \right) + \overline{v}_i \eta \left(\overline{W}_{iL}'(\vec{n}_{iL}^{+}, z) \right) \mathbf{J}\left(\vec{n}_{iL}^{+}, z, \varphi, \vec{W}'(\vec{n}_{iL}^{+}, z)\right) \right] \\ &+ \overline{p} n_{FT} \mathbf{J}\left(\vec{n}^{p}, z, \varphi, \vec{W}'(\vec{n}^{p}, z)\right) + n_{OEL} \tau \mathbf{J}\left(\vec{n}^{\tau}, z, \varphi, \vec{W}'(\vec{n}^{\tau}, z)\right) \\ &+ \sum_{z' \in \mathcal{Z}} \lambda(z'|z) \mathbf{J}\left(\vec{n}, z', \varphi, \vec{W}'(\vec{n}, z')\right) \right\} \end{split}$$
(A.4)

and where we have defined:

$$\overline{\rho}\left(\vec{n}, z, \varphi | \vec{C}\right) \equiv \rho + s^{F} + n_{FT}\overline{p} + n_{OEL}\tau + \sum_{i \in \mathcal{I}} \left[\overline{v}_{i}\eta\left(\overline{W}_{iL}'(\vec{n}_{iL}^{+}, z)\right) + \sum_{j \in \mathcal{J}} n_{ij}\left(\overline{\delta}_{ij} + s_{i}^{W}\right) \right]$$

as the effective discount rate of the firm. By monotonicity of preferences, if \vec{C} is optimal, then the promisekeeping constraint (A.3a) must hold with equality: $\mathbf{W}_{ij}\left(\vec{n}, z, \varphi; \vec{C}\right) = W_{ij}, \forall (i, j) \in \mathcal{I} \times \mathcal{J}.^{47}$ Imposing this into equation (A.1) allows us to solve for wages:

$$\overline{w}_{ij} = \overline{\rho} \left(\vec{n}, z, \varphi | \vec{C} \right) W_{ij} - \left[\left(\overline{\delta}_{ij} + s_i^W + s^F \right) \mathbf{U} + (n_{ij} - 1) \left(\overline{\delta}_{ij} + s_i^W \right) \overline{W}'_{ij}(\vec{n}_{ij}, z) + \sum_{(i',j') \neq (i,j)} n_{i'j'}(\overline{\delta}_{i'j'} + s_{i'}^W) \overline{W}'_{ij}(\vec{n}_{i'j''}, z) \right. \\
\left. + n_{FT} \overline{p} \left(\mathbf{1} [i = FT] \frac{\overline{W}'_{OEL}(\vec{n}^p, z) + (n_{FT} - 1) \overline{W}'_{FT}(\vec{n}^p, z)}{n_{FT}} + \mathbf{1} [i = OE] \overline{W}'_{OE,j}(\vec{n}^p, z) \right) \\
\left. + n_{OEL} \tau \left(\mathbf{1} [(i,j) = (OE,L)] \frac{\overline{W}'_{OEH}(\vec{n}^\tau, z) + (n_{OEL} - 1) \overline{W}'_{OEL}(\vec{n}^\tau, z)}{n_{OEL}} + \mathbf{1} [(i,j) \neq (OE,L)] \overline{W}'_{ij}(\vec{n}^\tau, z) \right) \\
\left. + \sum_{i' \in \mathcal{I}} \overline{v}_{i'} \eta \left(\overline{W}'_{i'L}(\vec{n}_{i'L}^+, z) \right) \overline{W}'_{ij}(\vec{n}_{i'L}^+, z) + \sum_{z' \in \mathcal{Z}} \lambda(z' | z) \overline{W}'_{ij}(\vec{n}, z') \right]$$
(A.5)

⁴⁷Suppose not. Then, the firm could increase its value by offering a combination of flow and continuation payoffs to the workers that would yield lower value to them and still comply with the firm's initial promises, a contradiction with \vec{C} being optimal.

where $\mathbf{1}[\cdot]$'s are indicator functions. Define the joint surplus under policy \vec{C} as:

$$\widetilde{\Sigma}\left(\vec{n}, z, \varphi, \vec{W} \middle| \vec{C}\right) \equiv \widetilde{J}\left(\vec{n}, z, \varphi, \vec{W} \middle| \vec{C}\right) + \sum_{i \in \mathcal{I}} \sum_{j \in \mathcal{J}} n_{ij} W_{ij}$$
(A.6)

Likewise, define the *maximized* joint surplus as:

$$\Sigma(\vec{n}, z, \varphi, \vec{W}) \equiv \max_{\vec{C}} \left\{ \widetilde{\Sigma}\left(\vec{n}, z, \varphi, \vec{W} \middle| \vec{C} \right), \text{ such that } \overline{W}'_{ij}(\vec{n}', z') \ge \mathbf{U}, \ \forall (\vec{n}', z'), \ \forall (i, j) \right\}$$
(A.7)

Plugging (A.5) inside (A.4) and collecting terms:

$$\begin{split} \widetilde{J}(\overrightarrow{n},z,\varphi,\overrightarrow{W} \mid \overrightarrow{C}) &+ \sum_{i \in \mathbb{Z}} \sum_{j \in \mathcal{J}} n_{ij} W_{ij} = \frac{1}{\overrightarrow{\rho}\left(\overrightarrow{n},z,\varphi|\overrightarrow{C}\right)} \left\{ y(\overrightarrow{n},z,\varphi) - \widetilde{\zeta} n_{FT} \overrightarrow{\rho}^{\theta} - \sum_{i \in \mathbb{Z}} A_i \overrightarrow{v}_i \zeta \right. \\ &+ \sum_{i \in \mathbb{Z}} \sum_{j \in \mathcal{J}} \left(n_{ij}(\overrightarrow{\delta}_{ij} + s_i^W + s^F) \mathbf{U} - \chi n_{ij} \overrightarrow{\delta}_{ij}^{\theta} \right) \\ &+ \sum_{i \in \mathbb{Z}} \sum_{j \in \mathcal{J}} n_{ij}(\overrightarrow{\delta}_{ij} + s_i^W) \underbrace{\left(\mathbf{J}(\overrightarrow{n}_{ij}, z, \varphi, \overrightarrow{W}'(\overrightarrow{n}_{ij}, z)) + (n_{ij} - 1) \overrightarrow{W}'_{ij}(\overrightarrow{n}_{ij}, z) + \sum_{(i',j') \neq (i,j)} n_{i'j'} \overrightarrow{W}'_{i'j'}(\overrightarrow{n}_{i'j'}, z) \right) \right) \\ &+ \sum_{i \in \mathbb{Z}} \sum_{j \in \mathcal{J}} n_{ij} \left(\sum_{i' \in \mathbb{Z}} \overrightarrow{v}_{i'} \eta\left(\overrightarrow{W}'_{i'L}(\overrightarrow{n}_{iL}^+, z) \right) \overrightarrow{W}'_{ij}(\overrightarrow{n}_{i'L}^+, z) \right) + \overline{v}_{i'} \eta\left(\overrightarrow{W}'_{iL}(\overrightarrow{n}_{iL}^+, z) \right) \mathbf{J}(\overrightarrow{n}_{iL}^+, z, \varphi, \overrightarrow{W}'(\overrightarrow{n}_{iL}^+, z)) \right) \\ &+ \sum_{i \in \mathbb{Z}} \left(\int_{i \in \mathcal{J}} n_{ij} \left(\sum_{i' \in \mathbb{Z}} \overrightarrow{v}_{i'} \eta\left(\overrightarrow{W}'_{i'L}(\overrightarrow{n}_{iL}^+, z) \right) \overrightarrow{W}'_{ij}(\overrightarrow{n}_{i'L}^+, z) \right) + (n_{OEL} + 1) \overrightarrow{W}'_{OEL}(\overrightarrow{n}^{\mu}, z) \right) \\ &+ n_{FT} \overrightarrow{p} \left(\int_{i' \in \mathbb{Z}} (\overrightarrow{n}^{\tau}, z, \varphi, \overrightarrow{W}'(\overrightarrow{n}^{\tau}, z)) + (n_{OEL} - 1) \overrightarrow{W}_{OEL}(\overrightarrow{n}^{\tau}, z) + (n_{OEH} + 1) \overrightarrow{W}'_{OEH}(\overrightarrow{n}^{\tau}, z) \right) \\ &+ n_{OEL} \tau \left(\int_{i' \in \mathbb{Z}} \lambda(z'|z) \left(\int_{i' \in \mathbb{Z}} (\overrightarrow{n}, z', \varphi, \overrightarrow{W}'(\overrightarrow{n}, z')) + \sum_{i \in \mathbb{Z}} \sum_{j \in \mathcal{J}} n_{ij} \overrightarrow{W}'_{ij}(\overrightarrow{n}, z') \right) \right) \right) \right\}$$
(A.8)

where the terms in blue follow from definitions (A.6) and (A.7). The term labeled [*] can be written as follows:

$$[*] = \sum_{i \in \mathcal{I}} \left[\sum_{j \in \mathcal{J}} n_{ij} \left(\sum_{i' \in \mathcal{I}} \overline{v}_{i'} \eta \left(\overline{W}'_{i'L}(\vec{n}^+_{i'L}, z) \right) \overline{W}'_{ij}(\vec{n}^+_{i'L}, z) \right) + \overline{v}_i \eta \left(\overline{W}'_{iL}(\vec{n}^+_{iL}, z) \right) \mathbf{J}(\vec{n}^+_{iL}, z, \varphi, \vec{W}'(\vec{n}^+_{iL}, z)) \right]$$

$$= \sum_{i \in \mathcal{I}} \overline{v}_{i} \eta \left(\overline{W}_{iL}^{\prime}(\vec{n}_{iL}^{+}, z) \right) \left[\mathbf{J} \left(\vec{n}_{iL}^{+}, z, \varphi, \vec{W}^{\prime}(\vec{n}_{iL}^{+}, z) \right) + n_{iL} \overline{W}_{iL}^{\prime}(\vec{n}_{iL}^{+}, z) + \sum_{(i^{\prime}, j^{\prime}) \neq (i, L)} n_{i^{\prime}j^{\prime}} \overline{W}_{i^{\prime}j^{\prime}}^{\prime}(\vec{n}_{iL}^{+}, z) \right]$$

$$= \sum_{i \in \mathcal{I}} \overline{v}_{i} \eta \left(\overline{W}_{iL}^{\prime}(\vec{n}_{iL}^{+}, z) \right) \left[\mathbf{J} \left(\vec{n}_{iL}^{+}, z, \varphi, \vec{W}^{\prime}(\vec{n}_{iL}^{+}, z) \right) + (n_{iL} + 1) \overline{W}_{iL}^{\prime}(\vec{n}_{iL}^{+}, z) + \sum_{(i^{\prime}, j^{\prime}) \neq (i, L)} n_{i^{\prime}j^{\prime}} \overline{W}_{i^{\prime}j^{\prime}}^{\prime}(\vec{n}_{iL}^{+}, z) - \overline{W}_{iL}^{\prime}(\vec{n}_{iL}^{+}, z) \right]$$

$$= \sum_{i \in \mathcal{I}} \overline{v}_{i} \eta \left(\overline{W}_{iL}^{\prime}(\vec{n}_{iL}^{+}, z) \right) \mathbf{\Sigma} \left(\vec{n}_{iL}^{+}, z, \varphi, \vec{W}^{\prime}(\vec{n}_{iL}^{+}, z) \right) - \sum_{i \in \mathcal{I}} \overline{v}_{i} \eta \left(\overline{W}_{iL}^{\prime}(\vec{n}_{iL}^{+}, z) \right) \overline{W}_{iL}^{\prime}(\vec{n}_{iL}^{+}, z)$$

Putting this back into equation (A.8), we find:

$$\begin{split} \widetilde{\Sigma}\left(\vec{n},z,\varphi,\vec{W}\middle|\vec{C}\right) &= \frac{1}{\overline{\rho}\left(\vec{n},z,\varphi\middle|\vec{C}\right)} \left\{ y(\vec{n},z,\varphi) - \xi n_{FT} \overline{p}^{\vartheta} - \sum_{i\in\mathcal{I}} A_i \overline{v}_i^{\varsigma} + \sum_{i\in\mathcal{I}} \sum_{j\in\mathcal{J}} \left(n_{ij} (\overline{\delta}_{ij} + s_i^W + s^F) \mathbf{U} - \chi n_{ij} \overline{\delta}_{ij}^{\psi} \right) \right. \\ &\left. - \overline{v}_i \eta \left(\overline{W}_{iL}'(\vec{n}_{iL}^+,z) \right) \overline{W}_{iL}'(\vec{n}_{iL}^+,z) + n_{ij} (\overline{\delta}_{ij} + s_i^W) \Sigma(\vec{n}_{ij}^-,z,\varphi,\vec{W}'(\vec{n}_{ij}^-,z)) \right. \\ &\left. + \overline{v}_i \eta \left(\overline{W}_{iL}'(\vec{n}_{iL}^+,z) \right) \Sigma(\vec{n}_{iL}^+,z,\varphi,\vec{W}'(\vec{n}_{iL}^+,z)) \right) + n_{FT} \overline{p} \Sigma(\vec{n}^p,z,\varphi,\vec{W}'(\vec{n}^p,z)) \right. \\ &\left. + n_{OEL} \tau \Sigma(\vec{n}^\tau,z,\varphi,\vec{W}'(\vec{n}^\tau,z)) + \sum_{z'\in\mathcal{Z}} \lambda(z'|z) \Sigma(\vec{n},z',\varphi,\vec{W}'(\vec{n},z')) \right\} \end{split}$$
(A.9)

Note that the right-hand side of (A.9) is independent of \vec{W} and w, so we can omit these as arguments of the Σ function in equation (A.9), and further simplify the equation into:

$$\widetilde{\Sigma}\left(\vec{n},z,\varphi\middle|\vec{C}_{\Sigma}\right) = \frac{1}{\overline{\rho}\left(\vec{n},z,\varphi\middle|\vec{C}\right)} \left\{ y(\vec{n},z,\varphi) - \xi n_{FT} \overline{p}^{\vartheta} - \sum_{i\in\mathcal{I}} A_i \overline{v}_i^{\varsigma} + \sum_{i\in\mathcal{I}} \sum_{j\in\mathcal{J}} \left(n_{ij} (\overline{\delta}_{ij} + s_i^W + s^F) \mathbf{U} - \chi n_{ij} \overline{\delta}_{ij}^{\psi} - \overline{v}_i \eta \left(\overline{W}_{iL}'(\vec{n}_{iL}^+,z) \right) \overline{W}_{iL}'(\vec{n}_{iL}^+,z) + n_{ij} (\overline{\delta}_{ij} + s_i^W) \Sigma\left(\vec{n}_{ij}^-,z,\varphi\right) + \overline{v}_i \eta \left(\overline{W}_{iL}'(\vec{n}_{iL}^+,z) \right) \Sigma\left(\vec{n}_{iL}^+,z,\varphi\right) \right) + n_{FT} \overline{p} \Sigma(\vec{n}^p,z,\varphi) + n_{OEL} \tau \Sigma(\vec{n}^\tau,z,\varphi) + \sum_{z'\in\mathcal{Z}} \lambda(z'|z) \Sigma(\vec{n},z',\varphi) \right\}$$
(A.10)

Thus, out of the full set $\vec{C} = \left\{ \overline{w}_{ij}, \overline{v}_i, \overline{\delta}_{ij}, \overline{p}, \overline{W}'_{ij}(\vec{n}', z') \right\}_{(i,j)\in \mathcal{I}\times \mathcal{J}}$, only the elements

$$\vec{C}_{\Sigma} \equiv \left\{ \overline{v}_i, \overline{\delta}_{ij}, \overline{p}, \overline{W}'_{iL}(\vec{n}^+_{iL}, z) \right\}_{(i,j) \in \mathcal{I} \times \mathcal{J}} \subset \vec{C}$$

matter for the joint surplus. The optimal contract is then:

$$\vec{C}_{\Sigma}^{*} = \arg\max_{\vec{C}_{\Sigma}} \left\{ \widetilde{\Sigma} \left(\vec{n}, z, \varphi \middle| \vec{C}_{\Sigma} \right) \text{ s.t. } W_{iL}'(\vec{n}_{iL}^{+}, z) \ge \mathbf{U}, \ \forall i \in \mathcal{I} \right\}$$
(A.11)

Wages $\{\overline{w}_{ij}\}_{(i,j)\in\mathcal{I}\times\mathcal{J}}$ are given by equation (A.5), while the remaining continuation values are chosen to be consistent with the solution to problem (A.11). Summing up: by expressing the firm's problem in terms of continuation promises, we have shown that the optimal contract must maximize the joint surplus. Conversely,

for any combination of continuation promises that maximizes the joint surplus, there is a unique wage and set of outstanding promises that maximizes the firm's value subject to the promise-keeping constraint.

A.2 Equilibrium Distributions of Firms and Workers

A.2.1 Preliminaries

Let $f_t(\vec{n}, z, \varphi) > 0$ be the measure of type- φ firms with employment vector $\vec{n} = \{n_{ij}\} \in \mathcal{N}$ and idiosyncratic productivity $z \in \mathcal{Z}$ at time $t \in \mathbb{R}_+$. Further, let $F_t \equiv \sum_{\vec{n} \in \mathcal{N}} \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} f_t(\vec{n}, z, \varphi)$ be the total measure of operating firms, and let $F_t^e > 0$ be the measure of potential entrants. Both F_t and F_t^e are endogenous objects, and are determined in equilibrium to be consistent with the sorting patterns of firms and workers.

Firms in state (\vec{n}, z, φ) seek to hire new workers of type (i, L) in market segment $W'_{iL}(\vec{n}^+_{iL}, z, \varphi)$ for each contract $i \in \mathcal{I}$. Let $\theta_i(\vec{n}^+_{iL}, z, \varphi) \equiv \theta \left(W'_{iL}(\vec{n}^+_{iL}, z, \varphi) \right)$ denote the equilibrium tightness in the submarket where firms of type (\vec{n}, z, φ) look to hire an additional (low-skill) worker under contract *i*. By equation (4), we have

$$\theta_i(\vec{n}_{iL}^+, z, \varphi) = \mu^{-1} \left(\frac{\rho \mathbf{U} - b}{W_{iL}'(\vec{n}_{iL}^+, z, \varphi) - \mathbf{U}} \right)$$
(A.12)

By definition, market tightness must adjust to guarantee that:

$$u_{it}(\vec{n}_{iL}^{+}, z, \varphi)\theta_{i}(\vec{n}_{iL}^{+}, z, \varphi) = v_{i}(\vec{n}, z, \varphi)f_{t}(\vec{n}, z, \varphi)$$
(A.13)

for all *t*, where $u_{it}(\vec{n}_{iL}^+, z, \varphi)$ is the measure of unemployed workers looking to be hired in a type-*i* contract by a firm in state (\vec{n}, z, φ) , and $v_i(\vec{n}, z, \varphi)$ denotes this firm's recruiting effort.⁴⁸ As the job-filling rate for an individual firm in state (\vec{n}, z, φ) is $v_i(\vec{n}, z, \varphi)\eta(\theta_i(\vec{n}_{iL}^+, z, \varphi))$, and there exist $f_t(\vec{n}, z, \varphi)$ firms with this identical job-filling rate, equation (A.13) simply states that, in equilibrium, the total number of vacancies created by firms in a given state equals the total number of vacancies found by workers in the corresponding submarket, or

$$\underbrace{\mu\left(\theta_{i}(\vec{n}_{iL}^{+}, z, \varphi)\right)}_{\text{Job-finding rate}} \underbrace{u_{it}(\vec{n}_{iL}^{+}, z, \varphi)}_{\# \text{ job seekers}} = \underbrace{v_{i}(\vec{n}, z, \varphi)\eta\left(\theta_{i}(\vec{n}_{iL}^{+}, z, \varphi)\right)}_{\text{Job-filling rate per firm}} \underbrace{\mathbf{f}_{t}(\vec{n}, z, \varphi)}_{\# \text{ firms}}$$
(A.14)

Next, denote by $e_{ij,t}(\vec{n}, z, \varphi)$ the measure of workers of type (i, j) employed by firms of type (\vec{n}, z, φ) at time *t*. The assignment of workers to firms satisfies:

$$\mathbf{e}_{ij,t}(\vec{n}, z, \varphi) = n_{ij} \mathbf{f}_t(\vec{n}, z, \varphi) \tag{A.15}$$

for every state. The unit measure of workers must be either matched with a firm or unmatched and searching. This gives us a formula for the unemployment rate, $U_t = 1 - E_t$, where:

$$\mathbf{E}_t = \sum_{i \in \mathcal{I}} \sum_{j \in \mathcal{J}} \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} n_{ij} \mathbf{f}_t(\{n_{ij}\}, z, \varphi).$$
(A.16)

⁴⁸Recall that each firm posts one vacancy of type *i* at each time *t*, so v_i can be interpreted both as the recruiting effort of an individual firm, as well as the *effective* measure of vacancies per firm.

A.2.2 Kolmogorov Forward Equations

Next, we write down the dynamics of the distribution of firms, which satisfy a set of Kolmogorov forward equations. The law of motion for the measure of firms in state (\vec{n}, z, φ) is given by

$$\frac{\partial \mathbf{f}_{t}(\vec{n}, z, \varphi)}{\partial t} + o(\vec{n}, z, \varphi) \mathbf{f}_{t}(\vec{n}, z, \varphi) = \sum_{i \in \mathcal{I}} v_{i}(\vec{n}_{iL}^{-}, z, \varphi) \eta \left(\theta_{i}(\vec{n}, z, \varphi)\right) \mathbf{f}_{t}(\vec{n}_{iL}^{-}, z, \varphi)
+ \sum_{i \in \mathcal{I}} \sum_{j \in \mathcal{J}} (n_{ij} + 1) \left(\delta_{ij}(\vec{n}_{ij}^{+}, z, \varphi) + s_{i}^{W}\right) \mathbf{f}_{t}(\vec{n}_{ij}^{+}, z, \varphi)
+ (n_{FT} + 1) p(\vec{n}_{p}^{-}, z, \varphi) \mathbf{f}_{t}(\vec{n}_{p}^{-}, z, \varphi)
+ (n_{OEL} + 1) \tau \mathbf{f}_{t}(\vec{n}_{\tau}^{-}, z, \varphi)
+ \sum_{\hat{z} \neq z} \lambda(z|\hat{z}) \mathbf{f}_{t}(\vec{n}, \hat{z}, \varphi),$$
(A.17)

with $\theta_i(\cdot)$ given by equation (A.12), and

$$o(\vec{n}, z, \varphi) \equiv s^{F} + \sum_{i \in \mathcal{I}} v_{i}(\vec{n}, z, \varphi) \eta \left(\theta_{i}(\vec{n}_{iL}^{+}, z, \varphi) \right) + \sum_{i \in \mathcal{I}} \sum_{j \in \mathcal{J}} n_{ij} \left(\delta_{ij}(\vec{n}, z, \varphi) + s_{i}^{W} \right) + n_{FT} p(\vec{n}, z, \varphi) + n_{OEL} \tau + \sum_{\hat{z} \neq z} \lambda(\hat{z}|z)$$

is the outflow rate. The right-hand side of equation (A.17) gives all the inflows into state (\vec{n}, z, φ) . Inflows come from firms with $\vec{n}_{iL}^- \in \{(n_{OEH}, n_{OEL}, n_{FT} - 1), (n_{OEH}, n_{OEL} - 1, n_{FT})\}$ that hire a worker with contract *i*, firms with $\vec{n}_{ij}^+ \equiv (n_{ij} + 1, \vec{n}_{-(ij)})$ that fire a type-(i, j) worker, firms with $\vec{n}_p^- \equiv (n_{OEH}, n_{OEL} - 1, n_{FT} + 1)$ that promote an FT worker into an OEC, firms with $\vec{n}_{\tau}^- \equiv (n_{OEH} - 1, n_{OEL} + 1, n_{FT})$ for whom a low-skill OE worker has a skill upgrade, and firms at $\vec{n} = (n_{ij}, \vec{n}_{-(ij)})$ that transition into productivity *z* from some $\hat{z} \neq z$.⁴⁹

On the other hand, the dynamics of potential entrants are given by

$$\frac{\partial \mathbf{F}_{t}^{e}}{\partial t} + \mathbf{F}_{t}^{e} \sum_{\varphi \in \Phi} \sum_{z^{e} \in \mathcal{Z}} \pi_{\varphi}(\varphi) \pi_{z}(z^{e}) \sum_{i \in \mathcal{I}} \eta \left(\theta_{i}(\vec{n}_{iL}^{e}, z^{e}, \varphi) \right) = s^{F} \mathbf{F}_{t} + \sum_{\varphi \in \Phi} \sum_{z \in \mathcal{Z}} \sum_{i \in \mathcal{I}} \sum_{j \in \mathcal{J}} \left(\delta_{ij}(\vec{n}_{ij}^{e}, z, \varphi) + s_{i}^{W} \right) \mathbf{f}_{t}(\vec{n}_{ij}^{e}, z, \varphi)$$
(A.18)

where, on the right-hand side, the first additive term are inflows from exiting firms and the second term are inflows from firms with $\vec{n}_{ij}^e \equiv (n_{ij}, \vec{n}_{-(ij)}) = (1, \vec{0})$ that lose their last worker. On the left-hand side, the second additive term collects outflows from successful entrants.

A.2.3 Finding the Stationary Distribution

To find the stationary distribution, we impose $\frac{\partial f_t(\vec{n},z,\varphi)}{\partial t} = \frac{\partial F_t^e}{\partial t} = 0$ into equations (A.17) and (A.18), and solve for the resulting system of equations. In the numerical implementation, we solve for the *share*, not the measure, of firms in each state $s \equiv (n_{OEH}, n_{OEL}, n_{FT}, z, \varphi)$. States take values in $\overline{\mathcal{N}}_{OEH} \times \overline{\mathcal{N}}_{OEL} \times \overline{\mathcal{N}}_{FT} \times \{z_1, \ldots, z_{k_z}\} \times \{\varphi_1, \varphi_2\}$, where we denote $\overline{\mathcal{N}}_{ij} \equiv \{0, 1, 2, \ldots, N_{ij}\}$, for some positive integers $\{N_{ij}\}$. Then, there is one inactive state, where $n_{OEH} = n_{OEL} = n_{FT} = 0$, which we label s = 0, and $S \equiv [(N_{OEH} + 1) \cdot (N_{OEL} + 1) \cdot (N_{OEL} + 1) \cdot (N_{FT} + 1) - 1] \cdot k_z \cdot 2$ active states. Denote by $\phi_s \in [0, 1]$ the share of firms in state $s = 1, \ldots, S$, such that $\sum_{s=1}^{S} \phi_s = 1 - \phi_0$, where $\phi_0 > 0$ is the share of potential entrant firms. Stacking all of these shares into the

⁴⁹Inflows from hiring must be multiplied by $\pi_z(z)\pi_{\varphi}(\varphi)$ whenever they come from successful entrants, i.e. for $(n_{iL}, n_H) = (1, 0), \forall i.$

column vector $\vec{\phi} \equiv [\phi_0, \phi_1, \dots, \phi_s]^{\top}$, we have a system of S + 1 non-linear equations, which in matrix form reads:

$$\mathbf{G}^{\top}\vec{\phi} = \vec{0} \tag{A.19}$$

The object **G** is a (S + 1)-dimensional infinitesimal generator matrix, which registers inflows in the diagonal and outflows in the off-diagonal, such that:

$$\mathbf{G} = \begin{pmatrix} -\sum_{s \neq 0} t_{0,s} & t_{0,1} & t_{0,2} & \dots & t_{0,S} \\ t_{1,0} & -\sum_{s \neq 1} t_{1,s} & t_{1,2} & \dots & t_{1,S} \\ t_{2,0} & t_{2,1} & -\sum_{s \neq 2} t_{2,s} & \dots & t_{2,S} \\ \vdots & \vdots & \ddots & \vdots \\ t_{5,0} & t_{5,1} & t_{5,2} & \dots & -\sum_{i \neq S} t_{5,i} \end{pmatrix}$$

The transition rates $t_{s,s'}$ are built using the optimal policies, following the laws of motion stated in equations (A.17) and (A.18). To solve for $\vec{\phi}$, we write system (A.19) as $(\mathbf{G}^{\top} + \mathbf{I})\vec{\phi} = \vec{\phi}$, where \mathbf{I} is a (S + 1)-dimensional identity matrix. This means that $\vec{\phi}$ can be computed as the eigenvector of $\mathbf{G}^{\top} + \mathbf{I}$ that is associated with the unit eigenvalue. We exploit this fact to find our invariant distribution.

In our calibration, we have S = 79,350, so **G** and **I** are very large matrices. However, as the vast majority of transitions are illegal, **G** has many zero entries, so in practice we define (**G**, **I**) as sparse matrices to be able to save on computing time and memory.

A.2.4 Aggregate Measures of Agents

Having found the invariant distribution, we make the following normalization:

$$\widetilde{\mathtt{f}}_s\equiv\frac{\phi_s}{1-\phi_0}$$

for each state s = 1, ..., S corresponding to a point $(n_{OEH}, n_{OEL}, n_{FT}, z, \varphi)$ in the state space. In words, \tilde{f}_s is the share of firms in state *s* relative to all operating firms, i.e. $\tilde{f}(\vec{n}, z, \varphi) \equiv f(\vec{n}, z, \varphi)/F$. To proceed, we use identity (A.15) to compute:

$$\widetilde{\mathbf{e}}_{ij}(\vec{n}, z, \varphi) = n_{ij}\widetilde{\mathbf{f}}(\vec{n}, z, \varphi)$$

That is, $\tilde{\mathbf{e}}_{ij} \equiv \mathbf{e}_{ij}/\mathbf{F}$ is the measure of workers of type (i, j) employed in firms of type (\vec{n}, z, φ) , as a share of the total measure of operating firms. From this we can find $\tilde{\mathbf{E}}_{ij} \equiv \mathbf{E}_{ij}/\mathbf{F} = \sum_{\vec{n} \in \mathcal{N}} \sum_{z} \sum_{\varphi} \tilde{\mathbf{e}}_{ij}(\vec{n}, z, \varphi)$, i.e. the total measure of employed individuals of type (i, j) per firm, and $\tilde{\mathbf{E}} \equiv \mathbf{E}/\mathbf{F} = \sum_{i \in \mathcal{I}} \sum_{j \in \mathcal{J}} \tilde{\mathbf{E}}_{ij}$, i.e. the average firm size.

To save on notation, let $f_i^v(\vec{n}, z, \varphi) \equiv v_i(\vec{n}, z, \varphi) f(\vec{n}, z, \varphi)$ denote the aggregate measure of effective vacancies available for contract *i* in a given state (\vec{n}, z, φ) . Similarly, let $\tilde{f}_i^v(\vec{n}, z, \varphi) \equiv f_i^v(\vec{n}, z, \varphi)/F$. Using (A.13), we know:

$$\mathbf{u}_{OE}(n_{OEH}, n_{OEL} + 1, n_{FT}, z, \varphi) = \mathbf{F} \frac{\tilde{\mathbf{f}}_{OE}^{v}(n_{OEH}, n_{OEL}, n_{FT}, z, \varphi)}{\theta_{OE}(n_{OEH}, n_{OEL} + 1, n_{FT}, z, \varphi)}$$
$$\mathbf{u}_{FT}(n_{OEH}, n_{OEL}, n_{FT} + 1, z, \varphi) = \mathbf{F} \frac{\tilde{\mathbf{f}}_{FT}^{v}(n_{OEH}, n_{OEL}, n_{FT}, z, \varphi)}{\theta_{FT}(n_{OEH}, n_{OEL}, n_{FT} + 1, z, \varphi)}$$

To make progress, in the first equation we add up both sides across all $(n_{OEH}, n_{FT}, z, \varphi)$, as well as over $n_{OEL} = 1, ..., N_{OE}$ (i.e. omitting $n_{OEL} = 0$). Similarly, in the second equation we add across all $(n_{OEH}, n_{OEL}, z, \varphi)$, as well as over $n_{FT} = 1, ..., N_{FT}$ (i.e. omitting $n_{FT} = 0$). That is:

$$\sum_{n_{OEH}} \sum_{n_{OEH}} \sum_{n_{OEL} \neq 0} \sum_{n_{FT}} \sum_{z} \sum_{\varphi} u_{OE}(n_{OEH}, n_{OEL} + 1, n_{FT}, z, \varphi) = F\left(\sum_{n_{OEH}} \sum_{n_{OEL} \neq 0} \sum_{n_{FT}} \sum_{z} \sum_{\varphi} \frac{\tilde{f}_{OE}^v(n_{OEH}, n_{OEL}, n_{FT}, z, \varphi)}{\theta_{OE}(n_{OEH}, n_{OEL} + 1, n_{FT}, z, \varphi)}\right)$$
(A.20)
$$\sum_{n_{OEH}} \sum_{n_{OEL}} \sum_{n_{FT} \neq 0} \sum_{z} \sum_{\varphi} u_{FT}(n_{OEH}, n_{OEL}, n_{FT} + 1, z, \varphi) = F\left(\sum_{n_{OEH}} \sum_{n_{OEL}} \sum_{n_{FT} \neq 0} \sum_{z} \sum_{\varphi} \frac{\tilde{f}_{OE}^v(n_{OEH}, n_{OEL}, n_{FT}, z, \varphi)}{\theta_{FT}(n_{OEH}, n_{OEL}, n_{FT} + 1, z, \varphi)}\right)$$
(A.21)

Developing the left-hand side of (A.20) and (A.21) yields:

$$\begin{split} \sum_{n_{OEH}} \sum_{n_{OEL} \neq 0} \sum_{n_{FT}} \sum_{z} \sum_{\varphi} u_{OE}(n_{OEH}, n_{OEL} + 1, n_{FT}, z, \varphi) &= U_{OE} - \sum_{n_{OEH}} \sum_{n_{FT}} \sum_{z} \sum_{\varphi} u_{OE}(n_{OEH}, 1, n_{FT}, z, \varphi) \\ &= \underbrace{1 - E - U_{FT}}_{= U_{OE}} - \sum_{z} \sum_{\varphi} \left(\frac{F^{e}}{\theta_{OE}(0, 1, 0, z, \varphi)} + \sum_{\substack{(n_{OEH}, n_{FT}) \neq (0, 0)}} \frac{f_{OE}^{v}(n_{OEH}, 0, n_{FT}, z, \varphi)}{\theta_{OE}(n_{OEH}, 1, n_{FT}, z, \varphi)} \right) \\ &= 1 - U_{FT} - F \left[\widetilde{E} + \sum_{z} \sum_{\varphi} \left(\frac{\widetilde{F}^{e}}{\theta_{OE}(0, 1, 0, z, \varphi)} + \sum_{\substack{(n_{OEH}, n_{FT}) \neq (0, 0)}} \frac{f_{OE}^{v}(n_{OEH}, 0, n_{FT}, z, \varphi)}{\theta_{OE}(n_{OEH}, 1, n_{FT}, z, \varphi)} \right) \right] \end{split}$$

and

$$\begin{split} \sum_{n_{OEH}} \sum_{n_{OEL}} \sum_{n_{FT} \neq 0} \sum_{z} \sum_{\varphi} u_{FT}(n_{OEH}, n_{OEL}, n_{FT} + 1, z, \varphi) &= U_{FT} - \sum_{n_{OEH}} \sum_{n_{OEL}} \sum_{z} \sum_{\varphi} u_{FT}(n_{OEH}, n_{OEL}, 1, z, \varphi) \\ &= U_{FT} - \sum_{z} \sum_{\varphi} \left(\frac{F^{e}}{\theta_{FT}(0, 0, 1, z, \varphi)} + \sum_{(n_{OEH}, n_{OEL}) \neq (0, 0)} \frac{f_{FT}^{v}(n_{OEH}, n_{OEL}, 0, z, \varphi)}{\theta_{FT}(n_{OEH}, n_{OEL}, 1, z, \varphi)} \right) \\ &= U_{FT} - F \left[\sum_{z} \sum_{\varphi} \left(\frac{\widetilde{F}^{e}}{\theta_{FT}(0, 0, 1, z, \varphi)} + \sum_{(n_{OEH}, n_{OEL}) \neq (0, 0)} \frac{\widetilde{f}_{FT}^{v}(n_{OEH}, n_{OEL}, 0, z, \varphi)}{\theta_{FT}(n_{OEH}, n_{OEL}, 1, z, \varphi)} \right) \right] \end{split}$$

respectively. Substituting these back into equations (A.20) and (A.21) yields:

$$1 - \mathbf{U}_{FT} - \mathbf{F} \left[\widetilde{\mathbf{E}} + \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} \left(\frac{\widetilde{\mathbf{F}}^e}{\theta_{OE}(0, 1, 0, z, \varphi)} + \sum_{(n_{OEH}, n_{FT}) \neq (0, 0)} \frac{\widetilde{\mathbf{f}}_{OE}^v(n_{OEH}, 0, n_{FT}, z, \varphi)}{\theta_{OE}(n_{OEH}, 1, n_{FT}, z, \varphi)} \right) \right]$$
$$= \mathbf{F} \left(\sum_{n_{OEH}} \sum_{n_{OEL} \neq 0} \sum_{n_{FT}} \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} \frac{\widetilde{\mathbf{f}}_{OE}^v(n_{OEH}, n_{OEL}, n_{FT}, z, \varphi)}{\theta_{OE}(n_{OEH}, n_{OEL} + 1, n_{FT}, z, \varphi)} \right)$$

and

$$\mathbf{U}_{FT} - \mathbf{F}\left[\sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} \left(\frac{\widetilde{\mathbf{F}}^e}{\theta_{FT}(0, 0, 1, z, \varphi)} + \sum_{(n_{OEH}, n_{OEL}) \neq (0, 0)} \frac{\widetilde{\mathbf{f}}_{FT}^v(n_{OEH}, n_{OEL}, 0, z, \varphi)}{\theta_{FT}(n_{OEH}, n_{OEL}, 1, z, \varphi)} \right) \right]$$

$$= \mathbb{F}\left(\sum_{n_{OEH}}\sum_{n_{OEL}}\sum_{n_{FT}\neq 0}\sum_{z\in\mathcal{Z}}\sum_{\varphi\in\Phi}\frac{\widetilde{\mathbb{f}}_{FT}^{v}(n_{OEH},n_{OEL},n_{FT},z,\varphi)}{\theta_{FT}(n_{OEH},n_{OEL},n_{FT}+1,z,\varphi)}\right)$$

Solving for F on each equation yields:

$$F = \frac{1 - U_{FT}}{\widetilde{E} + \widetilde{U}_{OE}}$$
 and $F = \frac{U_{FT}}{\widetilde{U}_{FT}}$ (A.22)

respectively, where we have defined:

$$\begin{split} \widetilde{\mathbf{U}}_{OE} &\equiv \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} \left[\frac{\widetilde{\mathbf{F}}^{e}}{\theta_{OE}(0, 1, 0, z, \varphi)} + \sum_{(n_{OEH}, n_{FT}) \neq (0, 0)} \frac{\widetilde{\mathbf{f}}_{OE}^{v}(n_{OEH}, 0, n_{FT}, z, \varphi)}{\theta_{OE}(n_{OEH}, 1, n_{FT}, z, \varphi)} \right] \\ &+ \sum_{n_{OEH}} \sum_{n_{OEL} \neq 0} \sum_{n_{FT}} \frac{\widetilde{\mathbf{f}}_{OE}^{v}(n_{OEH}, n_{OEL}, n_{FT}, z, \varphi)}{\theta_{OE}(n_{OEH}, n_{OEL} + 1, n_{FT}, z, \varphi)} \end{split}$$

and

$$\begin{split} \widetilde{\mathbf{U}}_{FT} \; \equiv \; \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} \left[\frac{\widetilde{\mathbf{F}}^{e}}{\theta_{FT}(0, 0, 1, z, \varphi)} \; + \sum_{(n_{OEH}, n_{OEL}) \neq (0, 0)} \frac{\widetilde{\mathbf{f}}_{FT}^{v}(n_{OEH}, n_{OEL}, 0, z, \varphi)}{\theta_{FT}(n_{OEH}, n_{OEL}, 1, z, \varphi)} \\ & + \sum_{n_{OEH}} \sum_{n_{OEL}} \sum_{n_{FT} \neq 0} \frac{\widetilde{\mathbf{f}}_{FT}^{v}(n_{OEH}, n_{OEL}, n_{FT}, z, \varphi)}{\theta_{FT}(n_{OEH}, n_{OEL}, n_{FT} + 1, z, \varphi)} \right] \end{split}$$

Solving for U_{FT} from equation (A.22) gives

$$U_{FT} = \frac{\widetilde{U}_{FT}}{\widetilde{E} + \widetilde{U}_{OE} + \widetilde{U}_{FT}}$$
(A.23)

From this, we can finally obtain the aggregate measure of operating firms as

$$\mathbf{F} = \left(\widetilde{\mathbf{E}} + \widetilde{\mathbf{U}}_{OE} + \widetilde{\mathbf{U}}_{FT}\right)^{-1}$$
(A.24)

Equation (A.24) reflects a feasibility condition: the sum of the measures of all employed and unemployed workers must equal one, the size of the labor force. Once we have the measure of operating firms F, we can compute all the remaining aggregates. Namely, (i) the total measure of potential entrants is $F^e = \phi_0 F$; (ii) the employment and unemployment rates are given by $E = \tilde{E}F$ and U = 1 - E; (iii) the aggregate temporary share is given by E_{FT}/E ; and (iv) the aggregate share of skilled workers (or "skill share") is E_{OEH}/E .

A.3 Cobb-Douglas Matching Function

Assume $M(V, U) = V^{\gamma}U^{1-\gamma}$. Using equation (8), some algebra shows:

$$W_{iL}'(\vec{n}_{iL}^+, z, \varphi) = \gamma \mathbf{U} + (1 - \gamma) \Big(\mathbf{\Sigma}(\vec{n}_{iL}^+, z, \varphi) - \mathbf{\Sigma}(\vec{n}, z, \varphi) \Big)$$
(A.25)

This expression is intuitive: the continuation promise $W'_{iL}(\vec{n}_{iL}+z)$ for a new worker under contract *i* is a weighted average of the worker's outside option, **U**, and the marginal net joint surplus gain from the hire,

 $\Sigma(\vec{n}_{iL}, z, \varphi) - \Sigma(\vec{n}, z, \varphi)$. The object $(1 - \gamma)$ gives the share of the overall gains (net of the outside option) that accrue to the new worker. On the other hand, and using the definition of joint surplus (equation (5)), the firm obtains the following change in value from hiring:

$$\mathbf{J}\left(\vec{n}_{iL}^{+}, z, \varphi, \vec{W}'(\vec{n}_{iL}^{+}, z, \varphi)\right) - \mathbf{J}\left(\vec{n}, z, \varphi, \vec{W}\right) = \underbrace{\gamma\left(\boldsymbol{\Sigma}(\vec{n}_{iL}^{+}, z, \varphi) - \boldsymbol{\Sigma}(\vec{n}, z, \varphi) - \mathbf{U}\right)}_{\text{New surplus, shared with new hire}} + \underbrace{\sum_{i' \in \mathcal{I}} \sum_{j \in \mathcal{J}} n_{i'j} \left(W_{i'j}'(\vec{n}, z, \varphi) - W_{i'j}'(\vec{n}_{iL}^{+}, z, \varphi)\right)}_{(A.26)}$$

Transfer of value between firm and pre-existing workers

The firm's marginal gain in value is composed of two terms. On the one hand, the firm absorbs the share γ of the total net gain in joint surplus that is not absorbed by the new hire. On the other hand, since we assume that all workers within the firm that are employed under the same contract must earn the same value, there must be a transfer of value between the firm and all its pre-existing workers (for *both* contract types) after hiring takes place. This is captured by the second additive term. Therefore, in equilibrium, firms must strike a balance between the surplus they extract from new hires and the surplus they extract from their pre-existing workers when a hire takes place.

The aggregate job-filling rate for type- ϕ firms in state (\vec{n}, z) can then be written as follows:

$$\eta(\vec{n}, z, \varphi) = \left[(1 - \gamma) \frac{\boldsymbol{\Sigma}(\vec{n}_{iL}^+, z, \varphi) - \boldsymbol{\Sigma}(\vec{n}, z, \varphi) - \mathbf{U}}{\rho \mathbf{U} - b} \right]^{\frac{1 - \gamma}{\gamma}}$$
(A.27)

Given our choice of $\gamma = 0.5$ (discussed below), equation (A.27) implies that the job-filling rate is linear in the ratio of the marginal net gain in joint surplus from a new match that accrues to a worker to the expected value of that worker's search. Therefore, in our calibrated model, conditional on a recruiting policy firms can expect to find workers at a rate that is proportional to the returns that the job delivers ex-post.

Finally, with Cobb-Douglas matching the free-entry condition (equations (12)-(13)), written in terms of joint surplus, reads:

$$\kappa = \gamma \left(\frac{1-\gamma}{\rho \mathbf{U}-b}\right)^{\frac{1-\gamma}{\gamma}} \left[\sum_{\varphi \in \Phi} \pi_{\varphi}(\varphi) \sum_{z^e \in \mathcal{Z}} \pi_z(z^e) \sum_{i \in \mathcal{I}} \left(\mathbf{\Sigma}(\vec{n}_{iL}^e, z^e, \varphi) - \mathbf{U}\right)^{\frac{1}{\gamma}}\right]$$
(A.28)

Note that the expected value of a single-worker firm is monotonically decreasing in the value of unemployment, because for higher **U** both the job-filling rate and the ex-post gains in match surplus that accrue to the firm are lower. This ensures that the equilibrium value of unemployment, **U**, is unique.

A.4 Worker Flow Rates

In the data We compute the EU and UE quarterly rates by type of contract using data from the *Encuesta de Población Activa* (EPA), the Labor Force Survey compiled by the *Instituto Nacional de Estadística* (INE), the Spanish national statistical agency. The data come at the quarterly frequency for the period 2006Q1-2019Q4. Denote by $UE_{t,t+1}^{i,data}$ the U-to-E flow from quarter t to t + 1 into a contract of type i = OE, FT, and similarly for $EU_{t,t+1}^{i,data}$. E-to-E flows from an FTC into an OEC are denoted by $EE_{t,t+1}^{FtoO,data}$. Labor market rates are then defined

as follows:

$$\widehat{UE}_{i,t}^{\text{data}} \equiv \frac{\sum UE_{t,t+1}^{i,\text{data}}}{\sum U_t^{\text{data}}} \quad \text{and} \quad \widehat{EU}_{i,t}^{\text{data}} \equiv \frac{\sum EU_{t,t+1}^{i,\text{data}}}{\sum E_t^{i,\text{data}}}$$

where \sum denotes the sum of sample weights for all observations in that category, $\sum U_t^{\text{data}}$ is the number of unemployed at time *t*, and $\sum E_t^{i,\text{data}}$ is the number of employed in contract type *i* at time *t*. Similarly, the promotion rate in the data is computed as follows:

$$\widehat{EE}_{FtoO,t}^{data} \equiv \frac{\sum EE_{t,t+1}^{FtoO,data}}{\sum E_{t}^{FT,data}}$$

where $E_t^{FT,\text{data}}$ is the stock of FT workers in quarter *t*. The values reported in Table 4 are time-series averages over the HP-filtered quarterly series $\left\{ \widehat{UE}_{i,t}^{\text{data}}, \widehat{EU}_{i,t}^{\text{data}}, \widehat{EE}_{FtoO,t}^{\text{data}} \right\}_{t=2005Q2}^{2019Q4}$.

In the model As we do not have empirical flows for within-firm skill upgrades, we categorize workers into four employment status: high-skill and low-skill employed with an OEC, employed with an FTC, and unemployed. The following set of equations describes flows between these states through the lens of the model:

$$\begin{split} \frac{\partial \mathbf{E}_{OEH}}{\partial t} &= \sum_{\vec{n}} \sum_{z} \sum_{\varphi} \left\{ \tau \mathbf{e}_{OEL}(\vec{n}, z, \varphi) - \left(\delta_{OEH}(\vec{n}, z, \varphi) + s_{OE}^{\mathsf{W}} + s^{F} \right) \mathbf{e}_{OEH}(\vec{n}, z, \varphi) \right\} \\ \frac{\partial \mathbf{E}_{OEL}}{\partial t} &= \sum_{\vec{n}} \sum_{z} \sum_{\varphi} \left\{ p(\vec{n}, z, \varphi) \mathbf{e}_{FT}(\vec{n}, z, \varphi) + \mu \left(\theta_{OE}(\vec{n}_{OEL}^{+}, z, \varphi) \right) \mathbf{u}_{OE}(\vec{n}_{OEL}^{+}, z, \varphi) \\ &- \left(\delta_{OEL}(\vec{n}, z, \varphi) + s_{OE}^{\mathsf{W}} + s^{F} \right) \mathbf{e}_{OEL}(\vec{n}, z, \varphi) - \tau \mathbf{e}_{OEL}(\vec{n}, z, \varphi) \right\} \\ \frac{\partial \mathbf{E}_{FT}}{\partial t} &= \sum_{\vec{n}} \sum_{z} \sum_{\varphi} \left\{ \mu \left(\theta_{FT}(\vec{n}_{FT}^{+}, z, \varphi) \right) \mathbf{u}_{FT}(\vec{n}_{FT}^{+}, z, \varphi) \\ &- \left(\delta_{FT}(\vec{n}, z, \varphi) + s_{FT}^{\mathsf{W}} + s^{F} \right) \mathbf{e}_{FT}(\vec{n}, z, \varphi) - p(\vec{n}, z, \varphi) \mathbf{e}_{FT}(\vec{n}, z, \varphi) \right\} \\ \frac{\partial \mathbf{U}}{\partial t} &= \sum_{\vec{n}} \sum_{z} \sum_{\varphi} \left\{ \sum_{j=L,H} \left(\delta_{OE,j}(\vec{n}, z, \varphi) + s_{OE}^{\mathsf{W}} + s^{F} \right) \mathbf{e}_{OE,j}(\vec{n}, z, \varphi) + \left(\delta_{FT}(\vec{n}, z, \varphi) + s_{FT}^{\mathsf{W}} + s^{F} \right) \mathbf{e}_{FT}(\vec{n}, z, \varphi) \right\} \end{split}$$

Each of these equations collect inflows (terms with a positive sign) outflows (terms with a negative sign). A more compact way of writing the dynamical system above is:

$$\frac{\partial \mathbf{E}_{OEH}}{\partial t} = -\lambda_{EU_{OEH}} \mathbf{E}_{OEH} + \lambda_{EE_{LH}} \mathbf{E}_{OEL}$$
(A.29a)

$$\frac{\partial \mathbf{E}_{OEL}}{\partial t} = -\left(\lambda_{EU_{OEL}} + \lambda_{EE_{LH}}\right) \mathbf{E}_{OEL} + \lambda_{EE_{FtoO}} \mathbf{E}_{FT} + \lambda_{UE_{OE}} \mathbf{U}$$
(A.29b)

$$\frac{\partial \mathbf{E}_{FT}}{\partial t} = -\left(\lambda_{EU_{FT}} + \lambda_{EE_{FtoO}}\right) \mathbf{E}_{FT} + \lambda_{UE_{FT}} \mathbf{U}$$

$$\frac{\partial \mathbf{U}}{\partial t} = \lambda_{EU_{OEH}} \mathbf{E}_{OEH} + \lambda_{EU_{OEL}} \mathbf{E}_{OEL} + \lambda_{EU_{FT}} \mathbf{E}_{FT} - \left(\lambda_{UE_{OE}} + \lambda_{UE_{FT}}\right) \mathbf{U}$$
(A.29d)
(A.29d)

where we have defined the following average intensities:

$$\lambda_{EU_{OE,j}} \equiv \frac{EU_{OE,j}}{E_{OE,j}} \qquad \lambda_{EE_{FtoO}} \equiv \frac{EE_{FtoO}}{E_{FT}} \qquad \lambda_{UE_{OE}} \equiv \frac{UE_{OE}}{U}$$
$$\lambda_{EU_{FT}} \equiv \frac{EU_{FT}}{E_{FT}} \qquad \lambda_{UE_{FT}} \equiv \frac{UE_{FT}}{U} \qquad \lambda_{EE_{LH}} \equiv \frac{EE_{LH}}{E_{OEL}}$$

with

$$EU_{OE,j} \equiv \sum_{\vec{n}} \sum_{z} \sum_{\varphi} \left(\delta_{OE,j}(\vec{n}, z, \varphi) + s_{OE}^{W} + s^{F} \right) \mathbf{e}_{OE,j}(\vec{n}, z, \varphi) \qquad EE_{FtoO} \equiv \sum_{\vec{n}} \sum_{z} \sum_{\varphi} p(\vec{n}, z, \varphi) \mathbf{e}_{FT}(\vec{n}, z, \varphi)$$
$$UE_{OE} \equiv \sum_{\vec{n}} \sum_{z} \sum_{\varphi} \mu \left(\theta_{OE}(\vec{n}_{OEL}^{+}, z, \varphi) \right) \mathbf{u}_{OE}(\vec{n}_{OEL}^{+}, z, \varphi) \qquad EU_{FT} \equiv \sum_{\vec{n}} \sum_{z} \sum_{\varphi} \left(\delta_{FT}(\vec{n}, z, \varphi) + s_{FT}^{W} + s^{F} \right) \mathbf{e}_{FT}(\vec{n}, z, \varphi)$$
$$UE_{FT} \equiv \sum_{\vec{n}} \sum_{z} \sum_{\varphi} \mu \left(\theta_{FT}(\vec{n}_{FT}^{+}, z, \varphi) \right) \mathbf{u}_{FT}(\vec{n}_{FT}^{+}, z, \varphi) \qquad EE_{LH} \equiv \sum_{\vec{n}} \sum_{z} \sum_{\varphi} \tau \mathbf{e}_{OEL}(\vec{n}, z, \varphi)$$

for j = H, L. Note, in particular, that since all firms face the same rate of skill upgrade, we have that $\lambda_{EE_{LH}} = \tau$. We can write system (A.29a)-(A.29d) in vector-matrix form as follows:

$$\frac{\partial}{\partial t} \begin{bmatrix} \mathbf{E}_{OEH} \\ \mathbf{E}_{OEL} \\ \mathbf{E}_{FT} \\ \mathbf{U} \end{bmatrix} = \begin{pmatrix} -\lambda_{EU_{OEH}} & \lambda_{EE_{LH}} & \mathbf{0} & \mathbf{0} \\ \mathbf{0} & -(\lambda_{EU_{OEL}} + \lambda_{EE_{LH}}) & \lambda_{EE_{FtoO}} & \lambda_{UE_{OE}} \\ \mathbf{0} & \mathbf{0} & -(\lambda_{EU_{FT}} + \lambda_{EE_{FtoO}}) & \lambda_{UE_{FT}} \\ \lambda_{EU_{OEH}} & \lambda_{EU_{OEL}} & \lambda_{EU_{FT}} & -(\lambda_{UE_{OE}} + \lambda_{EU_{FT}}) \end{pmatrix} \begin{bmatrix} \mathbf{E}_{OEH} \\ \mathbf{E}_{OEL} \\ \mathbf{E}_{FT} \\ \mathbf{U} \end{bmatrix}$$

Setting the right-hand side to the zero vector and solving the resulting system of linear equations will give us the stationary measures in the reduced-form model. Notice that, by construction, the resulting stocks coincide exactly, by construction, with the ones derived from firm-level flows in Online Appendix A.2.4.

Using these results, we can now construct flow rates. As the model is set in continuous time, we must produce discrete-time approximations in order to have numbers that can be compared to the ones from the quarterly data. For this, we compute for each contract type i = OE, *FT*:

$$\widehat{UE}_{i}^{\text{model}} = \frac{1 - e^{-UE_{i}dt}}{U} \quad \text{and} \quad \widehat{EU}_{i}^{\text{model}} = \frac{1 - e^{-\sum_{j=L,H} EU_{ij}dt}}{\sum_{j=L,H} E_{ij}}$$

In these ratios, in the numerator we have transformed instantaneous Poisson rates into quarterly probabilities by setting dt = 1/4.⁵⁰ For the overall UE and EU rates, we compute:

$$\widehat{UE}_{\text{total}}^{\text{model}} = \frac{1 - e^{-\sum_{i} UE_{i}dt}}{U} \quad \text{and} \quad \widehat{EU}_{\text{total}}^{\text{model}} = \frac{1 - e^{-\sum_{i} \sum_{j} EU_{ij}dt}}{\sum_{i} \sum_{j} E_{ij}}$$

⁵⁰In particular, the numerator is the probability that there is *at least* one transition (i.e. one or more transitions) within a given quarter, which we compute as the complementary probability of no transitions.

Similarly, to obtain the promotion rate at the quarterly frequency in the model, we compute:

$$\widehat{EE}_{FtoO}^{\mathrm{model}} = rac{1 - e^{-EE_{FtoO}dt}}{\mathrm{E}_{FT}}$$

For estimation purposes, we treat $\widehat{UE}_{i}^{\text{model}}$, $\widehat{EU}_{i}^{\text{model}}$ and $\widehat{EE}_{FtoO}^{\text{model}}$ as the direct model counterparts of $\widehat{UE}_{i}^{\text{data}}$, $\widehat{EU}_{i}^{\text{data}}$ and $\widehat{EE}_{FtoO}^{\text{data}}$, respectively.

A.5 Aggregate Matching Function and Aggregate Matching Efficiency

This appendix shows that our model with recruiting intensity generates endogenous differences in aggregate matching efficiency between contracts.⁵¹ Recall that a submarket is a pair (i, W), where $i \in \{FT, OE\}$ is the contract type and W is the promised utility. At any given time, a firm f posts one vacancy of each contract type i and chooses the intensity v_{if} with which to recruit into it. Assuming a Cobb-Douglas matching function, the number of matches in market (i, W) is $M_i(W) \equiv (V_i(W))^{\gamma} (U_i(W))^{1-\gamma}$, where $U_i(W)$ is the measure of unemployed workers seeking employment in submarket (i, W), and

$$\mathbf{v}_i(W) \equiv \int_{\Omega_i(W)} v_{if} \mathrm{d}f \tag{A.30}$$

is the effective measure of vacancies in submarket (i, W), where $\Omega_i(W)$ is the set of firms that offer value W for contract *i*. The total number of matches that occur for contract *i*, denoted M_i , can then be found by aggregating $M_i(W)$ across all promised values W. This can be written as

$$\mathbf{M}_{i} \equiv \int \mathbf{M}_{i}(W) \mathbf{d}W = \Gamma_{i} \mathbf{V}_{i}^{\gamma} \mathbf{U}_{i}^{1-\gamma}, \qquad (A.31)$$

where $V_i = F + F^e$ is the number of *actual* vacancies posted for contract *i* (equal to the sum of operating firms and potential entrants, as each such firm posts exactly one vacancy of each type), $U_i \equiv \int U_i(W) dW$ is the total number of job seekers for contract *i*, and Γ_i is the *aggregate matching efficiency* for contract *i*, defined by

$$\Gamma_i \equiv \int \left(\frac{\mathbf{v}_i(W)}{\mathbf{v}_i}\right)^{\gamma} \left(\frac{\mathbf{U}_i(W)}{\mathbf{U}_i}\right)^{1-\gamma} dW.$$
(A.32)

Notice that in a version of the model without endogenous recruiting intensity (e.g. $v_{if} = v$), differences in aggregate matching efficiency between markets would emerge only from the fact that firms and unemployed workers do not distribute uniformly across markets (as search is directed).⁵² With recruiting intensity, however, there is an added component affecting matching efficiency, stemming from underlying differences in recruiting costs across contracts, which operates over and above differences in promised values.

To see how directed search and recruiting costs both generate differences in matching efficiency, note that the *aggregate* job-finding and job-filling rates in contract *i* can be written as

$$\frac{M_i}{U_i} = \Gamma_i \Theta_i^{\gamma}$$
 and $\frac{M_i}{V_i} = \Gamma_i \Theta_i^{\gamma-1}$, (A.33)

 $^{^{51}}$ The intuitions are similar to those developed in Gavazza *et al.* (2018) and Carrillo-Tudela *et al.* (2023). Relative to the former, we explore aggregate matching intensity in the context of directed search. Relative to the latter, we introduce dual labor markets.

⁵²Indeed, in such an environment, we would have $\Gamma_i = 1$ if search was random.

respectively, where $\Theta_i \equiv V_i/U_i$ is the *aggregate market tightness* for contracts of type *i*. Therefore, we can compute the aggregate UE rate (the fraction of the unemployed that find a job with contract *i*) by

$$UE_i \equiv \frac{M_i}{U} = \Gamma_i \Theta_i^{\gamma} \frac{U_i}{U}, \qquad (A.34)$$

where $U \equiv \sum_{i} U_{i}$ is the total pool of unemployed workers. In the data, $UE_{FT} > UE_{OE}$. Note

$$\frac{UE_{FT}}{UE_{OE}} = \frac{\Gamma_{FT}}{\Gamma_{OE}} \left(\frac{\mathtt{U}_{FT}}{\mathtt{U}_{OE}}\right)^{1-\gamma}.$$
(A.35)

As OECs offer better deals ex-post, we have $U_{FT} < U_{OE}$. Therefore, in order to obtain $UE_{FT} > UE_{OE}$, it must be that $\Gamma_{FT} > \Gamma_{OE}$. In the calibration, this will be the result of recruiting costs being sufficiently low for FTCs compared to OECs, for given recruiting intensity (i.e. $A_{FT} < A_{OE}$).

In the stationary equilibrium of the model, all firms in a given state (\vec{n}, z, φ) deliver the same promised value and choose identical recruiting effort $v_i(\vec{n}, z, \varphi)$. Using (A.13), this allows us to write equation (A.32) as

$$\Gamma_{i} = \sum_{\vec{n}} \sum_{z} \sum_{\varphi} \left[\underbrace{\frac{v_{i}(\vec{n}, z, \varphi)}{\mathsf{V}_{i}}}_{\text{Relative recruiting intensity}} \left(\underbrace{\frac{\theta_{i}(\vec{n}, z, \varphi)}{\Theta_{i}}}_{\text{Relative market}} \right)^{1-\gamma} \mathbf{f}(\vec{n}, z, \varphi) \right].$$
(A.36)

Equation (A.36) says that the aggregate matching efficiency in contract *i* is determined by two components: (i) the relative recruiting intensity in each market, and (ii) the relative degree of market tightness in each market (which results from the fact that search is directed). How these translate into matching efficiency depends on the distribution of firms $\{f(\vec{n}, z, \varphi)\}$ and on the matching elasticity, γ .

A.6 Aggregate Welfare

We compute welfare in the stationary equilibrium as the present discounted sum of the value of production net of firing, recruiting, and promotion costs over all operating firms, plus the value of home production b across all unemployed workers, net of the total entry costs paid by all potential entrant firms, that is:

$$\mathcal{W} = \frac{1}{\rho} \Biggl\{ -\mathbf{F}^{e} \kappa + \mathbf{U}b + \sum_{\vec{n} \in \mathcal{N}} \sum_{\vec{z} \in \mathcal{Z}} \sum_{\varphi \in \Phi} \mathbf{f}(\vec{n}, z, \varphi) \times \dots$$

$$\cdots \times \Biggl[y(\vec{n}, z, \varphi) - \xi n_{FT} [p(\vec{n}, z, \varphi)]^{\vartheta} - \sum_{i \in \mathcal{I}} \left(A_{i} [v_{i}(\vec{n}, z, \varphi)]^{\varsigma} + \sum_{j \in \mathcal{J}} \chi n_{ij} [\delta_{ij}(\vec{n}, z, \varphi)]^{\psi} \right) \Biggr] \Biggr\}$$
(A.37)

B Productivity and Misallocation

B.1 Output-Maximizing Benchmark Allocation of Workers to Firms

Consider being able to freely allocate workers to firms in order to maximize output without being constrained by the search frictions present in the competitive equilibrium, but taking as given the distribution of firms across productivity types (z, φ) . In this section we show that, in such a benchmark allocation of workers, the marginal product of each type of labor is equalized across firms. Hence, the allocation of high and low skilled workers is identical across firms of the same productivity type, (z, φ) .

We explore two cases under which workers are allocated to firms of different productivities: (i) a case in which the measures of employed workers by skill type are fixed but total employment is not (Section B.1.1); and (ii) a case in which the total measure of employed workers is fixed but its split between skill types is not (Section B.1.2). This latter case is the one we discuss in the main text, and below in Section B.2. Once again, in both cases we assume that $F_{z,\varphi}$, the distribution of firms across productivity types (z, φ), is taken as given.

B.1.1 Benchmark Allocation of Workers when the Employment Measures by Skill are Fixed

For a given measure of operating firms $F_{z,\varphi}$ and given measures of high skilled and low skilled workers E_H and E_H , we can obtain the first-best allocation of workers $n_H^*(z,\varphi)$ and $n_L^*(z,\varphi)$ across skill types (z,φ) as the solution to the following problem:

$$\max_{\{n_H(z,\varphi),n_L(z,\varphi)\}} \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} y \Big(n_H(z,\varphi), n_L(z,\varphi), z, \varphi \Big) F_{z,\varphi} \quad \text{s.t.} \begin{cases} \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} n_H(z,\varphi) F_{z,\varphi} = E_H \\ \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} n_L(z,\varphi) F_{z,\varphi} = E_L \end{cases}$$

The FOC are given by $\partial y(n_H, n_L, z, \varphi) / \partial n_H = \lambda_H$ and $\partial y(n_H, n_L, z, \varphi) / \partial n_L = \lambda_L$, where $\lambda_H \ge 0$ and $\lambda_L \ge 0$ are the Lagrange multipliers associated to each constraint. Combining these two equations gives us the ratio of each type of worker,

$$\frac{n_H(z,\varphi;\lambda_H,\lambda_L)}{n_L(z,\varphi;\lambda_H,\lambda_L)} = \left(\frac{\omega(\varphi)}{1-\omega(\varphi)}\frac{\lambda_L}{\lambda_H}\right)^{\frac{1}{1-\alpha}}$$
(B.1)

which varies by φ but not by z, and varies by the aggregate relative scarcity of each type of worker given by λ_L/λ_H . Hence, all firms of the same type φ obtain the same ratio of high and low skilled workers. Let $E_H(\varphi) \equiv \sum_z n_H^*(z, \varphi) F_{z,\varphi}$ and $E_L(\varphi) \equiv \sum_z n_L^*(z, \varphi) F_{z,\varphi}$ be the total measures of high- and low-skilled workers allocated to firms of type φ . Then, using equation (B.1), the skill ratio at firms of type φ is given by

$$\frac{\mathsf{E}_{H}(\varphi)}{\mathsf{E}_{L}(\varphi)} = \left(\frac{\omega(\varphi)}{1-\omega(\varphi)}\frac{\lambda_{L}}{\lambda_{H}}\right)^{\frac{1}{1-\alpha}}$$

and aggregating over firms of type φ it must be that

$$\frac{\mathsf{E}_{H}}{\mathsf{E}_{L}} = \sum_{\varphi \in \Phi} \frac{\mathsf{E}_{H}(\varphi)}{\mathsf{E}_{L}(\varphi)} \frac{\mathsf{F}_{\varphi}}{\mathsf{F}} = \sum_{\varphi \in \Phi} \left(\frac{\omega(\varphi)}{1 - \omega(\varphi)} \frac{\lambda_{L}}{\lambda_{H}} \right)^{\frac{1}{1 - \alpha}} \frac{\mathsf{F}_{\varphi}}{\mathsf{F}}$$

where $F_{\varphi} \equiv \sum_{z} F_{z,\varphi}$. Therefore,

$$\left(\frac{\lambda_H}{\lambda_L}\right)^{\frac{1}{1-\alpha}} = \frac{\mathsf{E}_L}{\mathsf{E}_H} \sum_{\varphi \in \Phi} \left(\frac{\omega(\varphi)}{1-\omega(\varphi)}\right)^{\frac{1}{1-\alpha}} \frac{\mathsf{F}_{\varphi}}{\mathsf{F}}$$

Hence, using equation (B.1):

$$n_{HL}^{*}(\varphi) \equiv \frac{n_{H}^{*}(z,\varphi)}{n_{L}^{*}(z,\varphi)} = \frac{\left(\frac{\omega(\varphi)}{1-\omega(\varphi)}\right)^{\frac{1}{1-\alpha}}}{\sum_{\varphi' \in \Phi} \left(\frac{\omega(\varphi')}{1-\omega(\varphi')}\right)^{\frac{1}{1-\alpha}} \frac{\mathsf{F}_{\varphi'}}{\mathsf{F}}} \left(\frac{\mathsf{E}_{H}}{\mathsf{E}_{L}}\right)$$

Therefore, the skill share is a function of a firm's permanent type, but not of its idiosyncratic productivity: $h^*(\varphi) \equiv \frac{n_H^*(z,\varphi)}{n_H^*(z,\varphi) + n_L^*(z,\varphi)} = \left[1 + \frac{1}{n_{HL}^*(\varphi)}\right]^{-1}$. Next, using the FOC of say n_H we obtain the employment demand:

$$n(z,\varphi;\lambda_H) = \left[\frac{\nu}{\lambda_H}\omega(\varphi)\left(h^*(\varphi)\right)^{\alpha-1}\left(\omega(\varphi)\left(h^*(\varphi)\right)^{\alpha} + \left(1-\omega(\varphi)\right)\left(1-h^*(\varphi)\right)^{\alpha}\right)^{\frac{\nu}{\alpha}-1}e^{z+\zeta(\varphi)}\right]^{\frac{1}{1-\nu}}$$
(B.2)

To characterize $n^*(z, \varphi)$ we need to get rid of λ_H . To do so, we aggregate equation (B.2):

$$\mathbf{E}_{H} + \mathbf{E}_{L} = \left(\frac{\nu}{\lambda_{H}}\right)^{\frac{1}{1-\nu}} \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} \left[\omega(\varphi) \left(h^{*}(\varphi)\right)^{\alpha-1} \left(\omega(\varphi) \left(h^{*}(\varphi)\right)^{\alpha} + \left(1-\omega(\varphi)\right) \left(1-h^{*}(\varphi)\right)^{\alpha}\right)^{\frac{\nu}{\alpha}-1} e^{z+\zeta(\varphi)} \right]^{\frac{1}{1-\nu}} \mathbf{F}_{z,\varphi}$$

which gives us the total employment of firm (z, φ) :

$$n^{*}(z,\varphi) = \frac{\left(\mathsf{E}_{H} + \mathsf{E}_{L}\right)\left[\omega(\varphi)\left(h^{*}(\varphi)\right)^{\alpha-1}\left(\omega(\varphi)\left(h^{*}(\varphi)\right)^{\alpha} + \left(1 - \omega(\varphi)\right)\left(1 - h^{*}(\varphi)\right)^{\alpha}\right)^{\frac{\nu}{\alpha}-1}e^{z+\zeta(\varphi)}\right]^{\frac{1}{1-\nu}}}{\sum_{z'\in\mathcal{Z}}\sum_{\varphi'\in\Phi}\left[\omega(\varphi')\left(h^{*}(\varphi')\right)^{\alpha-1}\left(\omega(\varphi')\left(h^{*}(\varphi')\right)^{\alpha} + \left(1 - \omega(\varphi')\right)\left(1 - h^{*}(\varphi')\right)^{\alpha}\right)^{\frac{\nu}{\alpha}-1}e^{z'+\zeta(\varphi')}\right]^{\frac{1}{1-\nu}}\mathsf{F}_{z',\varphi'}}$$

B.1.2 Benchmark Allocation of Workers when the Employment Measures by Skill are Not Fixed

We can also characterize the output-maximizing allocation when the measure of operating firms $F_{z,\varphi}$ and total employment E are given, but the split of E into E_H and E_L is kept free. This would imply solving the problem:

$$\max_{\{n_H(z,\varphi),n_L(z,\varphi)\}} \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} y(n_H, n_L, z, \varphi) F_{z,\varphi} \quad \text{ s.t. } \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} n_H(z,\varphi) F_{z,\varphi} + \sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} n_L(z,\varphi) F_{z,\varphi} = E_H + E_L$$

Compared to our case above, we now only have one constraint, and therefore a single Lagrange multiplier, λ . The ratio of FOCs already gives the ratio of skilled to unskilled workers:

$$\frac{n_{H}^{**}(z,\varphi)}{n_{L}^{**}(z,\varphi)} = \left(\frac{\omega(\varphi)}{1-\omega(\varphi)}\right)^{\frac{1}{1-\alpha}}$$
(B.3)

which again implies that the skill share is only a function of permanent productivity. Then, following the same derivations as above, we obtain the total employment of a firm of type (z, φ) :

$$n^{*}(z,\varphi) = \frac{\left(\mathbf{E}_{H} + \mathbf{E}_{L}\right) \left[\omega(\varphi) \left(h^{*}(\varphi)\right)^{\alpha-1} \left(\omega(\varphi) \left(h^{*}(\varphi)\right)^{\alpha} + \left(1 - \omega(\varphi)\right) \left(1 - h^{*}(\varphi)\right)^{\alpha}\right)^{\frac{\nu}{\alpha} - 1} e^{z + \zeta(\varphi)}\right]^{\frac{1}{1-\nu}}}{\sum_{z' \in \mathcal{Z}} \sum_{\varphi' \in \Phi} \left[\omega(\varphi') \left(h^{*}(\varphi')\right)^{\alpha-1} \left(\omega(\varphi') \left(h^{*}(\varphi')\right)^{\alpha} + \left(1 - \omega(\varphi')\right) \left(1 - h^{*}(\varphi')\right)^{\alpha}\right)^{\frac{\nu}{\alpha} - 1} e^{z' + \zeta(\varphi')}\right]^{\frac{1}{1-\nu}} \mathbf{F}_{z',\varphi'}} (\mathbf{B.4})$$

B.2 Misallocation

B.2.1 Misallocation in the Baseline Calibration

The top two panels of Figure E.6 describe the allocation of the total number of workers across firm types φ (left vs. right panels) and transitory productivity states *z* (different crosses within each panel) for the competitive equilibrium (henceforth, "CE") of the calibrated economy (black schedule) and the Benchmark (red schedule) allocations, respectively. We see that the CE allocates too few workers to the most productivity firms: among the high types (φ_1), the market allocates an average of 20.5 workers to the highest productivity firms (z_5), while these same firms should employ on average 25.3 in an economy in which total output is maximized. This misallocation reflects the mean-reversion of productivity shocks and the rigidity of OECs. It is efficient for very productive type- φ_1 firms to employ many high-skilled workers. However, the good shocks may turn into bad shocks in the future, which in the CE would make it very hard for firms to get rid of redundant workers hired under OECs.

The bottom two panels in Figure E.6 describe the allocation of human capital (the ratio of high skilled workers to all workers) across firms. As discussed in the main text, the Benchmark's allocation is invariant in z and increases in φ . The CE fails to allocate the right amount of human capital across firms. In particular, while it does allocate more human capital to type- φ_1 firms, the allocation of human capital to productive type- φ_1 firms is way lower than the output-maximizing one. Type- φ_2 firms, by contrast, get a similar amount of human capital in both the CE and Benchmark. This uncovers and important static inefficiency of the menu of contracts available in the dual labor market economy. In particular, OE contacts are too rigid for type- φ_1 firms. For these firms, it would be efficient to face less worker turnover such that their employees have time to acquire more human capital. The tool to achieve this worker stability would be to employ more workers under OECs and less workers under FTCs. But OE workers are too rigid and expensive to undo whenever bad productivity shock arise, which leads to an inefficient provision of human capital.

B.2.2 Misallocation and Heterogeneous Effects in the Policy Counterfactual

Figure E.7 shows the percentage changes in total employment, the skill share, and the temporary share for firms of different permanent types and productivities when moving from the baseline calibration to the policy that limits the duration of FTCs to 1 month. When possible (i.e. for total employment and the skill share), we also show the policy response in the output-maximizing Benchmark economy (red lines).

First, we see that all low-type (φ_2) firms reduce employment. However, among them, highly productive (z_5) firms reduce it more than they should and unproductive (z_1) firms do not reduce it enough. In a nutshell,

the patterns of between-firm misallocation that we identified in the top panels of Figure E.6 get exacerbated. Second, though all firms increase their share of skilled workers, the adjustment is much more dramatic among low-type firms. However, as we saw in the bottom panels of Figure E.6, it is precisely the high-type firms that are furthest from their optimal allocation of human capital. Therefore, though beneficial for the overall levels of skill in the economy, the policy fails to sufficiently reallocate high-skill workers into the right firms. Finally, consistent with these results, all firms reduce their share of temporary employment, although the decrease is particularly pronounced among low-type firms (whose levels of temporary employment are overall much higher).

This changes in the allocation of workers across and within firms are reflected in changes in aggregate misallocation (see Table 6). Our results reveal that, as FTC duration decreases, all sources of misallocation worsen: the output loss goes from 3.91% in the baseline to 4.29% in the short duration policy, and total output declines by 6.86%. Interestingly, the contribution to these output losses by between-firm misallocation decreases as FTCs become of shorter duration. Namely, a policy that would make FTCs shorter would increase all forms of employment misallocation, but would hurt particularly strongly the allocation of workers between firms of the same type. Therefore, the declining output losses that we see as FTC duration increases are due to a better allocation of workers between firms of the same type.

B.3 Productivity Decomposition

In order to decompose aggregate productivity into the different parts shown in equation (20), we first need to transform equation (19) in two steps.

First, note that the measure of firms in state (n_H, n_L, z, φ) can always be written as $f(n_H, n_L, z, \varphi) = F_{z,\varphi} \tilde{f}_{z,\varphi}(n_H, n_L)$, where $\tilde{f}_{z,\varphi}(n_H, n_L)$ denotes the share of operating firms of type (z, φ) with employment vector (n_H, n_L) , and $F_{z,\varphi} \equiv \sum_{n_H} \sum_{n_L} f(n_H, n_L, z, \varphi)$ is the measure of firms in state (z, φ) , so that $\sum_z \sum_{\varphi} F_{z,\varphi} = F$. Then, we can write equation (19) as:

$$\frac{Y}{E} = \frac{F}{E} \left[\sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} \frac{F_{z,\varphi}}{F} \left(\sum_{n_H=0}^{+\infty} \sum_{n_L=0}^{+\infty} y(n_H, n_L, z, \varphi) \widetilde{f}_{z,\varphi}(n_H, n_L) \right) \right]$$

Second, given that the production function $y(n_H, n_L, z, \varphi)$ is homogeneous of degree ν in n_H and n_L , we can rewrite it as a function of total employment, $n \equiv n_H + n_L$, and the skill share in the firm, $h \equiv n_H/n$, so that

$$y(n_H, n_L, z, \varphi) = \underbrace{n^{\nu} y(h, 1-h, z, \varphi)}_{\equiv y^h(n, h, z, \varphi)}$$

Notice that, because of the discreteness of the state space, *h* takes values in the subset of rational numbers contained in the unit interval: if n = 1, then $h \in \mathcal{H}_1 \equiv \{0, 1\}$; if n = 2, then $h \in \mathcal{H}_2 \equiv \{0, 1/2, 1\}$; if n = 3, then $h \in \mathcal{H}_3 \equiv \{0, 1/3, 2/3, 1\}$; and so on. In general, we have $h \in \mathcal{H} \equiv \bigcup_{n \in \mathcal{N}} \mathcal{H}_n = \mathbb{Q} \cap [0, 1]$, where \mathbb{Q} is the set of all rational numbers.

To identify the size effect, instead of firm employment *n* we will use employment relative to average firm size, $\hat{n} \equiv n/(E/F)$, such that

$$y^{h}(n,h,z,\varphi) = \left(\frac{\mathsf{E}}{\mathsf{F}}\right)^{\nu} y^{h}_{z,\varphi}(\widehat{n},h)$$

Notice that, for given E/F, normalized firm employment \hat{n} is a random variable with discrete support, denoted

 $\widehat{\mathcal{N}}$. Given this, aggregate productivity Y/E can be written as:

$$\frac{Y}{E} = \left(\frac{F}{E}\right)^{1-\nu} \left[\sum_{z \in \mathcal{Z}} \sum_{\varphi \in \Phi} \frac{F_{z,\varphi}}{F} \left(\sum_{\widehat{n} \in \widehat{\mathcal{N}}} \sum_{h \in \mathcal{H}} y_{z,\varphi}^{h}(\widehat{n},h) g_{z,\varphi}(\widehat{n},h)\right)\right]$$
(B.5)

where

$$\mathbf{g}_{z,\varphi}(\widehat{n},h) \equiv \sum_{n_H=0}^{+\infty} \sum_{n_L=0}^{+\infty} \widetilde{\mathbf{f}}_{z,\varphi}(n_H,n_L) \, \mathbf{1}\left[\left(\frac{n_H+n_L}{\mathsf{E}/\mathsf{F}} = \widehat{n}\right) \wedge \left(\frac{n_H}{n_H+n_L} = h\right)\right] \tag{B.6}$$

is the share of firms of type (z, φ) that have relative size \hat{n} and skill share *h*. This is equivalent to the result shown in equation (20).

Finally, the joint probability $g_{z,\varphi}(\hat{n}, h)$ can be written as:

$$g_{z,\varphi}(\widehat{n},h) = \underbrace{g_{z,\varphi}^{A}(h|\widehat{n})}_{\text{Within component}} \underbrace{g_{z,\varphi}^{B}(\widehat{n})}_{\text{Between component}}$$

for some conditional probability mass function $g_{z,\varphi}^{A}(h|\hat{n})$ and some marginal probability mass function $g_{z,\varphi}^{B}(\hat{n})$. On the one hand, $g_{z,\varphi}^{A}(h|\hat{n})$ denotes the share of (z,φ) that have skill share h conditional on having relative size \hat{n} , so its contribution quantifies the within-firm, across worker component of productivity. On the other hand, $g_{z,\varphi}^{B}(\hat{n})$ denotes the share of firms of type (z,φ) that are of relative size \hat{n} , regardless of their skill share, so its contribution quantifies the between-firm component of productivity.

C Model Extension with Directed Entry

Using extreme value shocks, the directed entry problem in Section 5.1.4 obtains convenient closed-form solutions. In particular, let the shocks be distributed Gumbel, $\varepsilon_{\varphi} \sim G(\mu_{\varepsilon}, \sigma_{\varepsilon})$. Then, one can obtain: (i) an expression for the value of entry (equation (22)) as

$$\mathbf{J}^{e} = \sigma_{\varepsilon} \ln \left(\sum_{\varphi \in \Phi} \exp \left(\frac{\mathbf{J}^{e}(\varphi) - \kappa(\varphi)}{\sigma_{\varepsilon}} \right) \right) + \mu_{\varepsilon} + \sigma_{\varepsilon} \gamma_{\varepsilon}$$
(C.1)

where γ_{ε} is Euler's constant, and (ii) an expression for the fraction of entrants of each type as

$$\pi_{\varphi}(\varphi) = \frac{\exp\left(\frac{\mathbf{J}^{e}(\varphi) - \kappa(\varphi)}{\sigma_{\varepsilon}}\right)}{\sum_{\varphi' \in \Phi} \exp\left(\frac{\mathbf{J}^{e}(\varphi') - \kappa(\varphi')}{\sigma_{\varepsilon}}\right)}$$
(C.2)

To understand how selection upon entry works, note that the fraction of firms choosing technology φ increases with the value $\mathbf{J}^e(\varphi)$ of entering with technology φ , and that how much selection changes with the change of value is decreasing in the variance parameter σ_{ε} . In particular, for the case with only two values of φ one can easily write,

$$\frac{d\ln\left(\frac{\pi_{\varphi}(\varphi_1)}{\pi_{\varphi}(\varphi_2)}\right)}{d\left(\mathbf{J}^e(\varphi_1) - \mathbf{J}^e(\varphi_2)\right)} = \frac{1}{\sigma_{\varepsilon}}$$
(C.3)

Thus, $1/\sigma_{\varepsilon}$ gives us the semi-elasticity of the odds ratio of entering as a φ_1 type to the gain in value of entry as a type φ_1 . The *actual* entry elasticity is:

$$\mathcal{E}_{entry} \equiv \frac{d \ln \left(\frac{\pi_{\varphi}(\varphi_1)}{\pi_{\varphi}(\varphi_2)}\right)}{d \ln \left(\mathbf{J}^e(\varphi_1) - \mathbf{J}^e(\varphi_2)\right)} = \frac{\mathbf{J}^e(\varphi_1) - \mathbf{J}^e(\varphi_2)}{\sigma_{\varepsilon}}$$
(C.4)

To implement directed entry in practice, first we normalize the location parameter to $\mu_{\varepsilon} = -\sigma_{\varepsilon}\gamma_{\varepsilon}$. This is a way to undo the effect of σ_{ε} on the ex-ante value of entry, J^{e} .⁵³ Second, to recover the $\kappa(\varphi_{1})$ and $\kappa(\varphi_{2})$ parameters, we use equation (C.1) and the free entry condition $J^{e} = 0$ to write:

$$0 = \ln\left(\sum_{arphi \in \Phi} \exp\left(rac{\mathbf{J}^e(arphi) - \kappa(arphi)}{\sigma_arepsilon}
ight)
ight)$$

Then using equation (C.2) for $\pi_{\varphi}(\varphi_1)$, we obtain:

$$\kappa(arphi) = \mathbf{J}^e(arphi) - \sigma_{arepsilon} \ln \left(\pi_{arphi}(arphi)
ight)$$
 ,

for $\varphi \in {\varphi_1, \varphi_2}$. That is, the average cost of entry of each technology is given by the value of entry of that technology (which can be computed with equation (13)), the parameter σ_{ε} , and the calibrated entry fraction of firms of each type.

While the model with directed entry calibrated as explained above delivers the same model outcomes as the baseline model with random entry, the counterfactuals are going to differ. In particular, as equation (C.3) shows, counterfactuals that change the value of entry with each technology differently, $J^e(\varphi_1) - J^e(\varphi_2)$, will change the number of entrants with each technology in the model with random entry.

D Numerical Appendix

D.1 Idiosyncratic Productivity

In the model, idiosyncratic productivity z is governed by a continuous-time Markov chain (CTMC), with associated infinitesimal generator matrix:

$$\mathbf{\Lambda}_{z} = \begin{pmatrix} -\sum_{j \neq 1} \lambda_{1j} & \lambda_{12} & \dots & \lambda_{1k} \\ \lambda_{21} & -\sum_{j \neq 2} \lambda_{2j} & \dots & \lambda_{2k} \\ \vdots & \vdots & \ddots & \vdots \\ \lambda_{k1} & \lambda_{k2} & \dots & -\sum_{j \neq k} \lambda_{kj} \end{pmatrix}$$

⁵³When σ_{ε} is large, the probability of getting a high draw of ε_{φ} increases, and this increases the expected value function. Setting $\mu_{\varepsilon} = -\sigma_{\varepsilon}\gamma_{\varepsilon}$ removes this effect.

where $\lambda_{ij} > 0$ is short-hand for $\lambda(z_j|z_i)$, $z_i, z_j \in \mathbb{Z}$. For the numerical implementation, we recover these rates by assuming an Ornstein-Uhlenbeck (OU) process for z_t (in logs):

$$d\ln(z_t) = -\rho_z \ln(z_t) dt + \sigma_z dB_t$$

where B_t is a Wiener process, and $\rho_z, \sigma_z > 0.54$ This is a continuous-time process defined on a continuous support. To simulate such a process, and draw a one-to-one mapping between the (ρ_z, σ_z) parameters and the $\{\lambda_{ij}\}$ rates, we use the following steps:

1. First, we approximate the process in discrete time. For a given time interval $[0, T] \subset \mathbb{R}_+$, we partition the space into M subintervals of equal length dt > 0, i.e. $\mathcal{T} \equiv \{0 = t_0 < t_1 < \cdots < t_M = T\}$ with $t_{m+1} - t_m = dt$ and dt = T/M. As the model is calibrated at the quarterly frequency, dt represents a quarter. Then, we approximate the OU process using the Euler-Maruyama method:

$$\ln(z_m) = (1 - \rho_z dt) \ln(z_{m-1}) + \sigma_z \sqrt{dt} \varepsilon_m^z, \qquad \varepsilon_m^z \sim \text{i.i.d. } \mathcal{N}(0, 1)$$
(D.1)

for each m = 1, ..., M. From Ruiz-García (2021), we know that the autocorrelation coefficient and the dispersion in firm-level TFP in Spain is 0.81 and 0.34 at an annual frequency. Therefore, for persistence, we set $\rho_z = 1 - (0.81)^{1/4} = 0.0513$ for our quarterly calibration. Moreover, we compute a quarterly figure for dispersion from the yearly dispersion parameter as $\sigma_z = 0.34 \left(\sum_{q=1}^{4} (0.81)^{(q-1)/2} \right)^{-1/2} = 0.1833$.

- 2. To estimate the discrete-time AR(1) process (D.1), we use the Tauchen (1986) method. The outcome of this method is a transition probability matrix $\Pi_z = (\pi_{ij})$, where π_{ij} denotes the probability of a z_i -to- z_j transition in the \mathcal{T} space.
- 3. For the mapping from $\{\pi_{ij}\}$ transition probabilities to $\{\lambda_{ij}\}$ intensity rates, we use that any CTMC with generator matrix Λ_z maps into a discrete-time Markov chain with transition matrix $\Pi_z(t)$ at time *t* in which holding times between arrivals are independently and exponentially distributed, so that $\Pi_z(t) = e^{\Lambda_z t}$. Then, we can solve for $\{\lambda_{ij}\}$ to obtain:

$$\lambda_{ij} = \begin{cases} -\frac{\pi_{ij}}{1 - \pi_{ii}} \frac{\ln(\pi_{ii})}{dt} & \text{ for } i \neq j \\ \frac{\ln(\pi_{ii})}{dt} & \text{ otherwise} \end{cases}$$

D.2 Stationary Solution Algorithm

We solve the model on a grid $\overline{N}_{OEH} \times \overline{N}_{OEL} \times \overline{N}_{FT} \times \mathcal{Z} \times \Phi$, where $\overline{N}_{ij} \equiv \{0, 1, 2, ..., N_{ij}\}, (i, j) \in \mathcal{I} \times \mathcal{J}$, for sufficiently large positive integer N_{ij} . In practice, we use $(N_{OEH}, N_{OEL}, N_{FT}) = (30, 15, 15)$.

Step 0. Set $\iota = 0$. Choose a guess $\mathbf{U}^{(0)} > b/\rho$.

Step 1. At iteration $\iota \in \{0, 1, 2, ...\}$, given a guess $\mathbf{U}^{(\iota)}$, use Value Function Iteration to solve for the object $\mathbf{\Sigma}^{(\iota)} \in \overline{\mathcal{N}}_{OEH} \times \overline{\mathcal{N}}_{FT} \times \mathcal{Z} \times \Phi$:⁵⁵

⁵⁴In levels, this is a diffusion of the type $dz_t = \mu(z_t)dt + \sigma(z_t)dB_t$, with $\mu(z) = z\left(-\rho_z \ln(z) + \frac{\sigma_z^2}{2}\right)$ and $\sigma(z) = \sigma_z z$.

 $^{^{55}}$ To arrive at this expression, we have used results (8) and (A.27) into equation (A.10).

$$\begin{split} \boldsymbol{\Sigma}^{(\iota)}(\vec{n},z,\varphi) &= \frac{1}{\rho^{(\iota)}(\vec{n},z,\varphi)} \Biggl\{ y(\vec{n},z,\varphi) - \xi n_{FT} \bigl[p^{(\iota)}(\vec{n},z,\varphi) \bigr]^{\vartheta} - \sum_{i \in \mathcal{I}} A_i \bigl[v_i^{(\iota)}(\vec{n},z,\varphi) \bigr]^{\zeta} \\ &+ \sum_{i \in \mathcal{I}} \sum_{j \in \mathcal{J}} \left[n_{ij} \Bigl(\delta_{ij}^{(\iota)}(\vec{n},z,\varphi) + s_i^W + s^F \Bigr) \mathbf{U}^{(\iota)} - \chi n_{ij} \bigl[\delta_{ij}^{(\iota)}(\vec{n},z,\varphi) \bigr]^{\psi} + n_{ij} \Bigl(\delta_{ij}^{(\iota)}(\vec{n},z,\varphi) + s_i^W \Bigr) \boldsymbol{\Sigma}^{(\iota)}(\vec{n}_{ij}^-,z,\varphi) \Biggr] \\ &+ \sum_{i \in \mathcal{I}} \left[v_i^{(\iota)}(\vec{n},z,\varphi) \eta^{(\iota)}(\vec{n},z,\varphi) \max \left(\boldsymbol{\Sigma}^{(\iota)}(\vec{n}_{iL}^+,z,\varphi) - \mathbf{U}^{(\iota)}, \gamma \Bigl(\boldsymbol{\Sigma}^{(\iota)}(\vec{n}_{iL}^+,z,\varphi) - \mathbf{U}^{(\iota)} \Bigr) + (1-\gamma) \boldsymbol{\Sigma}^{(\iota)}(\vec{n},z,\varphi) \Bigr) \Biggr] \\ &+ n_{FT} p^{(\iota)}(\vec{n},z,\varphi) \boldsymbol{\Sigma}^{(\iota)}(\vec{n}^P,z,\varphi) + n_{OEL} \tau \boldsymbol{\Sigma}^{(\iota)}(\vec{n}^{\tau},z,\varphi) + \sum_{z' \in \mathcal{Z}} \lambda(z'|z) \boldsymbol{\Sigma}^{(\iota)}(\vec{n},z',\varphi) \Biggr\} \end{split}$$

for each $\varphi \in \Phi$, where:

$$\begin{split} \delta_{ij}^{(\iota)}(\vec{n},z,\varphi) &\equiv \left[\frac{1}{\psi\chi} \left(\mathbf{U}^{(\iota)} + \mathbf{\Sigma}^{(\iota)}(\vec{n}_{ij}^{-},z,\varphi) - \mathbf{\Sigma}^{(\iota)}(\vec{n},z,\varphi)\right)\right]^{\frac{1}{\psi-1}} \\ p^{(\iota)}(\vec{n},z,\varphi) &\equiv \left[\frac{1}{\psi\xi} \left(\mathbf{\Sigma}^{(\iota)}(\vec{n}^{p},z,\varphi) - \mathbf{\Sigma}^{(\iota)}(\vec{n},z,\varphi)\right)\right]^{\frac{1}{\psi-1}} \\ v_{i}^{(\iota)}(\vec{n},z,\varphi) &\equiv \left[\left(\frac{\gamma}{\zeta A_{i}}\right)\eta^{(\iota)}(\vec{n},z,\varphi) \left(\mathbf{\Sigma}^{(\iota)}(\vec{n}_{iL}^{+},z,\varphi) - \mathbf{\Sigma}^{(\iota)}(\vec{n},z,\varphi) - \mathbf{U}^{(\iota)}\right)\right]^{\frac{1}{\zeta-1}} \\ \eta^{(\iota)}(\vec{n},z,\varphi) &\equiv \left[\left(\frac{1-\gamma}{\rho\mathbf{U}^{(\iota)}-b}\right) \left(\mathbf{\Sigma}^{(\iota)}(\vec{n}_{iL}^{+},z,\varphi) - \mathbf{\Sigma}^{(\iota)}(\vec{n},z,\varphi) - \mathbf{U}^{(\iota)}\right)\right]^{\frac{1-\gamma}{\gamma}} \\ \rho^{(\iota)}(\vec{n},z,\varphi) &\equiv \rho + s^{F} + n_{FT}p^{(\iota)}(\vec{n},z,\varphi) + n_{OEL}\tau + \sum_{i\in\mathcal{I}} \left[v_{i}^{(\iota)}(\vec{n},z,\varphi)\eta^{(\iota)}(\vec{n},z,\varphi) + \sum_{j\in\mathcal{J}} n_{ij} \left(\delta_{ij}^{(\iota)}(\vec{n},z,\varphi) + s_{i}^{W}\right)\right] \end{split}$$

Step 2. Use a non-linear equation solver to find U^* as the solution to the free-entry condition:

$$-\kappa + \sum_{\varphi \in \Phi} \pi_{\varphi}(\varphi) \cdot \mathbf{J}_{e}^{(\iota)}(\varphi; \mathbf{U}^{*}) = 0$$

where

$$\mathbf{J}_{e}^{(\iota)}(\varphi;\mathbf{U}^{*}) \equiv \gamma \left(\frac{\rho \mathbf{U}^{*}-b}{1-\gamma}\right)^{\frac{\gamma-1}{\gamma}} \left[\sum_{z^{e} \in \mathcal{Z}} \pi_{z}(z^{e}) \left(\sum_{i \in \mathcal{I}} \left(\boldsymbol{\Sigma}^{(\iota)}(\vec{n}_{iL}^{e}, z^{e}, \varphi) - \mathbf{U}^{*}\right)^{\frac{1}{\gamma}}\right)\right]$$

is the value of entering with type φ , with $\vec{n}_i^e = (n_i^e, \vec{n}_{-(ij)}^e) = (1, \vec{0})$.

Step 3. Compute

$$oldsymbol{\epsilon}^{(\iota)} \equiv \left| rac{\mathbf{U}^{(\iota)} - \mathbf{U}^*}{\mathbf{U}^*}
ight|$$

Proceed to Step 4 if $\epsilon^{(l)} < \epsilon$ for some small tolerance $\epsilon > 0$. Otherwise, set $\mathbf{U}^{(l+1)} = \varrho \cdot \mathbf{U}^* + (1-\varrho) \cdot \mathbf{U}^{(l)}$, where $\varrho \in (0,1]$ is a dampening parameter, and go back to Step 1 with $[l] \leftarrow [l+1]$.

Step 4. Find the distribution and aggregate measures of agents as described in Online Appendix A.2.

E Additional Figures and Tables



Figure E.1: Aggregate temporary share over time

Notes: Light blue line: share of temporary workers in the Encuesta de Población Activa (the Spanish Labor Force Survey); grey line: average share of temporary workers across firms in our firm-level data; dark blue line: employment-weighted average share of temporary workers across firms in our firm-level data, which corresponds to the average share of temporary workers in the economy.



Figure E.2: Temporary share, by firm size, in the sub-sample.

Notes: This figure is the analogue of Figure 1 in the main text, but for the sub-sample used to calibrate the model.



Notes: Global identification results based on approximately 1.3 million model solutions. The black dots are the median of the distribution of each targeted moment for given value of the chosen parameter, generated from underlying random variation in all other parameters. The shaded areas are the 25th and 75th percentiles. The red dashed horizontal line is the data target.



Notes: This figure plots the distribution of firms and employment by firm size (total number of employees) and firm age (in years), in the data and in the calibrated model.





Notes: This figure plots the distribution of employment growth rates for total employment (top), FT employment (middle) and OE employment (bottom), in the data and in the calibrated model.



Figure E.6: Misallocation Across Productivity Classes in the Baseline Calibration.

Notes: This figure compares the allocation of employment (top row) and of the skill share (bottom row) for firms of different z's (x-axis) and φ 's (left and right panels), in the competitive equilibrium (CE) and the Benchmark's solutions. For the CE, we provide the average employment within each productivity group. For the Benchmark, we provide the employment level of the corresponding productivity group. The magnitudes can be read from the left-hand axes. We also plot the stationary distribution of firms across productivity levels in the CE (right-hand axes).



Notes: These figures show the percentage change in employment (top row), the skill share (middle row) and the temporary share (bottom row) for high-type (left column) and low-type (right column) firms and for each transitory productivity bin (bins on the horizontal axis) when moving from the baseline duration of FTCs to a policy that limits FTCs to 1 month. When available (i.e. for total employment and the skill share), we also show the percentage change for the Benchmark allocation that maximizes total output.

A. Productivity: $\ln(VA_{it}/Emp_{it})$			B. Temporary share: <i>TempSh</i> _{it}		
eta_0^A	$1[arphi_i=arphi_2]$	-0.003	eta_0^B	$1[\varphi_i=\varphi_2]$	0.201
β_1^A	$\ln(Emp_{it})$	0.081	eta_1^B	$Emp_{it} \in [1,5)$	0.005
β_2^A	TempSh _{it}	-0.104	eta_2^B	$Emp_{it} \in [5, 10)$	0.037
β_3^A	$TempSh_{it}^2$	-0.092	eta_3^B	$Emp_{it} \in [10, 15)$	0.080
			eta_4^B	$Emp_{it} \in [15, 20)$	0.099
			eta_5^B	$Emp_{it} \in [20, 25)$	0.113
			eta_6^B	$Emp_{it} \in [25, 30)$	0.122
			eta_7^B	$Emp_{it} \in [30, 35)$	0.129
			eta_8^B	$Emp_{it} \in [35, 40)$	0.135
			eta_9^B	$Emp_{it} \in [40, 45)$	0.142
			eta^B_{10}	$Emp_{it} \in [45, 50)$	0.150
			eta^B_{11}	$Emp_{it} \in [50, 55]$	0.156
			eta^B_{12}	$Emp_{it} \in [55, 60]$	0.190
# observations		6,316,320	# obse	rvations	6,664,229
<i>R</i> ²		0.01	R^2		0.14

Notes: Results, in the firm-level data, from regression (17), in Panel A, and regression (18), in Panel B. Variables are all net of aggregate FE (sector, region, and province). All coefficients are significant at the 1% level.
	(A)	(B)	(C)
	Short duration	Baseline	Long duration
	(1 month)	(6.5 months)	(1 years)
Measure of operating firms	0.106	0.121	0.140
Share of type- φ_1 firms	65.5 %	40.2 %	19.0 %
Average firm size	7.78	7.01	6.68
Relative size φ_2 firms	0.53	0.65	0.80
Firm entry rate (annualized)	7.4 %	7.5 %	7.5 %
Average temporary share	0.6 %	21.8 %	60.9 %
within φ_1 firms	0.3 %	5.2 %	13.3 %
within φ_2 firms	1.8 %	32.9 %	74.8 %
Share of <i>H</i> workers	60.4 %	46.0 %	30.7 %
within φ_1 firms	71.7 %	71.3 %	70.2 %
within φ_2 firms	20.1 %	19.7 %	19.1 %
Matching efficiency (FT)	0.21	1.40	0.57
Matching efficiency (OE)	0.09	0.03	0.01
UE rate (total)	8.0 %	20.9 %	24.5 %
UE rate (FT)	2.4 %	18.4 %	22.1 %
UE rate (OE)	5.6 %	2.5 %	2.4 %
EU rate (total)	1.8 %	3.7 %	1.7 %
EU rate (FT)	76.4 %	11.9 %	2.3 %
EU rate (OE)	1.3 %	1.5 %	0.9 %
Promotion rate	11.1 %	3.0 %	0.2 %
Unemployment rate	17.8 %	14.9 %	6.3 %

Table E.2: Effects of changes in average FTC duration on macro aggregates when entry is directed

Notes: This table is the same as Table 5 in the main text, but for the economy with directed entry and an entry elasticity of one. For the description of the extension of the model with directed entry, see Online Appendix C.

Table E.3: Effects of changes in average FTC duration on productivity, misallocation and welfare when entry is directed

	(A)	(B)
	Short duration (1 month)	Long duration (4 years)
Change in output per worker, of which:	-0.08 %	-0.49 %
(a) Firm size channel	-1.24 %	0.57 %
(b) Firm selection channel	12.38 %	-11.36 %
(c) Reallocation channel, of which:	-14.42 %	6.06 %
between-firm component	-14.33 %	5.68 %
within-firm component	-9.11 %	4.25 %
Change in output	-3.48 %	9.64 %
Output loss from misallocation (in levels), of which:	4.49 %	3.12 %
share due to between-firm misallocation	44.81 %	59.84 %
Change in welfare	0.05 %	0.88 %

Notes: This table is the same as Table 6 in the main text, but for the economy with directed entry and an entry elasticity of one. For the description of the extension of the model with directed entry, see Online Appendix C.

		(A)	(B)	(C)
	Baseline calibration	Short duration	Ban on FTCs	Linear Tax on FT Employment
Measure of operating firms	0.121	0.112	0.120	0.114
Share of type- φ_1 firms	40.2 %	47.5 %	45.7%	47.5 %
Average firm size	7.01	7.03	6.91	7.05
Relative size φ_2 firms	0.65	0.54	0.54	0.54
Firm entry rate (annualized)	7.5 %	7.4 %	7.4 %	7.4 %
Average temporary share	21.8 %	0.9 %		0.9 %
within φ_1 firms	5.2 %	0.3 %		0.4 %
within φ_2 firms	32.9 %	1.8 %		1.6 %
Share of H workers	46.0 %	52.5 %	51.5 %	52.6 %
within φ_1 firms	71.3 %	71.7 %	71.7 %	71.7 %
within φ_2 firms	19.7 %	20.2 %	19.7%	20.4 %
Matching efficiency (FT)	1.40	0.28	0.42	0.04
Matching efficiency (OE)	0.03	0.11	0.04	0.09
UE rate (total)	20.9 %	8.3 %	7.5 %	7.0 %
UE rate (FT)	18.4 %	2.7 %		0.9 %
UE rate (OE)	2.5 %	5.6 %		6.1 %
EU rate (total)	3.7 %	2.3 %	1.7 %	1.8 %
EU rate (FT)	11.9 %	75.9 %		16.7 %
EU rate (OE)	1.5 %	1.6 %		1.6 %
Promotion rate	3.0 %	10.9 %		12.9 %
Unemployment rate	14.9 %	21.0 %	17.8 %	19.6 %

Table E.4: Effects on macro aggregates of various alternative policies

Notes: This table shows the effects of various policies on a number of macroeconomic aggregates of interest. The tax rate in column (C) have been chosen to ensure that the effect of the policy on the average temporary share is the same as the baseline policy discussed in the paper, in which the duration of FTCs decreases to 1 month.

Table E.5: Effects on productivity, misallocation and welfare of various alternative policies

	(A)	(B)	(C)
	Short duration	Ban on FTCs	Linear Tax on FT Employment
Change in output per worker, of which:	0.26 %	0.24 %	0.24 %
(a) Firm size channel	-0.05 %	0.17 %	-0.07 %
(b) Firm selection channel	4.91 %	2.70 %	4.95 %
(c) Reallocation channel, of which:	-5.77 %	-3.56 %	-5.77 %
between-firm component	-5.62 %	-3.32 %	-5.64 %
within-firm component	-0.70 %	0.87 %	-0.84 %
Change in output	-6.86 %	-3.18 %	-5.24 %
Output loss from misallocation (in levels), of which:	4.29 %	4.22 %	4.29 %
share due to between-firm misallocation	39.52 %	37.87 %	39.64 %
Change in welfare	-0.87 %	-0.31 %	-0.65 %

Notes: This table shows the effect of various policies, expressed in percentage changes with respect to the baseline calibration (with the exception of the output loss from misallocation and the share of it that is due to between-firm misallocation, which are expressed in levels). Welfare is computed as in equation (A.37), see Online Appendix A.6. The taxes rates in columns (B) and (C) have been chosen to ensure that the effect of the policy on the average temporary share is the same as the baseline policy discussed in the paper, in which the duration of FTCs decreases to 1 month.

BANCO DE ESPAÑA PUBLICATIONS

WORKING PAPERS

- 2320 JUAN S. MORA-SANGUINETTI, LAURA HOSPIDO and ANDRÉS ATIENZA-MAESO: The numbers of equality regulation. Quantifying regulatory activity on non-discrimination and its relationship with gender gaps in the labour market.
- 2321 ANDRES ALONSO-ROBISCO and JOSÉ MANUEL CARBÓ: Analysis of CBDC Narrative of Central Banks using Large Language Models.
- 2322 STEFANIA ALBANESI, ANTÓNIO DIAS DA SILVA, JUAN F. JIMENO, ANA LAMO and ALENA WABITSCH: New technologies and jobs in Europe.
- 2323 JOSÉ E. GUTIÉRREZ: Optimal regulation of credit lines.
- 2324 MERCEDES DE LUIS, EMILIO RODRÍGUEZ and DIEGO TORRES: Machine learning applied to active fixed-income portfolio management: a Lasso logit approach.
- 2325 SELVA BAHAR BAZIKI, MARÍA J. NIETO and RIMA TURK-ARISS: Sovereign portfolio composition and bank risk: the case of European banks.
- 2326 ANGEL-IVAN MORENO and TERESA CAMINERO: Assessing the data challenges of climate-related disclosures in european banks. A text mining study.
- 2327 JULIO GÁLVEZ: Household portfolio choices under (non-)linear income risk: an empirical framework.
- 2328 NATASCHA HINTERLANG: Effects of Carbon Pricing in Germany and Spain: An Assessment with EMuSe.
- 2329 RODOLFO CAMPOS, SAMUEL PIENKNAGURA and JACOPO TIMINI: How far has globalization gone? A tale of two regions.
- 2330 NICOLÁS FORTEZA and SANDRA GARCÍA-URIBE: A Score Function to Prioritize Editing in Household Survey Data: A Machine Learning Approach.
- 2331 PATRICK MACNAMARA, MYROSLAV PIDKUYKO and RAFFAELE ROSSI: Taxing consumption in unequal economies.
- 2332 ESTHER CÁCERES and MATÍAS LAMAS: Dividend Restrictions and Search for Income.
- 2333 MARGARITA MACHELETT: Gender price gaps and competition: Evidence from a correspondence study.
- 2334 ANTON NAKOV and CARLOS THOMAS: Climate-conscious monetary policy.
- 2335 RICARDO BARAHONA, STEFANO CASSELLA and KRISTY A. E. JANSEN: Do teams alleviate or exacerbate the extrapolation bias in the stock market?
- 2336 JUAN S. MORA-SANGUINETTI and ANDRÉS ATIENZA-MAESO: "Green regulation": A quantification of regulations related to renewable energy, sustainable transport, pollution and energy efficiency between 2000 and 2022.
- 2401 LAURA HOSPIDO, NAGORE IRIBERRI and MARGARITA MACHELETT: Gender gaps in financial literacy: a multi-arm RCT to break the response bias in surveys.
- 2402 RUBÉN DOMÍNGUEZ-DÍAZ, SAMUEL HURTADO and CAROLINA MENÉNDEZ: The medium-term effects of investment stimulus.
- 2403 CLODOMIRO FERREIRA, JOSÉ MIGUEL LEIVA, GALO NUÑO, ÁLVARO ORTIZ, TOMASA RODRIGO and SIRENIA VAZQUEZ: The heterogeneous impact of inflation on households' balance sheets.
- 2404 JORGE ABAD, GALO NUÑO and CARLOS THOMAS: CBDC and the operational framework of monetary policy.
- 2405 STÉPHANE BONHOMME and ANGELA DENIS: Estimating individual responses when tomorrow matters.
- 2406 LAURA ÁLVAREZ-ROMÁN, SERGIO MAYORDOMO, CARLES VERGARA-ALERT and XAVIER VIVES: Climate risk, soft information and credit supply.
- 2407 JESÚS FERNÁNDEZ-VILLAVERDE, JOËL MARBET, GALO NUÑO and OMAR RACHEDI: Inequality and the zero lower bound.
- 2408 PABLO BURRIEL, MAR DELGADO-TÉLLEZ, CAMILA FIGUEROA, IVÁN KATARYNIUK and JAVIER J. PÉREZ: Estimating the contribution of macroeconomic factors to sovereign bond spreads in the euro area.
- 2409 LUIS E. ROJAS and DOMINIK THALER: The bright side of the doom loop: banks' sovereign exposure and default incentives.
- 2410 SALOMÓN GARCÍA-VILLEGAS and ENRIC MARTORELL: Climate transition risk and the role of bank capital requirements.
- 2411 MIKEL BEDAYO and JORGE E. GALÁN: The impact of the Countercyclical Capital Buffer on credit: Evidence from its accumulation and release before and during COVID-19.
- 2412 EFFROSYNI ADAMOPOULOU, LUIS DÍEZ-CATALÁN and ERNESTO VILLANUEVA: Staggered contracts and unemployment during recessions.
- 2413 LUIS FÉRNANDEZ LAFUERZA and JORGE E. GALÁN: Should macroprudential policy target corporate lending? Evidence from credit standards and defaults.
- 2414 STÉPHANE BONHOMME and ANGELA DENIS: Estimating heterogeneous effects: applications to labor economics.

- 2415 LUIS GUIROLA, LAURA HOSPIDO and ANDREA WEBER: Family and career: An analysis across Europe and North America.
- 2416 GERALD P. DWYER, BILJANA GILEVSKA, MARÍA J. NIETO and MARGARITA SAMARTÍN: The effects of the ECB's unconventional monetary policies from 2011 to 2018 on banking assets.
- 2417 NICOLÁS FORTEZA, ELVIRA PRADES and MARC ROCA: Analysing the VAT cut pass-through in Spain using webscraped supermarket data and machine learning.
- 2418 JOSÉ-ELÍAS GALLEGOS: HANK beyond FIRE: Amplification, forward guidance, and belief shocks.
- 2419 DANIEL ALONSO: Stabilisation properties of a SURE-like European unemployment insurance.
- 2420 FRANCISCO GONZÁLEZ, JOSÉ E. GUTIÉRREZ and JOSÉ MARÍA SERENA: Shadow seniority? Lending relationships and borrowers' selective default.
- 2421 ROBERTO BLANCO, MIGUEL GARCÍA-POSADA, SERGIO MAYORDOMO and MARÍA RODRÍGUEZ-MORENO: Access to credit and firm survival during a crisis: the case of zero-bank-debt firms.
- 2422 FERNANDO CEREZO, PABLO GIRÓN, MARÍA T. GONZÁLEZ-PÉREZ and ROBERTO PASCUAL: The impact of sovereign debt purchase programmes. A case study: the Spanish-to-Portuguese bond yield spread.
- 2423 EDGAR SILGADO-GÓMEZ: Sovereign uncertainty.
- 2424 CLODOMIRO FERREIRA, JULIO GÁLVEZ and MYROSLAV PIDKUYKO: Housing tenure, consumption and household debt: life-cycle dynamics during a housing bust in Spain.
- 2425 RUBÉN DOMÍNGUEZ-DÍAZ and SAMUEL HURTADO: Green energy transition and vulnerability to external shocks.
- 2426 JOSEP GISBERT and JOSÉ E. GUTIÉRREZ: Bridging the gap? Fintech and financial inclusion.
- 2427 RODOLFO G. CAMPOS, MARIO LARCH, JACOPO TIMINI, ELENA VIDAL and YOTO V. YOTOV: Does the WTO Promote Trade? A Meta-analysis.
- 2428 SONER BASKAYA, JOSÉ E. GUTIÉRREZ, JOSÉ MARÍA SERENA and SERAFEIM TSOUKAS: Bank supervision and non-performing loan cleansing.
- 2429 TODD E. CLARK, GERGELY GANICS, and ELMAR MERTENS: Constructing fan charts from the ragged edge of SPF forecasts.
- 2430 MIGUEL GARCÍA-POSADA and PETER PAZ: The transmission of monetary policy to credit supply in the euro area.
- 2431 KLODIANA ISTREFI, FLORENS ODENDAHL and GIULIA SESTIERI: ECB communication and its impact on financial markets.
- 2432 FRUCTUOSO BORRALLO, LUCÍA CUADRO-SÁEZ, CORINNA GHIRELLI and JAVIER J. PÉREZ: "El Niño" and "La Niña": Revisiting the impact on food commodity prices and euro area consumer prices.
- 2433 VÍCTOR CABALLERO, CORINNA GHIRELLI, ÁNGEL LUIS GÓMEZ and JAVIER J. PÉREZ: The public-private wage GAP in the euro area a decade after the sovereign debt crisis.
- 2434 LIDIA CRUCES, ISABEL MICÓ-MILLÁN and SUSANA PÁRRAGA: Female financial portfolio choices and marital property regimes.
- 2435 RODOLFO G. CAMPOS, ANA-SIMONA MANU, LUIS MOLINA and MARTA SUÁREZ-VARELA: China's financial spillovers to emerging markets.
- 2436 LUDOVIC PANON, LAURA LEBASTARD, MICHELE MANCINI, ALESSANDRO BORIN, PEONARE CAKA, GIANMARCO CARIOLA, DENNIS ESSERS, ELENA GENTILI, ANDREA LINARELLO, TULLIA PADELLINI, FRANCISCO REQUENA and JACOPO TIMINI: Inputs in Distress: Geoeconomic Fragmentation and Firms' Sourcing.
- 2437 DANIEL DEJUAN-BITRIA, WAYNE R. LANDSMAN, SERGIO MAYORDOMO and IRENE ROIBÁS: How do changes in financial reporting standards affect relationship lending?
- 2438 ALICIA AGUILAR and RICARDO GIMENO: Discrete Probability Forecasts: What to expect when you are expecting a monetary policy decision.
- 2439 RODOLFO G. CAMPOS, JESÚS FERNÁNDEZ-VILLAVERDE, GALO NUÑO and PETER PAZ: Navigating by Falling Stars: Monetary Policy with Fiscally Driven Natural Rates.
- 2440 ALEJANDRO CASADO and DAVID MARTÍNEZ-MIERA: Local lending specialization and monetary policy.
- 2441 JORGE ABAD, DAVID MARTÍNEZ-MIERA and JAVIER SUÁREZ: A macroeconomic model of banks' systemic risk taking.
- 2442 JOSEP PIJOAN-MAS and PAU ROLDAN-BLANCO: Dual labor markets and the equilibrium distribution of firms.